WiFiProtect® 3e-520 SERIES USER'S GUIDE

Document Number: 29010012-001, Revision J2

21 December 2020

Prepared by:



Ultra Intelligence & Communications

12410 Milestone Center Drive, Germantown, MD 20876 Tel 800-449-3384 Fax 301-515-1027 www.Ultra-3eTI.com

CONFIDENTIAL & PROPRIETARY INFORMATION OF 3eTI: THE INFORMATION CONTAINED IN THIS DOCUMENT IS CONFIDENTIAL TO AND THE PROPRIETARY PROPERTY OF ULTRA ELECTRONICS, 3eTI. REPRODUCTION OR USE OF THIS INFORMATION WITHOUT THE WRITTEN CONSENT OF 3eTI IS STRICTLY PROHIBITED.

Copyright © 2020 3e Technologies International, Inc. (3eTI). All rights reserved. No part of this documentation may be reproduced in any form or by any means or to make any derivative work (such as translation, transformation, or adaptation) without written permission from 3eTI.

3eTI provides this documentation without warranty, term or condition of any kind, either implied or expressed, including, but not limited to, the implied warranties, terms, or conditions of merchantability, satisfactory quality, and fitness for a particular purpose. 3eTI may make improvements or changes in the product(s) and/or the program(s) described in this documentation at any time. Certain features listed may have restricted availability and/or are subject to change without notice.

If there is any software or removable media described in this documentation, it may be furnished under a license agreement included with the product as a separate document, in the printed documentation, or on the removable media in a readable file such as license.txt or the like. If you are unable to locate a copy of the license, please contact 3eTI to determine whether a copy should be provided to you.

Government Rights Legend

The U.S. Government's rights in this documentation, the products described, and all technical data and computer software are limited by DFARS 252.227 7014 pertaining to restricted rights software, and DFAR 252.227-7015 pertaining to limited rights technical data developed at private expense; or currently limited under DFARS 252.227-7018 small business innovative research programs, whichever is applicable.

3e Technologies International, the 3e Technologies International logo and WiFiProtect are registered trademarks.

Windows is a registered trademark of Microsoft Corporation. Any other company and product name mentioned herein is a trademark of the respective company with which they are associated.

Export Restrictions

This product contains components, software, and/or firmware which, if and when required to be exported from the United States will be done in accordance with U. S. export administration regulations. Diversion contrary to U.S. law is prohibited.

Revision History

Revision Letter	Date	Description
A	05-12-2014	Initial Release
В	03-19-2015	Revised for Common Criteria
С	03-26-2015	Updated with review comments
D	08-11-2015	Updated with reivew comments and fixed missing figures
E	09-10-2017	Updated for Common Criteria NDcPP20/WLANAScEP10
F	05-04-2018	Updated with review comments
G	05-08-2018	Updated with review comments
Н	03-24-2020	Updated for DoDIN APL
I	10-22-2020	Include model 3e-523E-900
J2	12-21-2020	Updated 3e-523E-900 installation, channel and tx level specs,

Table of Contents

Gov	Government Rights Legendi		
Ехр	ort Restrictionsi		
1.	Introduction1		
1.1	Typical Deployment1		
1.2	Products and Features		
1.2.1	3e-523N Access Point2		
1.2.2	3e-525N Access Point		
1.3	Wireless and Networking Basics9		
1.3.1	IEEE 802.119		
1.3.2	Wireless Network Topologies9		
1.3.3	MAC Address Filtering		
1.3.4	DHCP Server		
1.3.5	Data Encryption and Security13		
1.3.6	Ultra 3eTI Wireless VLAN		
1.4	Device Management and Administration15		
1.4.1	User Roles15		
1.4.2	Local User Management		
2.	Device Configuration		
2.1	Quick Setup17		
2.1.1	3e-523N		
2.1.2	3e-525N		
2.2	Initial Device Configuration		
2.2.1	Initial Setup		
2.2.2	Login20		
2.3	System Configuration		
2.3.1	Configuration Information20		
2.3.2	General21		
2.3.3	Noisy Channel Control		
2.3.4	WAN		
2.3.5	LAN (Local Management)		

2.3.6	Bridge	28
2.3.7	Ethernet VLAN	28
2.3.8	MAC Address Filtering	29
2.3.9	Certificate Store	31
2.4	3e-520 Series Radio Configuration	37
2.4.1	Introduction	37
2.4.2	Radio WLAN Mode Configuration	38
2.4.3	Radio PHY Setting Configuration	39
2.4.4	Access Point Configuration	41
2.4.5	Mesh Point Configuration	48
2.4.6	Wireless Client Configuration	52
2.5	IPsec Configuration	55
2.5.1	IPsec Tunnel Profiles Configuration	55
2.5.2	IPsec Tunnel Status	58
2.6	Services Settings	58
2.6.1	DHCP Server	58
2.6.2	SNMP Agent	59
2.6.3	Serial Port (3e-523N Only)	62
2.6.4	Serial Communication (3e-523N Only)	64
2.6.5	Remote Administration	65
2.6.6	Web Server	67
2.7	Admin User Management	68
2.7.1	List all Users	68
2.7.2	Add New User	69
2.7.3	User Login Policy	71
2.8	Remote Authentication and Authorization	72
2.8.1	Remote A&A Setup	72
2.8.2	Remote A&A User Groups	73
2.8.3	LDAP Server Configuration	73
2.9	Two-Factor Authentication	74
2.9.1	Two-Factor Authentication Overview	74
2.9.2	First-Factor Authentication—CAC and OCSP	75
2.9.3	Second-Factor Authentication—LDAP Server	75

2.10 I	Monitoring/Reports	76
2.10.1	System Status	76
2.10.2	Bridge Status	77
2.10.3	Bridge Site Map	78
2.10.4	Adjacent AP List	79
2.10.5	DHCP Client List	80
2.11 l	Logs	80
2.11.1	System Log	80
2.11.2	Web Access Log	81
2.12	Auditing	82
2.12.1	Audit Configuration	82
2.12.2	Audit Log	86
2.13	System Administration	87
2.13.1	Email Notification Configuration	87
2.13.2	Radio TX Off Control	88
2.13.3	System Upgrade	89
2.13.4	Default Configuration	92
2.13.5	Remote Logging	93
2.13.6	Reboot	94
2.13.7	Self-Test	94
2.13.8	Utilities	97
2.13.9	Help	98
3. 3e	e-523N Hardware Installation	99
3.1 P	reparation for Use	99
3.1.1	Installation Instructions	99
3.2 D	evice Installation	
3.2.1	DIN Rail Mounting	
3.2.2	Rear Mounting	
3.2.3	Base Mounting	
3.2.4	3e-523 Mounting Dimensions	
3.3 A	ccessory Kit Installation	101
4. 3e	e-525N Hardware Installation	102
4.1 In	stallation Preparation	

4.2	Installation Instructions	103
4.2.1	3e-525N Ethernet Cable Assembly	103
4.2.2	Pole Mounting	104
5. 3	3e-523E-900 Hardware Installation	102
5.1	Installation Preparation	102
5.1.1	Specifications	102
5.1.2	Mounting Pattern	103
5.1.3	Installation Instructions	104
5.1.4	RF Connections	105
5.1.5	RF Safety Information	105
App	endix A. Common Criteria	1
A.1	Overview	1
A.2	Common Criteria Compliant Steps	1
A.3	Identifying Secure Delivery of the Device	1
A.4	Running Compliant Firmware	1
A.5	Adding Administrative Users	4
A.6	Configuring System Time	6
A.7	Configuring Access Point Mode	6
A.8	Configuring Wireless Client Session Establishment	7
A.9	Configuring Audit Event Loggin	8
A.10	Understanding Audit Events	9
A.10.	1 Audit Event Record Structure	9
A.10.2	2 Audit Event: Audit Startup and Shutdown	10
A.10.3	3 Audit Event: Audit Coverage Change	10
A.10.4	4 Audit Event: Loss of Connectivity to Server	11
A.10.	5 Audit Event: Administrative Actions Event	11
A.10.0	6 Audit Event: Failure of Key Generation/Distribution	13
A.10.	7 Audit Event: Failure of Encryption or Decryption	13
A.10.8	8 Audit Event: Failure of Cryptographic Signature	14
A.10.9	9 Audit Event: Failure of Hashing Function	14
A.10.	10 Audit Event: Failure of WPA2 Encryption or Decryption	14
A.10.	11 Audit Event: Failure to Establish a HTTPS/TLS Session	14
A.10.	12 Audit Event: Establishment or Termination of a HTTPS/TLS Session	14

A.10.13	Audit Event: Failure of IPsec Security Association Establishment	. 14
A.10.14	Audit Event: Establishment or Termination of an IPsec Security Association	. 15
A.10.15	Audit Event: Failure of Random Bit Generation	. 15
A.10.16	Audit Event: Authentication Failure Handling	. 15
A.10.17	Audit Event: Admin User Authentication	.16
A.10.18	Audit Event: 802.1X Authentication	.16
A.10.19	Audit Event: Certificate Validation and Upload	.16
A.10.20	Audit Event: Failure of the TSF	. 17
A.10.21	Audit Event: Changes to the System Time	. 17
A.10.22	Audit Event: Self-Test	. 17
A.10.23	Audit Event: Successful or Failed System Updates	. 19
A.10.24	Audit Event: Maximum Quota	. 19
A.10.25	Audit Event: Attempts at Unlocking an Interactive Session	. 19
A.10.26	Audit Event: Termination of a Remote Session by the Session Locking Mechanism	. 19
A.10.27	Audit Event: Termination of an Interactive Session	. 20
A.10.28	Audit Event: Denial of a Session Establishment due to MAC Address Filtering	. 20
A.10.29	Audit Event: Initiation of a Trusted Channel	. 20
A.10.30	Audit Event: Termination of a Trusted Channel	.20
A.10.31	Audit Event: Failure of Trusted Channel or Trusted Path Functions	. 20
A.10.32	Audit Event: Trust Channel Connection Re-establishment	.21
A.10.33	Audit Event: Load Certificates that Fail to Meet the NDcPP Requirements	.21
A.10.34	Audit Event: Configure Session Inactivity and Time out	.21
A.10.35	Audit Event: Detection of Modification of Channel Data for 802.11 and 802.1x	.21
A.10.36	Audit Event: Configure/Change the Reference Identifier for Peer	.21
A.11 K	ey Zeroization	.21
A.12 U	Iser Space Processors	.21
Appen	dix B. Term Reference Guide	1
B.1 Ac	ronyms and Abbreviations	1
B.2 Te	rm Definitions	2
Appen	dix C. Serial I/O Interface Board	1
Appen	dix D. 3e-523E-900 Specific Operation	1
D.1 3e	-523E-900 Overview	1

D.2	PHY Settings Specific to 3e-523E-900	. 1
Арр	endix E. Technical Support	2
E.1	Manufacturer's Statement	.2

List of Figures

Figure 1: Typical Deployment Diagram1
Figure 2: 3e-523 External Connectors and Indicators2
Figure 3: 3e-523N Power Supply and Ground Connections
Figure 4: 3e-523N External Connectors and Indicators4
Figure 5: 3e-525N External Interfaces5
Figure 6: AUX Port Pin Definition
Figure 7: Indicator LEDs
Figure 8: Standalone AP Network10
Figure 9: Multiple APs Connected to an Existing Ethernet Network10
Figure 10: Multiple APs Connected to an Existing Ethernet Network with DHCP Server
Figure 11: Dual Bridge Configuration11
Figure 12: Bridge Relay with Different Frequency Bands12
Figure 13: Wireless VLAN Communications14
Figure 14: 3e-525N Quick Setup
Figure 15: Local PC IP Address Configuration
Figure 16: Certificate Error
Figure 17: Login Screen
Figure 18: Login Screen with Defaults Username/Password20
Figure 19: Internet Explorer Successful Login Web Page21
Figure 20: System Configuration – General21
Figure 21: Manual Time Source
Figure 22: NTP Time Source
Figure 23: NTP Time Source with IPsec Protection23
Figure 24: IEEE 1588 Master Time Source & Login Banner24
Figure 25: System Configuration – Noisy Channel Control25
Figure 26: System Configuration – WAN27
Figure 27: System Configuration – LAN (Local Management)
Figure 27: System Configuration – LAN (Local Management)
Figure 27: System Configuration – LAN (Local Management)

Figure 31: Enable Time of Day for MAC Address Filtering	31
Figure 32: System Configuration – Certificate Store	32
Figure 33: Certificate Store Device CSR	33
Figure 34: CSR Generation	34
Figure 35: Device Certs	35
Figure 36: Intermediate CA Certificate Upload	35
Figure 37: Trust Root CA	36
Figure 38: Local CRL	36
Figure 39: OCSP Signer	37
Figure 40: Radio – WLAN Mode	38
Figure 41: Radio 1 – PHY Setting	39
Figure 42: Radio – AP General	41
Figure 43: Radio – AP Security	43
Figure 44: RADIUS Configuration with IPsec Protection	44
Figure 45: RADIUS Configuration with IPsec ID	45
Figure 46: Radio – AP Wireless Clients	45
Figure 47: Radio – Wireless VLAN Mapping	46
Figure 48: Radio – Wireless VLAN Mapping – Enable Wireless VLAN	46
Figure 49: Radio – Wireless VLAN Mapping – Create Wireless VLAN	47
Figure 50: Radio – Wireless Mesh General	49
Figure 51: Wireless Bridge Information (Monitoring)	51
Figure 52: Radio – Auto Bridge Security	51
Figure 53: Radio – PHY Setting	52
Figure 54: Radio – Client General	53
Figure 55: Radio – Client Security Type	53
Figure 56: Radio – Client Security – WPA2-EAP-TLS-CCMP	54
Figure 57: Certificate Store – Loading Client Certificates for WPA2-EAP-TLS-CCMP	54
Figure 58: IPsec Tunnel Profiles	56
Figure 59: IPsec Tunnel Status	58
Figure 60: Services Settings — DHCP Server	59
Figure 61: Services Settings — SNMP Agent	60
Figure 62: System Configuration – Serial Port	62
Figure 63: Services Settings – Serial Communication – Raw Socket	64

Figure 64: Services Settings – Serial Communication – TCP Socket	65
Figure 65: Services Settings — Remote Administration Access Control	66
Figure 66: Services Settings — Web Server	67
Figure 67: Admin User Management — List All Users	68
Figure 68: Admin User Management — Edit User	69
Figure 69: Admin User Management — Add New User	69
Figure 70: Password Aging	70
Figure 71: Admin User Management — User Login Policy	71
Figure 72: Admin User Management – Remote A&A Setup	73
Figure 73: Two-Factor Authentication Overview	74
Figure 74: Admin User Management – Two-Factor Authentication	75
Figure 75: Monitoring/Reports — System Status	76
Figure 76: Monitoring/Reports — Bridge Status	77
Figure 77: Monitoring Reports — Bridging Site Map	78
Figure 78: Monitoring/Reports — Adjacent AP List	79
Figure 79: Monitoring/Reports — DHCP Client List	80
Figure 80: Logs – System Log	81
Figure 81: Logs – Web Access Log	82
Figure 82: Auditing – Configuration	83
Figure 83: Audit Records Limited By User ID	85
Figure 84: Audit Record Local Storage Action	85
Figure 85: Remote Audit Logging with IPsec Tunnel	85
Figure 86: Auditing – Log	87
Figure 87: System Administration – Email Notification Configuration	88
Figure 88: Email Test Result	88
Figure 89: System Administration – Radio TX Off Control	89
Figure 90: System Administration — System Upgrade – Firmware Upgrade	90
Figure 91: System Administration – System Upgrade – Configuration Export/Import	90
Figure 92: System Administration — Factory Default	92
Figure 93: System Administration — Remote Logging	93
Figure 94: System Administration — Reboot	94
Figure 95: System Administration — On Demand Self-test	95
Figure 96: System Administration — Periodic Self-Test	96

Figure 97: System Administration — Utilities	97
Figure 98: System Administration — Help: Hardware and Software Version Information	98
Figure 99: 3e-523N DIN Rail Mounting	100
Figure 100: 3e-523N Rear Mounting	100
Figure 101: 3e-523N Base Mounting	101
Figure 102: 3e-523N Mounting Dimensions	101
Figure 103: 3e-525N Ethernet Cable Assembly	103
Figure 104: 3e-525N Pole Mount Installation	104
Figure 105: 3e-523E-900 Installation	104
Figure 106: Software Version	3
Figure 107: Serial I/O Interface Board	1
Figure 108: Serial I/O Configuration DIP Switch	2

List of Tables

Table 1: 3e-523N External Connectors and Indicators	3
Table 2: 3e-523N Indicator LEDs	4
Table 3: 3e-525N External Interfaces	6
Table 4: 3e-525N AUX Port	7
Table 5: Indicator LED Descriptions	7
Table 6: User Role Services	15
Table 7: Add MAC Address Settings	
Table 8: Frequency Channel Numbers	
Table 9: Radio Settings	
Table 10: 802.1X Configuration	
Table 11: Wireless Mesh General Setting Options	
Table 12: IPsec Tunnel Profile Settings	
Table 13: Cipher Suites	57
Table 14: Service Settings – Serial Communication	63
Table 15: User Group Role & Privilege in LDAP Server	73
Table 16: Auditing – Configuration Event Types and Description	
Table 17: 3e-523N Accessories	

Table 18: 3e-525N Accessories	. 102
Table 19: Operational Environment Objective	1
Table 20: Device Organizational Assumption	2
Table 21: Firmware Requirements	3
Table 22: Administration and Login Policy Requirements	5
Table 23: Time Stamp Requirements	6
Table 24: Access Point Security Requirements	7
Table 25: Access Point Session Establishment	8
Table 26: Audit Log Requirements	9
Table 27: Fields in Audit Event Message	10
Table 28: The List of User Space Processes	21
Table 29: Serial I/O Termination	1
Table 30: Serial I/O Configuration DIP Switch	3
Table 31: DB-9 Serial I/O Connector	3
Table 32: 3e-523E-900 Frequency Channel Numbers	1

1. Introduction

This manual covers the installation and operation of 3e Technologies International's (3eTI) latest members of the WiFiProtect[®] product family. The WiFiProtect 3e-520 series product family consists of the 3e-523N and the 3e-525N Access Point (hereinafter referred to as WiFiProtect products, or 3e-520 series product unless otherwise specified). The WiFiProtect products are secure, ruggedized wireless devices that have been designed and tested for use in harsh demanding environments where durability is a key requirement. These WiFiProtect products support the latest 802.11a/g/n wireless standards and provide security consistent with Federal Information Processing Standards (FIPS) 140-2 requirements using Advanced Encryption Standard (AES) cryptographic modules for wireless encryption and Hyper Text Transfer Protocol Service (HTTPS) / Transport Layer Security (TLS) for secure web communication.

1.1 Typical Deployment

The 3e-520 series products can be deployed in any commercial and industrial solutions to protect the sensitive network and user data. The following diagram (Figure 1) represents a typical deployment using 3e-520 series products to transmit sensitive information between remote sites to the NOC (Network Operations Center) via secured wireless connections. In this deployment, the 3e-520 series products are configured in wireless Mesh, Access Point and Client modes.

The 3e-520 series products support remote management via Active directory LDAP server and OCSP responder (Refer to Section 2.8 and Section 2.9 for more detail). The management communications are protected between itself and the Authentication Server, NTP Server and Remote Audit Server using IPsec tunnels.



Figure 1: Typical Deployment Diagram

1.2 Products and Features

The 3e-520 series includes two different 802.11n products – 3e-523N and 3e-525N. Both products use a common hardware platform and run the same software. The major difference between these two models is single radio vs dual radio. The 3e-523N includes a single 802.11a/g/n radio card, indoor enclosure and is powered by a DC power adapter. The 3e-525N comes with dual 802.11a/g/n radio cards, an outdoor-rated enclosure and is powered by a PoE injector.

Please refer to the product-specific hardware sections later in this document for additional technical details.

1.2.1 3e-523N Access Point

The 3e–523N Data Point has the following key features:

- FIPS 140-2, 802.11 Wi-Fi (Wireless Fidelity) and Common Criteria,
- Supports 802.11n wireless standard with Multiple-input and Multiple-output (MIMO), channel bonding and packet aggregation to achieve link rates of up to 300 Megabits per second (Mbps),
- Self-forming, self-healing mesh network for always-on availability,
- Standards-based interface connectivity for devices with Ethernet or serial data interfaces:
 - o One 10/100/1000 Base-T WAN (UPLINK) Ethernet port,
 - One 10/100/1000 Base-T LAN (LOCAL) Ethernet port,
 - o Three MIMO antenna ports,
 - Supports RS-232 / 422 / 485 serial interface.
- Power: +5 to +12 Volts Direct Current (VDC).

1.2.1.1 External Interfaces

The purpose of this section is to describe the device and its identifiable parts so that the user is sufficiently familiar to interact with the physical unit. Figure 2 shows the location of the external connectors, indicator Light Emitting Diodes (LEDs) and Reset Button on the 3e–523N.



	Table 1: 3e-523N External Connectors and Indicators
Interfaces	Description
RF A, RF B, RF	For 802.11 a/g operation, connect a single antenna to RF A ; it can be mounted directly to the unit, or mounted remotely.
C Antenna Ports	For 802.11n MIMO operation, it requires the use of three antennas. Although they can be mounted directly to the unit, it is recommended that they be separated to improve MIMO performance.
UPLINK (WAN)	The Uplink port is used to connect the unit to the enterprise local network. It supports the 10/100/1000 BaseT Ethernet standard.
LOCAL (WLAN)	The Local port is used to access the device's web management Graphical User Interface (GUI). It can only access the management interface; it cannot access the wireless radio or the uplink data path. This port supports the 10/100/1000 BaseT Ethernet standard.
SERIAL – I/O	This connector is used to connect to the (optional) Serial I/O Interface Board as shown in Appendix C. Serial I/O Interface Board.
PWR IN	This connector supplies 5-12 VDC power to the unit; see Section 0 below for details.
RST	The operation of this Reset Button is described in Section 1.2.1.4 below. Note that the unit can also be reset via the Serial I/O Interface Board and the unit's Web GUI.

1.2.1.2 Power

Power is supplied to the unit via the PWR IN connector. Figure 3 shows the optional 9 VDC power supply (Desktop Power Kit), and the location of the ground connection provided on the side of the 3e-523N. A DIN-Rail Mounted Power Kit and a Power Pigtail Cable are also available with the custom power connector used on the 3e-523N – see Section 3.1, Table 17, for a listing of available accessories.



Figure 3: 3e-523N Power Supply and Ground Connections

1.2.1.3 Indicator LEDs

Figure 2 shows the location of the external connectors, indicator LEDs, and Reset Button on the 3e–523N. The LEDs and their function are described below in Table 2.

	Table 2: 3e-523N Indicator LEDs
LED	Description
Power	The Power Indicator LED will light to indicate that power is being supplied to the unit.
WAN	 The WAN LED indicates activity and uplink signal strength on the WAN. LED Off: no connection on the uplink, or the signal is very weak LED blinks slowly (every 1 second): there is a connection, and the signal quality is poor LED blinks fast: there is a connection, and the signal quality is good LED remains on: there is a connection, and the signal quality is excellent This LED is not used when the operating mode of the device is set for access point mode.
WLAN	The WLAN LED will flash to indicate activity on the WLAN. It is also used to provide status information when the 3e-523N is reset; see Section 1.2.1.4 for details.
FIPS/ALARM	The FIPS/ALARM LED indicates whether the 3e–523N is in FIPS mode. When this LED is lit, the system is in FIPS mode; when it is not lit, the system is in non-FIPS mode.

1.2.1.4 Reset (RST) Button



Figure 4: 3e-523N External Connectors and Indicators

Figure 4 shows the location of the Reset Button on the 3e–523N and Figure 107 in Appendix C. shows the location of the Reset Button on the Serial I/O Interface Board. Both buttons operate identically.

The reset button has several purposes:

- Commanding a simple reboot,
- Commanding a restore to factory default configuration.

When the reset button is first pressed, the WLAN LED will light up, providing feedback that the button has indeed been pressed. If one continues to press the button, the LED will go dark after 5 seconds, light again after 10 seconds, and will return to showing WLAN radio activity after 15 seconds. If the button is pressed for more than 15 seconds, the button press is ignored, until released and subsequently pressed again.

For a simple reboot, hold the button for slightly more than 5 seconds, specifically until the WLAN LED goes dark. Then release the button. The device will reboot, and the LEDs will indicate the normal pattern one sees just after power up.

NOTE: If you hold the button too long, you could accidentally restore to factory defaults and wipe out your device configuration. To avoid this possibility, consider resetting the unit by removing power, or through the unit's web interface.

In order to restore to the factory default, press the button, then hold for at least 10 seconds, specifically until one sees the WLAN LED come on for a second time. Then release the button. The device will return to the user selected factory default state.

1.2.2 3e-525N Access Point

The 3e–525N Access Point has the following key features:

- Highly secure AES 256, FIPS 140-2 validation AES and 802.11i,
- Dual radios with MIMO antennas with maximum data rate of 300 Mbps,
- 802.1Q Virtual Local Area Network (VLAN) allows use of multiple Service Set Identifiers (SSIDs),
- Power-over-Ethernet (PoE) standard 802.3 at compliant and 802.3 af backward compatible,
- Standards-based interfaces:
 - o One 10/100/1000 Base-T WAN (UPLINK) Ethernet port,
 - One 10/100/1000 Base-T LAN (LOCAL) Ethernet port,
 - Two 802.11n compatible wireless interfaces,
 - Six MIMO antenna ports.
- Power
 - o 3e-525N supports 48 VDC PoE on UPLINK Ethernet Port

1.2.2.1 External Interfaces

The purpose of this section is to describe the device and its identifiable parts so that the user is sufficiently familiar to interact with the physical unit. Figure 5 shows the location of the external connectors, indicator Light Emitting Diodes (LEDs) and Reset Button on the 3e–525N.



Figure 5: 3e-525N External Interfaces

	Table 3: 3e-525N External Interfaces
Interfaces	Description
Antenna Ports	Six antenna ports; Antenna ports RF1A, 1B, 1C are connected to Radio 1 Antenna ports RF2A, 2B, 2C are connected to Radio 2
Uplink (WAN Port)	The Uplink Port is used to connect the unit to the enterprise local network. This port accepts PoE compatible with 802.3at and 802.3af (PoE cable connection is illustrated in Section 2.1.2) and supports the 10/100/1000 BaseT Ethernet standard.
Local (Management Port)	The Local Port is used to access the device's web management GUI. It can only access the management interface and cannot access the wireless radio or the uplink data path. This port supports the 10/100/1000 BaseT Ethernet standard.
AUX	The AUX port includes Contact Alarm Inputs and Reset input for the 3e- 525N
Ground	Proper connection to between Ground connection and ground is essential for the device to be compliant with emission requirements and to avoid lightning damage.
Reset Screw	The screw must be unscrewed, and an un-bended paper clip can be used to press the reset button as described below. The screw should be replaced after reset; the screw is watertight and should not be unscrewed more than once. It is recommended that you use the AUX port to reset the device rather than using the reset screw.

The accessory 3e-CPLR-1 in Table 18 is used to connect with the WAN or LAN port.

The pin definition of AUX port is shown in Figure 6.



Figure 6: AUX Port Pin Definition

The functional description of AUX pins is shown in Table 4.

-	Table 4: 3e-525N /	AUX Port
Signal Name	Pin	Device
Alarm 1 +	С	3e-525N
Alarm1 -	В	3e-525N
Alarm 2 +	М	3e-525N
Alarm 2 -	Н	3e-525N
Reset	А	3e-525N
Reset Return	D	3e-525N

Connector 3e-CONN-A1 listed in Table 18 is used to connect with this AUX port on 3e-525N model. The Alarm functions are reserved for future use.

1.2.2.2 Power

3e-525N product operates with PoE, which requires the installation of a separate power injector that "injects" Direct Current (DC) into the Ethernet cable. These PoE devices should comply with 802.3at or 802.3af specifications. Standard versions of the 3e-525N product draws a maximum 14 Watts of power.

1.2.2.3 Indicator LEDs

The top panel of the unit contains a set of indicator LEDs (Figure 7) that help describe the state of various networking and connection operations. **Table 5** describes in detail what they represent.



Figure 7: Indicator LEDs

	Table 5: Indicator LED Descriptions
LED	Description
Power	The Power indicator LED informs you when the Access Point (AP) is on or off. If this light is on, the access point is on; if it is not on, the access point is off.
WAN	The WAN LED indicates the state of your connection to the organization's Ethernet LAN network. When the LED is on, the unit is connected to the network. When the

	Table 5: Indicator LED Descriptions
LED	Description
	LED blinks, Ethernet packets are being transmitted or received by the interface. When the LED is off, the AP does not have an active connection to the network.
WLAN1	When the LED is on, radio 1 is turned on. When the LED is off, radio 1 is turned off.
WLAN2	When the LED is on, radio 2 is turned on. When the LED is off, radio 2 is turned off.
Uplink SS (Signal Strength)	LED Off: Uplink is only connected through the WAN interface. Neither radio is used as an uplink bridge. The Uplink LED blinks in four rates (very slow, slow, fast, solid on) represents four levels of signal level (poor, fair, good, excellent) LED is red: The device becomes the root of mesh network.
FIPS/MODE	LED is solid green: the device is in FIPS mode. LED is yellow blinking over green: the device is in a periodic self-test. LED is red blinking: the device failed the self-test. LED is solid red: physical tempering of the device is detected.

NOTE: When the WLAN1, WLAN2, and FIPS/MODE LEDs blink simultaneously then the system is halted, it indicates that the software discovers a problem with the encryption algorithm or that the system configuration does not pass the integrity check.

1.2.2.4 Reset Function

The reset function in the AUX port or the Reset Screw has several purposes:

- Commanding a simple reboot,
- Commanding a restore to factory default configuration.

When the reset button is first pressed, the **WLAN1** LED will light up, providing feedback that the button has indeed been pressed. If one continues to depress the button, the LED will go dark after 5 seconds, light again after 10 seconds, and will return to showing WLAN radio activity after 15 seconds. If the button is pressed for more than 15 seconds, the button press is ignored, until released and subsequently pressed again.

For a simple reboot, hold the button for slightly more than 5 seconds, specifically until the **WLAN1** LED goes dark. Then release the button. The device will reboot, and the LEDs will indicate the normal pattern one sees just after power up.

NOTE: If you hold the button too long, you could accidentally restore factory defaults and wipe out your device configuration. To avoid this possibility, consider resetting the unit by removing power, or through the unit's web interface.

For a restore to factory defaults, press the button, then hold for at least 10 seconds, specifically until one sees the WLAN LED come on for the second time. Then release the button. The device will restore to the factory default state.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

1.3 Wireless and Networking Basics

Wireless networking uses electromagnetic Radio Frequency (RF) waves to transmit and receive data. Communication occurs by establishing radio links between the wireless AP and devices configured to be part of the WLAN.

1.3.1 IEEE 802.11

The WiFiProtect product family incorporates Institute of Electrical and Electronics Engineers (IEEE) 802.11 Wi-Fi standards and FIPS 140-2 security for wireless communication.

1.3.1.1 802.11a

The IEEE 802.11a standard is an extension to 802.11 which applies to WLANs and provides up to 54 Mbps in the 5 Gigahertz (GHz) band. Depending on radio design and RF channel issues, 802.11a devices can operate at rates between the 54 Mbps maximum and 6 Mbps. 802.11a uses an Orthogonal Frequency Division Multiplexing (OFDM) encoding scheme rather than Frequency-Hopping Spread Spectrum (FHSS) or Direct-Sequence Spread Spectrum (DSSS).

1.3.1.2 802.11g

The IEEE 802.11g standard is an extension to 802.11 which applies to WLANs and provides up to 54 Mbps in the 2.4 GHz band. Depending on radio design and RF channel issues, 802.11g devices can operate at rates between the 54 Mbps maximum and 6 Mbps. Much like the 802.11a standard, 802.11g uses an OFDM encoding scheme for data transmission.

802.11g is backwards-compatible with 802.11b and therefore, it is a popular component in WLAN construction. 802.11g broadens 802.11b's data rates to 54 Mbps within the 2.4 GHz band providing higher data transmission rates.

1.3.1.3 802.11n

802.11n is an extension to 802.11 that improves network throughput over 802.11a and 802.11g, with a significant increase in the maximum net data rate from 54 Mbps to 600 Mbps. 802.11n standardized support for features such as, MIMO, frame aggregation, and security improvements.

1.3.2 Wireless Network Topologies

The WiFiProtect product family supports a variety of network functions, depending on the model and configuration.

1.3.2.1 Access Point Operation

When an WiFiProtect product is used as an AP, the AP virtually connects wireless users to a host wired network. All wireless devices connected to the AP are configured on the same subnetwork as the wired network interface and can be accessed by devices on the wired network. Possible AP topologies include:

1) Standalone AP Network

a) An AP can be used as a standalone AP without any connection to a wired network. In this configuration, the AP simply provides a standalone wireless network for a group of wireless devices (Figure 8).



Figure 8: Standalone AP Network

2) Independent AP Networks

 a) Multiple APs (Figure 9) can be connected to an existing Ethernet network to bridge between the wired and wireless environments. Each AP can operate independently of the other APs on the LAN. Multiple APs can coexist as separate individual networks at the same site with a different network identifier (ID), SSID.



Figure 9: Multiple APs Connected to an Existing Ethernet Network

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

3) Integrated AP Networks

a) Integrated AP networks are the most prevalent topology choice. In this topology, multiple APs (Figure 10) are connected to a wired network and operate off of that network's Dynamic Host Configuration Protocol (DHCP) server to provide a wider coverage area for wireless devices, enabling the devices to "roam" freely about the entire site. The APs must use the same SSID.



Figure 10: Multiple APs Connected to an Existing Ethernet Network with DHCP Server

1.3.2.2 Bridging Operation

The wireless bridging function of the WiFiProtect product family supports number of bridging configurations, including the following:

- Point-to-point bridging of two Ethernet links
- Point-to-multipoint bridging of several Ethernet links (mesh)
- Dual bridge configuration to connect networks with dissimilar radio, frequency, channel, and/or SSID (see Figure 11 below).



Figure 11: Dual Bridge Configuration

1.3.2.2.1 Dual Bridge Mode Operation

In 3e-525N products, there are two radios, which can work as bridges at different frequency bands. As shown in Figure 11, the two groups of bridges can work at two different frequency bands. The bridge in the middle uses two radios and works on different frequency bands to connect the two groups together. Since the two groups are using difference frequency bands, there will be less congestion in the bridge backhaul traffic when compared to using single frequency bands for all eleven bridges.



Figure 12: Bridge Relay with Different Frequency Bands

Figure 12 shows a bridge relay using different frequency bands. The bridge throughput will not reduce at each hop. If all bridges on the relay chain are using the same frequency bands for bridge, the bridge throughput put will be reduced for each additional hop.

1.3.2.3 Client Operation

In the client mode of operation, the device acts as a wireless endpoint providing a communication link from a local device into a WLAN. Because this requires only a single radio, the dual-radio AP 3e-525N is not typically used in client mode.

1.3.3 MAC Address Filtering

The Media Access Control (MAC) address is a hardware address that uniquely identifies each node of a network. In IEEE 802 networks, the Data Link Control (DLC) layer of the Open Systems Interconnection (OSI) Reference Model is divided into two sub-layers: The Logical Link Control (LLC) layer and the MAC layer. The MAC layer interfaces directly with the network media. Consequently, each type of network media requires a unique MAC address.

The WiFiProtect product family, if set up to use MAC address filtering, detects an attempt to connect by a client and compares the client's MAC address to those on a predefined MAC address filter list. Only client addresses passing the filtering rule are allowed to associate. MAC addresses are pre-assigned by the manufacturer for each wireless card.

1.3.4 DHCP Server

The DHCP function is accessible only from the local LAN port to be used for initial configuration. This simple DHCP server is for local configuration convenience and is not accessible through wireless ports.

NOTE: Installers should not rely on this server to allocate IP addresses for other devices in their network.

1.3.5 Data Encryption and Security

Authentication mechanisms are used to authenticate an operator accessing the WiFiProtect device and to verify that the operator is authorized to assume the requested role and perform services within that role.

Bridging encryption can be established between WiFiProtect devices using AES-Cipher Block Chaining (CBC) or AES-Counter with CBC MAC (CCM) encryption (approved by the National Institute of Standards and Technology (NIST) for U.S. Government and Department of Defense (DoD) agencies).

AP encryption can utilize Wi-Fi Protected Access 2 (WPA2) or AES, depending on the mode of operation. WPA2 supports using a WPA PSK or a RADIUS Server for key management with AES-Counter-mode/CBC-MAC Protocol (CCMP).

1.3.5.1 Advanced Encryption Standard (AES)

The AES was selected by NIST in October 2000 as an upgrade from the previous Data Encryption Standard (DES). AES uses a 128-bit block cipher algorithm and encryption technique for protecting computerized information. AES has the ability to use larger 192-bit and 256-bit keys, if desired.

802.11i and WPA2 employ AES-CCM, which is a combination of AES Counter Mode (CTR) per packet data encryption, combined with AES-CBC.

AES-CBC provides per packet data integrity/authentication of the entire packet including the Message Authentication Code (MAC) header.

1.3.5.2 WPA2 / 802.11i

WPA2 utilizes IEEE 802.11i, a Layer 2 specification that focuses on strengthening IEEE 802.11 security at the MAC sublayer. It is completely separate from and independent of Virtual Private Network (VPN) designs/architectures, which are often implemented at Layer 3. IEEE 802.11i includes specifications on encryption, authentication and key management in a multi-layered approach to security. IEEE 802.1X-based authentication mechanisms are used, with AES in CCMP mode, to establish an 802.11 Robust Security Network (RSN).

IEEE 802.1X defines a framework based on the EAP over LANs, also known as EAPoL. EAPoL is used to exchange EAP messages which execute an authentication sequence and for key derivation between a Station (STA) and an EAP entity known as the Authentication Server (AS).

IEEE 802.11i defines a four-way handshake using EAPoL for key management and group key derivation. Four major categories or primary functions of 802.11i are invoked within the WiFiProtect products:

- **EAP-TLS**: Extensible Authentication Protocol Transport Layer Security is typically supported by products receiving WPA2 certification.
- IEEE 802.1X: Known as port based network access control, 802.1X provides an authentication framework within 802.11i. 802.11i depends upon 802.1X to control the flow of MAC Service Data Units (MSDUs) between the Data Server (DS) and STAs by use of the IEEE 802.1X Controlled/Uncontrolled Port model. The 802.1X Controlled Port is blocked from passing general data traffic between two STAs until an 802.1X authentication procedure completes successfully over the 802.1X Uncontrolled Port. 802.11 depends upon IEEE 802.1X and the EAPoL-Key 4-Way and Group Key Handshakes to establish and change cryptographic keys. Keys are established after authentication is complete.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

- **4-Way Handshake**: The 4-way handshake defined in 802.11i achieves the following important goals within the security protocol:
 - It confirms the Primary Master Key (PMK) between the supplicant (client) and authenticator (AP)
 - o It establishes the temporal keys to be used by the data-confidentiality protocol
 - o It authenticates the security parameters that were negotiated
 - o It provides keying material to implement the group key handshake within 802.11i
- **AES CCMP**: 802.11i and WPA2 employ AES-CCM, which is a combination of AES CTR per packet data encryption, combined with AES-CBC. AES-CBC provides per packet data integrity / authentication of the entire packet including the MAC header.

1.3.6 Ultra 3eTI Wireless VLAN

According to the IEEE, VLANs define broadcast domains in a Layer 2 network. Ultra 3eTI Wireless VLANs have the same attributes as physical LANs with the additional capability to group end stations physically to the same LAN segment regardless of the end stations' geographical location.

When wireless VLAN is enabled, an AP can be configured to have multiple SSIDs, so that it supports multiple wireless networks, with each network belonging to a VLAN. A wireless client communicates with the AP inside a wireless network defined by a SSID, so it does not know that the wireless VLAN exists. The mapping between the wireless network and the wireless VLAN happens inside the AP. Each wireless VLAN can have its own security level set. For example, the VLAN for an enterprise network access may use 802.11i with Extensible Authentication Protocol (EAP)-TLS authentication; while the VLAN for guest internet access may simply use 802.11i with Pre-Shared Key (PSK) (Figure 13).





Optionally, the user can configure the WAN Ethernet port on a specific VLAN. In this case, only traffic with this specific VLAN ID can be sent out of the WAN Ethernet port and it also remains untagged. All Ethernet traffic coming into the WAN port is automatically tagged with the configured VLAN tag.

1.4 Device Management and Administration

After initial setup, maintenance of the system and programming of security functions should be performed by personnel trained in the procedure using the embedded web-based management screens. WiFiProtect devices support two categories of users.

There are two ways of managing 3e-520 series devices: locally and remotely

Local user management is done by locally accessing the 3e-520 series GUI through the LOCAL MGMT Ethernet port. Users created in the 3e-520 series local database have the highest privilege ('3e-local' user role), no other roles are possible. See Section 1.4.2 Local User Management for more details.

Remote user authentication and authorization (Remote A&A) is centrally managed at enterprise level. The 3e-520 series device uses the Lightweight Directory Access Protocol (LDAP) to authenticate username and password against an Active Directory Server. Users can be assigned to any of three user roles: '3e-local', '3e-CryptoOfficer', and '3e-administrator'. For more information see Section 1.4.1 User Roles and 2.8 Remote Authentication and Authorization.

1.4.1 User Roles

3e-520 series devices permit only '3e-local' roles for local access users (up to ten in local user database). It allows any of three user roles for remotely authenticated and authorized users: '3e-local', '3e-CryptoOfficer', and '3e-administrator'. '3e-local' users have the highest privilege, and can perform all cryptographic configurations for the module, which include: loading digital certificates and private keys for IPsec; configuration of 801.1X supplicant; setting firewall and deep packet inspection policies; create and manage users locally; uploading new firmware and bootloader; setting the password policy and performing self-tests on demand; and performing key zeroization. A '3e-CryptoOfficer' user can perform most functions permitted to '3e-local' minus the ability to create and manage users locally and configure 3e-520 series Remote A&A settings. A '3e-administrator' user has a more reduced set of permitted functions as listed on Table 6 User Role Services below.

The following table describes the 3e-520 series device user roles services, including purpose and functions, details about the service, and user roles:

			User Role	S
Service and Purpose	Details	3e-local	3e-Crypto Officer	3e-administrator
Input of Keys	IKE v2 digital certificate private key, 802.1X supplicant private key, device HTTPS private keys, authentication key with RADIUS Server.	Х	Х	
Create and manage users locally	Support up to 10 administrator users and 5 crypto officer users.	Х		

Table 6: User Role Services

			User Role	S
Service and Purpose	Details	3e-local	3e-Crypto Officer	3e-administrator
Configuring Remote A&A	Allow to configure remote A&A parameter	Х		
Change password	Administrator changes his own password only.	Х	Х	Х
Show system status	View traffic status and systems log excluding security audit log.	Х	Х	Х
Manage audit logging	Select audit events to be logged. Configure remote audit logging. View audit event records.	Х	Х	
Key zeroization via reboot		Х	Х	Х
Factory default	Delete all configurations and set device back to factory default state.	Х	Х	
Perform Self-Test	Run algorithm KAT through reboot.	Х	Х	Х
Load New Firmware	Upload 3eTI digitally signed firmware.	Х	х	
SNMP Management	Manage all SNMP settings including SNMPv3 encryption key.	Х	Х	Х
HTTPS Management	Load HTTPS server certificate and private key.	Х	Х	
Key Generation	Create asymmetric key pairs and X509v3 Certificate Signing Request.	Х	Х	Х

Note: User/roles created locally are independent of user/roles created at Remote A&A Server.

1.4.2 Local User Management

The 3e-520 series devices have a factory default setting of only one '3e-local' user role account enabled with the username 'CryptoOfficer' and a password of 'CryptoFIPS'. When in the factory default state, these credentials can be used to access the device and allow the user to change the default credentials and add new user accounts. When reverting to the factory default state, all user-defined accounts will be deleted and will have to be recreated. The device always requires at least one '3e-local' account to be active. If the user attempts to delete the last '3e-local' account, the user will be informed that this is not possible. When successfully logging in for the first time, every new user will be instructed that his/her password has expired, and he/she will need to enter new credentials before he/she can proceed. All the local User Management functionality is available to '3e-local' over the graphical user interface under the 'User Management' subheading on the left side of the screen.

2. Device Configuration

2.1 Quick Setup

In order to begin configuring your device, you will need to perform the following steps (details in the referenced sections):

2.1.1 3e-523N

- 1) Connect power to the unit:
 - a) Use a 5-12 VDC power supply (Section 0).
- 2) Connect your laptop to the Designated Management Interface Local port:
 - a) Use a standard RJ-45 Cat6 Ethernet cable (Section 3.1).
- **3)** Log in to the unit (Section 2.2.1):
 - a) URL: https://192.168.15.1 (requires secure HTTPS, TLS 1.2),
 - b) Dismiss the certificate warning dialog box,
 - c) Default Username is "CryptoOfficer" and the default Password is "CryptoFIPS".
- 4) You can now begin configuring the unit for operation (Section 2.2.2).

2.1.2 3e-525N

- **1)** Connect power to the unit:
 - a) Connect a PoE injector using the Cat6 Ethernet/PoE cable via WAN port (Section 1.2.2.2).
- 2) Connect your laptop to the Designated Management Interface Local port:
 - a) Use a Cat6 Ethernet cable (Section 4.1).
- **3)** Log in to the unit (Section 2.2.1):
 - a) URL: https://192.168.15.1 (requires secure HTTPS, TLS 1.2),
 - b) Dismiss the certificate warning dialog box,
 - c) Default Username is "CryptoOfficer" and the default Password is "CryptoFIPS".
- 4) You can now begin configuring the unit for operation (Section 2.2.2).

Figure 14 shows the connections of 3e-525N installation. The UPLINK port is used to connect the unit to the organization's Local Area Network (LAN). For the 3e-525N, the uplink cable is run from the unit to the power injector. This POE power injector connects to Alternating Current (AC) power and provides a port to connect into the wired network. The Local Management port is used to access and manage the device's web management GUI only.



Figure 14: 3e-525N Quick Setup

2.2 Initial Device Configuration

The user can access the device's Web Management UI application either through the dedicated Local Ethernet port or the Uplink (WAN) Ethernet port. The access is identical over IP and HTTPS. For initial access to the device, the user shall use the dedicated Local Ethernet port since the WAN IP is not set up.

2.2.1 Initial Setup

The Designated Management Interface Local port on the unit connects you to the device's internal DHCP server, which will dynamically assign an IP address to your laptop so you can access the device for configuration. In order to connect properly to the unit on the Local (Management) port, the TCP/IP parameters on your laptop must be set as in Figure 15.

nternet Protocol Version 4 (TCP/I	Pv4) Properties 21
General You can get IP settings assigned aut this capability. Otherwise, you need for the appropriate IP settings.	tonatically if your network supports to ask your network administrator
C gotain an IP address automati	cally
· Use the following IP address:	35 C C C
(Pathen:	192 . 168 . 15 . 10
Sybnet mask:	255 . 255 . 255 . 0
Default gateway:	1
C Ogan Did serve address av	new Kalv
- F Use the following DHS server a	ddresses:
Breferred DNS server:	
Alternate DNS servers	· · · ·
Vajidate settings upon exit	Adgenoed
	CK Cancel

Figure 15: Local PC IP Address Configuration

Plug one end of an Ethernet cable to the Local (Management) port of the device and connect the other end to a RJ-45 Ethernet port on your laptop.

On your computer, pull up a browser window and type in the default Uniform Resource Identifier (URL) for the unit's Local LAN in the address line (https://192.168.15.1). Note that https is required to ensure a secure connection.

1) Browser Notes:

- a) SSL 2.0 must be disabled in browser configuration.
- b) The 3e-520 series product supports TLS 1.2, The recommended browser includes:
 - i) Microsoft IE version 9 and later,
 - ii) Mozilla Firefox version 27 and later,
 - iii) Google Chrome 38 and later.
- c) When Two-Factor Authentication is enabled, only Microsoft IE and Edge support CAC authentication.

A warning window appears stating "There is a problem with this website's security certificate." Select and click "Continue to this website (not recommended)" to go to the 3e-520 series login page.

)	e 🗃 https:/192168151/ 🔹 😽 🗙 🔂 Bing	
10	Certificate Error: Navigation Blocked	
3	There is a problem with this website's security certificate.	
	The security certificate presented by this website was not issued by a trusted certificate authority. The security certificate presented by this website was issued for a different website's address.	
	Security certificate problems may indicate an attempt to fool you or intercept any data you send to the server.	
	We recommend that you close this webpage and do not continue to this website.	
	Click here to dose this webpage.	
	Continue to this website (not recommended).	
	More information	

Figure 16: Certificate Error

Next, the Login window will appear.

2.2.2 Login

	Ultra Electronics, 3eTi
	3eTI AirGuard [™] Wireless Access Point
	3e-523N Version 5.1
	Usemame
	Password.
	I agree to the terms and conditions below.
	Sign bi
	Terms and Conditions:
This devic	e is for authorized use only. Any unauthorized use of this product is prohibite
Copyright	© 2020 Ultra Electronics - 3e Technologies International. All rights reserved

Figure 17: Login Screen

You will be asked for your Username and Password (Figure 18). The default Username is "CryptoOfficer" and the default Password is "CryptoFIPS" to give full access for setup configuration (Username and Password are case-sensitive). Please read the terms and conditions and check the checkbox, then click **Sign In** to continue configuration. Please note that you will be forced to change password after first login.

Utra Electronics, 3eTI 3eTI AirGuard™ Wireless Access Point 3e-523N Version 5.1	
Usensame: CryptoOfficer Password: +++++++++ 22 Tagree to the terms and conditions below.	
Terms and Conditions: This device is for authorized use only. Any unauthorized use of this product is prohibite	đ.
Copyright © 2020 Ultra Electronics - 3e Technologies International. All rights reserved	L

Figure 18: Login Screen with Defaults Username/Password

2.3 System Configuration

2.3.1 Configuration Information

For the initial configuration, the network administrator may need the following information:

- IP address a list of IP addresses available on the organization's LAN that are available to use for assignment to the devices,
- Subnet Mask for the Uplink network,



- Default IP address of the Local Management port (https://192.168.15.1),
- DNS IP address,
- The MAC addresses of all wireless cards that will be used to access the network,
- The appropriate AES encryption key,
- SSID to identify all members of the WLAN.



Figure 19: Internet Explorer Successful Login Web Page

2.3.2 General



Figure 20: System Configuration – General

You will immediately be directed to the **System Configuration – General** screen (Figure 20). This screen lists the 3e-520 series firmware version number for your unit and allows you to enter a **Host Name, Domain Name,** and **Description** (a description of the physical location of the unit is useful when deploying units to remote locations). Defaults are "default location", "default", and "default" respectively.

You should also select the appropriate **Time Zone** for your location.

NOTE: Only the user with 3e-local and 3e-CryptoOfficer role can set the date and time.

The 3e-520 series is equipped with a real time clock that keeps System Time. This clock can be set manually or can be synchronized with an external time source using the Network Time Protocol (NTP). Set the System Time by selecting the appropriate Time Source radio button on the GUI.

If the System Time is incorrect, you need to modify the **Time Source** (default is Manual):

• **Manual:** Please enter the correct date in MM/DD/YYYY format, and the correct time in 24-hour HH:MM format. The system date must be set to a date later than 01/01/2005.

System Time:	Date: 03/11/2020 Time:11:41			
Time Source:	Manual	O NTP Server	O IEEE 1588 Master	
	New Date: New Time:		/ (Month / Day / Year) (Hour:Minute)	
Time Service:	☐ Make this	device the IEEE 1	1588 master	

Figure 21: Manual Time Source

• NTP Server: Please enter at least one valid IPv4 or IPv6 address; examples are provided for your convenience. The correct time must be set up for X509v3 certificates to work properly, because certificates have a valid time frame associated with them.

The device supports the NTPv3 protocol. An MD5 HMAC key can be provided when using NTPv3 in order to authenticate each NTP packet. Select the 'NTP Server' in order to synchronize with an external time source.
10000	3eTI AirQuard	Wireless Access Pol	nt	100M		
Ultra 3eTI	Usemane: C Role: 3	kyploOfficer e-local	Hust Neres Radio Rogion	default United States		
Second Sec	System Configuration -> General					
Energy Revenue WAN LAN	Version Description	3e 523N Version 5.1 default location				
Bridge Tithemet VLAN MAC Address Filtoring	Domain Name	default	_			
Reibi 1 WLAV-Model Client PHY Setting	Time Zune	(GMT-05:00) Eeste	m Time (US & Canada)	N		
Client Ceneral Client Seculty	System Tane	Date: 03/25/2020 T	me.11:19			
Free Tannel Profiles Solution	Time Source:	O Manual 🔍 NTI	P Server O IEEE 1588 Master			
Services Settings DHCP Server BNMP Againt Remote Administration Web Gener		Time Server 1 Cools Server IP	192.168.200.101 2 Key Numb	with MOS Key 🖌		
Admin Usor Management Lad Al Usen Add New User User Login Policy Burson A&A Serup		IPset Tunnel	Rex Value Profile Server1	Specity (Poet 1Ds		
Tao Factor Arth Wonitoring/Reports System Status Bridging Status Bridging Status Bridging Status Dirt Clarkt Lint		Server IP IPsec Turnel Time Server 3 Config	192 168,206,23	with MD5 key 🗌		
Linga Bysiam Log Web Access Log		Server IP IPset Tunnel		with MD5 key		
Auditing Audit Configuration Audit Log		NIST servers	tine-a rist.gov, time-b.nest.g	os, lime mst.gov		
Ernal Notification Cont Radio Tx Of Control System Upgrace Default Configuration Benote Logging		NTP Server 3P Addr 1 132 16 2. 192 16 3	ess 8 200 101 8 205 23	Statu		
Reform On Demand Self-test	Time Service	Make this device	the IEEE 1508 master			

Figure 22: NTP Time Source

While the NTP protocol provides an authentication mechanism, it does not provide for confidentiality. Therefore, the device can be configured to tunnel all NTP packets through an IPsec tunnel. Select the "IPsec Tunnel" check box and provide a "Tunnel Profile" in order to protect time synchronization packets with IPsec.

System Time:	Date: 03/25/2020 Time:11:19						
Time Source:	O Manual NTP Server O IEEE 1588 Master						
	Time Server 1 Configuration						
	Server IP: 192.168.200.101 with MD5 key						
	2 Key Number						
	Key Value						
	Part Turnel Dodle Sever1 V Specify Deer IDe						
	abertante de Prote Dertant						
	Time Server 2 Configuration						
	Server IP: 192.168.205.23 with MD6 key						
	IPsec Tunnel						
	Time Server 3 Configuration						
	Server IP: with MD5 key						
	IPsec Tunnel						
	NIST servers: time-a.nist.gov, time-b.nist.gov, time.nist.gov						
	NTP Server IP Address Status						
	1. 192.168.200.101						
	2. 192.168.205.23						
L	3.						

Figure 23: NTP Time Source with IPsec Protection

Note: IPsec Tunnel Profiles must be configured before IPsec can be used to protect NTP packets. See Section 0 in order to configure an IPsec Tunnel Profile.

Note: Compliance with Common Criteria (PPWLAN) requires that all NTP traffic be transmitted within IPsec tunnels.

The 3e-520 series initiates the IPsec protocol when building and IPsec tunnel to an NTP server. During the authentication phase of the protocol, the 3e-520 series (Initiator) will specify to which of the NTP server's (Responder's) identities it wants to communicate with. Most NTP servers will have one IPsec identity bound to the NTP server's IP address. By default, the 3e-520 series will talk to this identity. There are instances where an NTP server may host multiple IPsec identities. In this case, the desired identity can be specified by checking the "Specify IPsec IDs" check box and supplying an IPsec identity.

Note: When selecting an IPsec profile that uses Public Key authentication, the 3e-520 series will validate the supplied IPsec ID against the NTP server's X509v3 certificate. The NTP server's X509v3 certificate must contain the identity as either the subject or one of the subject alternative names.

• From IEEE 1588 Master:

You can select a **Time Source** to be synchronized to an IEEE 1588 Master when one of the 3e-520 series device is designated as the IEEE 1588 Master (Figure 24: **IEEE 1588 Master Time Source & Login Banner**

). IEEE 1588 Time Synchronization is a standard protocol for synchronizing clocks connected via a network to provide fault-tolerant synchronization among heterogeneous networked clocks.

100	3eTI AirGuai	d TH Wireless Access Point			(Lugiur
Ultra 3eTi	Usemame:	CryptoOfficer	Host Name:	default	
ELECTROPICS	Role	3e-local	Radio Region:	United States	
Surban Configuration	System Cont	figuration -> General			-
General Notay Channel Control Wath	Version Description:	3e-523NF Version 5.1 default location			
LAN	Host Name	defeult			
Ethemat VLAN MAC Address Filtering	Domain Name:	default			
Certificate Store Radio 1 WLAN Mode: AP	Time Zone:	(GMT-05:00) Eastern Time (US & Canada)	Y	
PHY Satting AP General AD Security	System Time:	Date: 04/02/2020 Time:15:3		-	
AP Wireless Clients Wreless VLAN Napping	Time Source	Manual O NTP Server	O IEEE 1588 Master		
Piece Tunnal Ptofikis Status		New Date /	/ (Month / Day	/ Year)	
Services Settings DHCP Server SMMP Agent Serval Port	Time Service	New Time:	(Hour Minute 588 master	9)	
Serual Communication Remote Administration Web Server	Login Bannet:		100		-
Admin User Nanagement List All Users Add New User User Login Policy Remote A&A Setup Tayo Factor Auth					
Monitoriag/Reports System Status Bridging Status Bridging Status Bridging Site Map Adjacant AP List Dirch Church Let		Maximum 1400 characters Mini	num 5 characters. If the	field is left blank, the system of	v
Logs System Log Wab Access Log	Apply	Current number of characters		~ 1A	

Figure 24: IEEE 1588 Master Time Source & Login Banner

You can modify the terms and conditions in the **Login Banner** which is displayed on the Login screen. The default is **"This device is for authorized use only. Any unauthorized use of this product is prohibited."**

When you are satisfied with your changes, click Apply.

2.3.3 Noisy Channel Control

The **System Configuration – Noisy Channel Control** screen (Figure 25) allows the installer the ability to manually eliminate channels from the 802.11 channel list. It provides a convenient listing of recent radar activity for each channel to assist the user. However, we strongly recommend that an RF spectrum site survey be performed to determine potentially noisy or radar-affected channels that should be added to this list.

The channels selected on this screen will not be available for selection when configuring radio channels. Some channels may not be available for selection if Dynamic Frequency Selection (DFS) is required.

NOTE: This noisy channel list persists until it is explicitly removed from the **System Configuration – Noisy Channel Control** screen. The list will not be cleared automatically, even if the device is reset to the "Factory Default" configuration. This configuration applies to the system rather than each radio.

Click **Apply** to confirm your selection.

Ultra	JOIT ANGU	NO WITCHESS ACC	saa Point		C PAULTY A
ILICTIONICS 3eT	Username:	CryptoOfficer		Host Name:	default
	Role:	3e-local		Radio Region:	United States
tem Configuration	System Co	nfiguration -> Nois	y Channel Con	trol	
eneral		Channel	Noisy?	DFS Required	Radar Detected (seconds ago)
orea commune commune	1	(2412 MHz)			
N	2	(2417 MHz)			
ternet VLAN	3	(2422 MHz)			
C Address Filtering dificate Store		(2427 MHz)			
01	6	(2432 MHz)			
Are wrote: AP I'r Setting	6	(2437 MHz)			
General	2	(2442 MHz)			
Wroless Clients	8	(2447 MHz)			
reless VLAN Mapping	- 5	(2452 MHz)			
dies	4	0 (2457 MHz)			
ces Settings	1	1 (2462 MHz)			
ICP Server	3	6 (5180 MHz)			
næ Port	4	0 (5200 MHz)			
nal Communication	4	4 (6220 MHz)			
b Server	4	8 (8240 MHz)			
n User Management t All Users	5	2 (5260 MHz)		~	
d Now User or Loom Policy		5 /5280 8467)		1	
mine A&A Setup		o (ozoo nii z)			
toringReports	6	0 (5300 MHz)		×	
slam Status dono Status	10	0 (5500 MHz)		~	
dging Site Map	10	4 (5520 MHz)		2	
CP Client List	40	0 (5540 Miles)		1	
tern-Log		o (ande milie)	13		
b Access Log	11	2 (0560 MHz)		×	
dit Configuration	11	6 (5580 MHz)		V	
m Administration	13	2 (5660 MHz)		2	
Email Notification Conf Radio Tx Off Control System Upgrade Default Configuration Remote Logging		R ISERD MILLIN	-		
		0.05345.5554	-		
	- 19	9 (3743 MP12)			
Demand Self-test	10				
nodic Self lest	10				
łp	16	((DOGO MIFIZ)			

Figure 25: System Configuration – Noisy Channel Control

2.3.4 WAN

The System Configuration – WAN screen is shown in Figure 26.

You can select the specific link speed and duplex configuration of the **WAN/LAN Link**. By default, this will be automatically negotiated.

If using an **IEEE 802.1Q VLAN tag** on this port, please enter its value; a value of 0 indicates no VLAN support and is configured by default.

2.3.4.1 IPv4 Address

Unless you are using DHCP to obtain an IP address, you should **Specify a static IP address** in order for the device to be managed from the wired LAN. You will need to specify **IP Address**, **Subnet Mask**, **Default Gateway**, and if necessary, Domain Name System (DNS) 1 and DNS 2. Defaults are 192.168.254.254, 255.255.255.0, 192.168.254.1 respectively (with DNS fields blank).

NOTE: After changing the static network address you will no longer be able to access the device from this port using the default IP address. In order to log into the unit through the WAN port, you will need to change the browser URL to reflect the new IP address and log in again.

NOTE: If **Using DHCP to obtain an IP address** is selected, a new IP address will be automatically assigned to the WAN port after clicking **Apply**. In order to log into the unit through the WAN port, the new IP address needs to be obtained. Either contact your Network Administrator, or set up "Remote Logging" (see Section 2.13.5) before configuring the WAN to use DHCP.

Click **Apply** to accept changes.

2.3.4.2 IPv6 Address

IPv6 addresses can be configured disabled, automatically or statically. By default, the IPv6 Addresses is disabled. If **Automatic Configuration** is selected, you can choose to use **Stateless Autoconfiguration** or **DHCPv6**. If you select **Stateless Autoconfiguration**, the device will attempt to configure itself automatically when connected to a routed IPv6 network using Internet Control Message Protocol version 6 (ICMPv6) router discovery messages.

NOTE: A 64-bit prefix is required for this configuration.

If IPv6 stateless address autoconfiguration is unsuitable for an application, a network may use stateful configuration with the Dynamic Host Configuration Protocol version 6 (**DHCPv6**).

If Static Configuration is selected, you will need to specify IPv6 Address, Default Gateway, and if necessary, DNS 1 and DNS 2. Enter all IPv6 addresses in the standard format of eight groups of four hexadecimal digits separated by colons. An IPv6 address can also be abbreviated with the following standard rules:

- Omit leading zeroes in a 16-bit value.
- Replace one group of consecutive zeroes by a double colon.

Enter a **Prefix Length** appropriate to your application (see RFC 6177 for guidance). It is set to a value of 48 by default.

Click **Apply** to accept changes.

Free Control of Contro	3eTI AirGu	ard™ Wireless Access Point			Logun
Ultra 3eTI	Usemame:	CryptoOfficer	Host Name:	default	
ELECTRONICS	Role:	3e-lotal	Radio Region:	United States	
	System Co	nfiguration -> WAN			
General General	Link Speed a	nd Duales			
WAN	WAN/LAN LIN		Au	uto 🔽	
Bridge	Property and and				
MAC Address Filtering	Hanagement	/LAN teg	0	(Range 0.d	0940
Radio 1	1177551247932		Not	te: Enter 0 for "No 802.10) header"
WLAN Mode: AP PHY Setting	Trends and the second	4			
AP Deneral AP Sucurity	1PV4 Address				
AP Winkess Clents Wreless VLAN Mapping	O Using DH	CP to optain an IP appress			
IPsec Tunnel	Specify a	static IP address			
Status	IP Address	r.	19	2 168 254	254
Services Settings DHCP Server	Subnet Ma	isk:	25	5 255 255	0
SMMP Agent	Default Ge	teway:	19	2 168 254	. 1
Senal Communication	DNS 1:				
Hartiche Agerieration Web Server	ONS 2:				
Admin User Management	01				
Add New User	IPv6 Address				
User Login Policy Remote A&A Setup	Default Ga	teway:			
Two-Factor Auth	ONS 2				
System Status					
Endging Status Endging Site Map	@ 10.0 DW				
Adjacent AP List DHCP Client List	@ IPV6 Disa	Dieg			
Logs Sustant on	O Automatic	Configuration			
Web Access Log	Statel	ess Autoconfiguration			
Auditing Audit Configuration Audit Los	Онсе	V6			
System Administration	O Static Cor	flouration			
Radio Tx Off Control	IPv6 Addre	155			
System Upgrade Befault Configuration	Prefix Len	gth (0-128):	64		

Figure 26: System Configuration – WAN

2.3.5 LAN (Local Management)

The **System Configuration – LAN** (Local Management) screen (Figure 27) configures the **IPv4 Address** and **Subnet Mask** for the Local Management port, which provides local access for configuration (defaults are 192.168.15.1 and 255.255.255.0 respectively). It is not advisable to change the private LAN address during initial setup, or any other time that you wish to remain connected to the Local LAN.



WARNING: The default value for the LAN IP address is 192.168.15.1. Normally this value should not be changed. If the value is changed and forgotten, the user will lose access to the device through the Local Management port.



100		3eTI AirGua	rd™ Wireless	Acces	s Point	9			
Ulfra ELECTRONICS 3eT		Username: Role:	CryptoOfficer 3e-local					Host Name: Radio Region:	default United States
System Configurati	ion	System Co	nfiguration ->	LAN		2.5			
General Noisy Channel Co	Introl	IPv4 Address	,	192	. 168	. 15	. 1		
WAN LAN		Subnet Mask		255	. 255	. 255	. 0		
Bridge Ethernet VLAN MAC Address Filte Certificate Store	anng	Apply							
Radio 1 WLAN Mode: AP									

Figure 27: System Configuration – LAN (Local Management)

2.3.6 Bridge

RSTP (Rapid Spanning Tree Protocol) requires each bridge in the network to be assign with a priority index. The bridge with lowest priority index will be assigned as root after the tree topology converges.

The **System Configuration – Bridge** screen (Figure 28) allows you to assign bridge priority to the device. The device's **Bridge Priority** is set to a midrange value (32768) by default. For example, the intended root bridge is assigned a priority of 100, while the rest of mash bridges holds the default priority of 32768.

100	3eTi AirGu	ard [™] Wireless A	ccess Point			Local D
ELECTRONECS 3eT1	Username: Role:	CryptoOfficer 3e-local		Host Name: Radio Region:	default United States	
Bystem Configuration General Noisy Channel Control WAN Entyment VLAN MAC Address Filturing Certificate Store Redio 1 WLAN Motio: AP PHY Setting AP General AP Security AP Security AP Security	System Co Spanning Tre Bridge Priority Apply	nfiguration -> Br e Protocol:	tidge Enable ☑ 32768 (Rønge: 0-6553	5. Default: 3276	8)	

Figure 28: System Configuration – Bridge Priority

2.3.7 Ethernet VLAN

By clicking the link **Ethernet VLAN** under System Configuration, Ethernet VLAN configuration screen will show up. Figure 29 shows the Configuration Screen for Ethernet VLAN configuration.

Ultra 3eTI	1	3eTI AirGu	ard [™] Wireless Access Poln	t		Lager
	Usemame: Role:	CryptoOfficer 3e-local	Host N Radio F	ime: default Tegion: United States		
System Configurati General Noisy Channel Co YiAN LAN Bickou Enternet VLAN MCC Address Film Cottlocale Store Radio 1 WLAN Intole AP PHY Solfoo	ion. mitol	System Co Ethernet WAN	nfiguration -> Ethernet VLA Truni	N N	Note: Enter 0 fo Effective only in	(Range 0-4094) "No 802 10 header" Single VLAN Mode.



The WAN Ethernet interface can be configured as VLAN truck or a single VLAN in 3e-520 series as shown in Figure 29. In 3e-523N series, the WAN Ethernet interface is always in VLAN trunk mode. In trunk mode, packets are sent and received unmodified.

2.3.8 MAC Address Filtering

The **System Configuration – MAC Address Filtering** screen (Figure 30) is used to configure MAC address filtering for the unit.

A STATE OF THE OWNER	3eTI AirGuard™ Wireless Access Point	Logovi
Ultro 3eTI	Username: CryptoOfficer	Host Name: dafault
ture more so a so	Role: 3e-local	Radio Region: United States
System Configuration	System Configuration -> MAC Address F	iltering
General Noay Channel Control WAN LAN Bridge Fiberated VLAN MAC Address Fibering	MAC Address Filtering Policy for Wireless Access MAC Address Filtering O Enable Default Access Policy: Time of Day Filtering: C Enable Disable	Point and Auto Bridge
Constructs State Constructs State Redie 1 WLAN Mode: AP PhY Sating AP General AP Sociality AP Wreless Clents Wreless Clents Wreless VLAN Mapping Profiles State State State State State Shaft Scrief Port Scrief Port Scrief Port Scrief Port Scrief Port Scrief Port	Add MAC Address MAC Address Radio Card: Radio 1 WLAN Service Wireless AP Note: Piter List: Writelist O Blecklist Add Whitelist (Access from/to these MAC addresses a Dekte: MAC Address Radio Card WLAN S	Import From CSV File Choose a file to import Export Export To CSV File Export Export ervice Note

Figure 30: System Configuration – MAC Address Filtering

If **MAC Address Filtering** is Enabled (it is disabled by default) and the **Default Access Policy** is **Deny All** (default), only those devices with a MAC address, which has been entered in the MAC Address listing will be able to communicate with the AP. To configure this **Whitelist**, input the MAC addresses of all the devices that WILL be authorized to access this AP with the **Default Access Policy** set to **Deny All**.

If **MAC Address Filtering** is Enabled and the **Default Access Policy** is **Accept All**, those devices with a MAC address which has been entered in the MAC Address listing will NOT be able to communicate with the AP. To configure this **Blacklist**, input the MAC addresses of all the devices that will NOT be authorized to access this AP with the **Default Access Policy** set to **Accept All**. Figure 30 depicts a Blacklist where all clients with proper credentials can connect to the AP except the device with MAC address 000B6B560C1F.

To **Add MAC Address** to the filtered list, enter the address in standard IPv4 or IPv6 format. You can select the applicable radio card(s) and WLAN service, then add a note if desired to help document the listing. Each of these settings can be independently configured for each listed address, and is displayed in the list as shown in Figure 30.

Table 7: Add MAC Address Settings					
Settings	Options	Description	Default		
Radio Card	All Radio 1 Radio 2	Select the applicable radio(s) for Mac filtering list item.	Radio 1		
WLAN Service	Wireless AP Wireless Bridge	Select the applicable WLAN Service Wireless Bridge is available only if All radios are selected for filtering.	Wireless AP		

Click Add to include the selected MAC Address on the list.

You can enter **Whitelist** (default) and **Blacklist** addresses by setting the **Default Access Policy** accordingly while entering each MAC address.

NOTE: that only one list will be active and visible at any time, as determined by the current **Default Access Policy**. If both lists exist, you can select which list to view using the **Filter List** selection.

You may find it useful to go to the **Radio 1** (and/or **Radio 2**) - **AP Wireless Clients** screen and copy the MAC address of selected Wireless Clients, then paste them into the **System Configuration – MAC Address Filtering** screen (Figure 30).

You also have the option to **Import From CSV File** – browse to select a file with MAC Address, Radio Card, WLAN Service and Note fields to be added to the displayed list. At any time, you can also export the displayed list by clicking on the **Export To CSV File** button.

By default, MAC filtering lists are applied 24 hours a day, 7 days a week. By enabling 'Time of Day Filtering', the device will apply the MAC filtering list only during the specified times.

Enable Time of Day Filtering: By enabling Time of Day Filtering, you can select the Day of the week, start and End time to apply the MAC address Filtering rule for Blacklist or Whitelist.

In Figure 31, the Blacklist filter is only applied on Friday. The Start Time and End Time are represented in Coordinated Universal Time (UTC). In the example configuration, the wireless client with MAC address 000B6B560C1F can connect to the AP every day of the week except on Friday.

10000	3eTI AirGuard™ Wireless Access Point		Lugan
Olfree 3eTI	Username: CryptoOfficer	Host Name: default	
ELECTIONICS	Role: 3e-local	Radio Region: United States	
Soutian Confirmation	System Configuration -> MAC Address Filtering	3	
General Native Channel Control	HAC Address Filtering Policy for Wireless Access Point a	ad Auta Bridge	5
WAN	MAC Address Filtering:		
LAN Bridge	Default Access Policy: Deny All		
Etternist M AN	Time of Day Filtering.		
Latting and Store	Day of the Week: Sunday Monday Tuesd	ay 🗆 Wednesday 🗋 Thursday 🗹 Friday 🗋 Saturday	£
Radio 1 Wi AN Mode: AP	Start Time Hours 00 Minutes 00 Se	econds: 00	
PHY Setting	End Time: Hours 23 Minutes 59 Sr	conds 59	
AP General AP Security	Apply		
AP Wreless Clients	Add MAC Address.	Import From CSV File	
Wreless VLAN Mapping	MAG	Choose a file to import	
Profiles	Address:	Browse .	
Status Sarvicas Settings	Radio Calo Hadio I	Import	
DHCP Server	Service Windloss AP	Export To CSV File	8
SNMP Agent Remote Administration	Note:	Execut	
Web Server	Filter List @WhiteJust @Disceller	CAPAIL	
Admin User Management	Add		
Add New Liter	Whitelist (Access from/to these MAC addresses are allow	(ed)	
User Login Policy Remote A&A Setup	Delete MAC Address Radio Card WLAN Service	hote	
Teo-Factor Auth	+ 000B6B560C1F Radio 1 Wireless AP	1072	
Monitoring/Reports	La a reconcersory contest, consects,		
Bridging Status			
Bridging Site Map advant AP List			
DHCP Client Lat			

Figure 31: Enable Time of Day for MAC Address Filtering

The Time of Day Filtering works in conjunction with the Access Point Client Session LifeTime (see Section 2.4.4 for details on setting the Client Session LifeTime). When a client is successfully associated with the Access Point, it is afforded the session lifetime independent of the time of day filtering. Only when the session lifetime has expired will the client be asked to re-associate and at that point the time of day filtering will be applied.

2.3.9 Certificate Store

The 3e-520 series devices use X.509v3 certificates as defined by RFC 5280 to support authentication for IPsec VPN, Remote User Authentication and web server applications.

Figure 32 shows the **System Configuration – Certificate Store** screen. All certificates used in the 3e-520 series products are centrally managed via the Certificate Store webpage, the certificate store which handles the following certificates:

- Certificate/key for web server https,
- Certificate/key for IPsec credential,
- Intermediate and trusted root certificate to verify peer certificates, e.g., radius server, IPsec peer, LDAP server.

The 3e-520 series devices generate its own public and private key pairs and create a Certificate Signing request (CSR) to apply for a certificate from the Certificate Authority (CA). The certificate returned from the CA can be installed on the device over the management interfaces.

The 3e-520 series devices utilize 3eTI FIPS certified OpenSSL module to generate a CSR with the common identify information (organization name, common name, locality, and country, etc.) and other device-

specific information such as MAC address and IP address when applicable. All CSRs and their associated private keys are centrally stored and managed in the device KeyStore.

If the user is deploying a 3e-520 series device into a system that already has a public key infrastructure and a certificate authority that can only accept CSRs, then the 3e-520 series device can be used to generate the required CSR. The 3e-520 series device produces a PKCS#10 compliant CSR in Base64 encoding.

Contraction of the Contraction o	3eTI AirGuard	Wireless Acc	ess Point			Lunost
Ulfra 3eTi	Usernamie: Role:	CryptsOfficer 3e-local		Host Name: Radio Region:	default United States	
System Configuration Ceneral Noisy Channal Control WAN LAN Bridge	System Config Device CSR	puration -> Certi	ficate Store	Root CA L	ocal CRI, DCSP	Responder
MAC Actives Elemang Centils de Store MLAN Mode: AP Phrty Setting AP General AP Security AP Writess Cients Windows U AN Mannen	Combined Certifica Private Key Passw	ite i Private Køy File ord:	(*.pem, *.pfx, *p12)			Browse.
Profiles Status	Device Certificat	Issued To	issued By		Valid From	Valid To
Services Settings DHCP Server SNMP Agent Serial Port Serial Port Bend Communication Remote Administration Web Server	Certificate Notifi Notify: Email Notification:	cation 60	days before cert expl	res and every 7	days thereafter.	12
Admin User Management List All Dears Add New User User Login Policy Remote A&A Satap Teo-Factor Auth Monitorina/Resports	Email Address		navie (Cr. Disable		5)	

Figure 32: System Configuration – Certificate Store

2.3.9.1 Device CSR

To generate a CSR, click on the '**Device CSR**' tab under the '**Certificate Store**' sub-heading on the left side of the screen. Fill in the appropriate details and click '**Apply**'

A CSR is generated by an applicant when applying for a digital identity certificate to a Certificate Authority (CA). The 3e-520 series device generates a block of encoded text containing the information that will be included in a certificate such as organization name, common name (domain name), locality, and country. It also contains the public key that will be included in the certificate. A private key is also created at the time the CSR is generated; this key can be exported separately but will not be included in the CSR itself.

A PKI's certificate authority will be able to sign the CSR, which can then be loaded back into the 3e-520 series device, along with the associated private key.

		3eTI AirGuard	Wireless Ac	cess Point				Logaut
Ultra ELECTRONICS	3eTI	Usemane: Role:	CryptoOfficer 3e-local		Host Nam Radio Reg	e del jon: Un	fault ited States	
Bystem Configuratio General Noisy Channel Cor	stem Coofiguration General Noisy Channel Control		Device Certs	ificate Store	Root CA	Local CR	L OCSP Respor	ndar
Invive LAN Battge Ethernor VLAN <u>HAC Address Ether</u> Cartheate Store Addia 1 10LAN Mode AP PHY Sattang AP Genungi AP Genug	ng s pping	Fill in the followin Common Name: Organization: Department: City: State/Province: County:	ıg fields to penera	te a certificate signing	request,			
Services Settings DIECP Service SNNP-Again Benal Polt Senal Communicat Nomotise Administra With Service Admin User Manage List All Users Add New User List All Users Add New User User Lagin Policy Remote AdA Seting	ton ton ment	Key Size Passphrase: Confirm Passphras		© RSA OECDSA 2048⊻]		
Monitoring/Reports System Status		No CSR to be sign	this device to be ed	signed				

Figure 33: Certificate Store Device CSR

Note: The CSR feature will rarely be used since most institutions will load their own certificates generated by their CAs.

Note: The following characters are not allowed in the Common Name field:

<>~!@#\$%^*/()? ,&

Characters $<>\sim!@#$ \$%^*/()?.,& are not allowed in the Organization, Department, City, State/province, and Country fields.

Common Name: Enter a fully qualified domain name.

Organization: Full legal name of your organization.

Department: Name of department.

City: The city where organization is located.

State/Province: The state or province where your organization is located. This should not be abbreviated.

Country: The two-letter ISO code for the country where your organization is located (e.g., US, GB, FR, etc.).

Key Type: Select from RSA or ECDSA options.

Key Size: Select from 2048 or 4096 options.

Passphrase: Enter an alphanumeric passphrase (0-128 characters long).

Confirm Passphrase: re-enter passphrase.

Once all information has been entered correctly on the CSR page, press the 'Generate CSR' button. The new CSR will be displayed under the 'CSR generated on this device to be signed' section of the screen. It

will display confirmation of CSR generation; from here the user is offered options to 'View CSR', 'Export CSR', and 'Delete CSR', as well as an option to upload a signed certificate.

Ultra ELECTRONICS	3eTI	Usemane: Rolei	CryptoOfficer Jellocal		Host Nam Hadio Reg	e: defaut gios: United	t States	
a Confourati	and .	System Confi	guration -> Cert	ificate Store				
norsti ay Charnel Co W	ntol	Device CSR	Device Certs	Intermediate CA	Boot CA	Local CRI	OCSP Responder	
lger		Fill in the follow	ing fields to genera	te a certificate signing	request.			
Address Fills	010	Common Name			and an excite	12		
icale Store :	100	Organization:		-		÷.		
N Mode: AP		Department				1		
Secural Secural		Department.		-				
Vicelans Clari		Chata: Decision						
inen VLAN M:	sebend	State/Province:						
las		Country:		ware a success				
us Settinge		Key Type		O RSA O ECDSA	81 - E			
P Server		Key Size:		2048 🗸				
note Administr	apos	Passphrase:						
b Server User Manag	ement	Confirm Passphra	ise:	1				
Al Usen		201100000000000000000000000000000000000						
r Login Policy note A&A Setu 5 Factor Auth	ø	Generate CSR						
oring/Reports								
Iging Status		CSR generated o	a this device to be:	signed		74		
iging Site Map acent AP Liel		Common Name	www.utra-	3et con				
CP Count List		Organization	Ultra Elect	ronics SeTI				
ters Log		Department	Engineerin	9				
Access Log		City:	Germanto	WTI				
it Configurate		State/Province:	Maryland					
n Administra	tion	Country	US					
all Notification	Cent	Кеу Туре	RSA					
idio Ts Off Dontrol alium Upgradie Raut Configuration mole Logging		Key Size:	2048					
		Passphrase:						
boot Demand Salf-t Rock: Self-test	12921	Vev CSR	Export CSR	Delete CSR				

Figure 34: CSR Generation

2.3.9.2 Device Certs

The Device Certs page allows users to upload combined certificate/private key to the 3e-520 series device. The certificate/private key file must be in the *.pem, *.pfx or *.p12 formats. The page lists certificates uploaded to the 3e-520 series device. To view a list of certificates in store click on the Device Certs tab in the Certificate Store page.

This page also allows user to view the contents of a certificate or delete a particular certificate altogether.

You can also enable the Email Notification to notify CryptoOfficer in the email address with preset days before the certificate is expired.

100	3eTI AirGua	rd ** Wireless Acces	ss Point				Logout	
GECTIONICS 3eTI	Usemante:	CryptoOfficer Se local	1	lost Name: India Rasiana	default United States			
ystum Configuration General Noisy Channel Control WWN	System Con	figuration -> Certific	cate Store	ot CA Local		- Select th	ne certificate fi	
LAN Bridge Ethermet VILAN WCC Additione Ethering : Certificate Store dito 1	Combined Centr Private Key Pas	Icate / Private Key File (*, sword.	pem, * ptx, *p12)	+	<i>.</i>	Browse		
ALAN Mode: AP PHY Setting AP General AP Security AP Writess Clents	Upland	Click uplo	bed			Enter p	private k <mark>ey</mark> pas	
ec Tunnel	Device Certific	ates	leaded a	100	Valid From	Ve State		
Status	USMan	Fand Littra Seti Testion	US Maruland Rockville	Iltra-Seli Testino	Aug 1 21 30 03	14 27 21 30 03	View	
vices Settings IHCP Server	1 server	1 server1@3et.com	rootca chaoxing ling	ultra-3et.com	2016 GMIT	2036 GMT	Delete	
NMP Agent emote Administration feb Server	2 US Mary device	land Ultra 3eti Testing 1 device1@3eti.com	US Maryland Rockville rootca chaoxing ling	Ultra-Seti Testing ultra-Seti com	Aug 1 21:10:22 2016 GMT	Jul 27 21 18:22 2035 GMT	View	
nin User Management int All Users udd New User Jeer Logan Policy Secolo & Schen			Click to view or delete certificate					
eo-Factor Auth	Certificate Not	incation	date balances and available an		the Presentation			
nitoring/Reports system Status intging Status sittging Site Map diacere AP Lat IHCP Cient List	Email Notificatio	n: OEna F	says berore cett expires an ble	a every 17 a	ays percaller			
gs System Log Web Access Log	Apply							

Figure 35: Device Certs

2.3.9.3 Intermediate CA

The Intermediate CA page allows users to upload Intermediate CA certificates to the 3e-520 series device. The certificates can be in the *.pem, or *.der format. The page lists certificates uploaded to the 3e-520 series device. To view a list of certificates in store click on the Intermediate CA tab in the Certificate Store page.

This page also allows user to view the contents of a certificate or delete a particular certificate altogether.

	1	3eTI AirG	3eTi AirGuard™ Wireless Access Point Le							
ELECTRONICS	3eTI	Usernama: Role:	GryptoOtficer 3e-local		Host Name: Radio Region:	default United States				
stem Configuration		System C	onfiguration -> Certil	ficate Store						
Veneral Neisy Channel Col NAN	latin	Device 0	CSR Device Carts	Intermediate CA	Root CA Local	CRL DCSP R	ospondar			
JAN Bindge						-	Select the ce	ertificate		
Thernet VLAN		1000000000000		5.0.			Bununn			
Inc Add on Elle Dertificate Store dio 1 VLAN Mode: AP INY Setting	ing	Upkad y	CA Certificates (* pem, * d	er)	1		DIOWSB			
Certificate Store dio 1 MLAN Hode: AP PEY Setting MP General AP Security AP Wireless Class	ng	Upload	Click u	er) pload			DIOWSB			
Certificate Store dia 1 MLAN Mode: AP PEY Setting NP General AP Security AP Security AP Wireless Client Wireless VLAN Ma	ts tring	Upload	Click u Click u Ite CA Certificates	pload	wed By	Valid From	Valid To			
ALC Address Elle Certificate Strin dio 1 ALAN Hode: AP PEY Setting AP Convert AP Convert AP Security AP Wireless VLAN Ma Vereless VLAN Ma Vereless VLAN Ma Status	alibud	Internediate	Click u Click u Ite CA Certificates Issued To: Haryand Utra-3et Testing ercal intercat@3eti.com	pload US Maryland Roc rootca chacxin	ued By kvite Ultra-3eti Testing glingultra-3eti com	Valid From Aug 23 20:52:21 2016 GMT	Valid To Aug 18 20:52 21 2036 GMT	View		
ACC Address Clife entirelle Stime No 1 VLAN Mode AP Pres Setting IP General P Wireless Client Wreless VLAN Ma ec Tunnel Yorlies Ratus vices Settings HCP Server	skibiuð sé	Internedia No.	Click u Click u te CA Certificates Issued To Manyand Ultra-3et Testing stal intercal@3eti.com OCSP © CRL @ None	pload US Maryland Roc rootca chaoxin	ued By kvite Ultra-3eti Testing glingultra-3eti com	Valid From Aug 23 20:52:21 2018 GMT	Valid To Aug 18 20 52 21 2096 GM/T	View Dekte Set		

Enter URL address for OCSP or



2.3.9.4 Trust Root CA

The Trust Root CA page allows users to upload Trust Root CA certificates to the 3e-520 series device. The certificates can in the *.pem, or *.der format. The page lists certificates uploaded to the 3e-520 series device. To view a list of certificates in store click on the Trust Root CA tab in the Certificate Store page.

This page also allows user to view the contents of a certificate or delete a particular certificate altogether.

-		3eTI AirGuan	d TM Wireless Access P	oint					Logos
Ultra	3eTI	Usemame	CryptoOfficer		Host Nam	ei defa	suit.		
the critical		Role:	3e-local		Radio Reg	gion: Unit	ted States		_
iystem Configuratio	-	System Confi	guration -> Certificate	Store					
General Noisy Channel Con WAN	lort	Device CSR	Device Certs Inter	mediate CA	Root CA	Local CBI	OCSP Re	sponder	
DAN Bridge					-				
MAP Address Film	prog	Root CA Certifica	te File (* pem, * der)					Browse	
MAC Address Eter MAC Address Eter Certificate Store adio 1 WLAN Mode: AF PTIV Setting AP General AP Security AP Melloss Client	ang .	Root CA Certifica	te Fie (* pim, * der) Click uple	oad		*	- Select	Browse	file
Contribute Store Mark Advance Edu Centricate Store adio 1 WLAN Mode: AF PHY Setting AP General AP Security AP Windows Client Windows VLAN Ma	ang s pping	Root CA Certifica	te Fie (* pim, * der) Click uple ertificates Issued To	oad	Issued By	*	- Select	Browse the certificate Valid To	file
Ethemel VLAW Mar Advisors Ethe Certificate Store adio 1 WLAN Mode: AF Ptill Setting AP General AP General AP General AP General AP Winakes Clarith Winsless VLAN Ma Peor Tunnel Profiles Status	pping	Root CA Certifica Upload Intest Root CA C No. L25 Maryten + motes ch	te Fie (* pim, * der) Click uple	oad US Maryland rootca cha	Issued By Rockville Uba	Jes Testing	Valid From Aug 1 20 55:45 2016 GMT	Browse the certificate Valid To Jul 27 20 55 45 2026 GMT	file View Delete

Figure 37: Trust Root CA

2.3.9.5 Local CRL

The Local CRL page allows users to upload Local CRL certificates to the 3e-520 series device. The certificates can be in the *.pem, or *.der formats. The page lists certificates uploaded to the 3e-520 series device. To view a list of certificates in store click on the Local CRL tab in the Certificate Store page. This page also allows user to view the contents of a certificate or delete a particular certificate altogether.

100	3eTI AirGuard	Wireless Ac	cess Point				Loguet		
Ultra 3eTI	Usersanie: Rote:	CryptoOfficer Se-local		flost N Radio	lame: default Region: United	States			
System Configuration	System Confi	System Configuration -> Certificate Store							
Noisy Channel Control WAN	Device CSR	Device Certa	Intermediate CA	Reot CA	Local CRL	OCSP Responder			
LAN Bridge Ethernot VLAN <u>MAC Address Ethering</u> Cartificate Store Nation 1 WLAN Mode AP PHY Setting AP General	Centificate Revok	ition List File (* pen Click up	* der). Ioad		~	Select the cert Browse	tificate file		
AP Vireless Clients Windess VLAN Mapping	Local CRL Certif	icates Isoa	ed fly	1	LastUndate	Next Update			
IPsec Tunnel Profilee Status	1 US Maryland	Rockville Ultra-3eti 3eti	Testing rootca chaoxing .com	kn@ultra-	Dec 17 18 56 18 20 GMT	119 Dec 12 18 56 18 2039 GMT	View Delete		
Services Settings DHCP Server		C	lick to view or delet	te certifica	ate				



2.3.9.6 OCSP Signer

The OCSP Signer tab allows users to upload OCSP Signer certificates to the 3e-520 series device. The certificates can be in the *.pem, or *.der formats. The page lists certificates uploaded to the 3e-520 series device. To view a list of certificates in store click on the OCSP Signer tab in the Certificate Store page. This page also allows user to view the contents of a certificate or delete a particular certificate altogether.

100		3eTI AirGua	rd [™] Wireless Acc	ess Point					Lopest	
Ultra	3eTI	Usermanet	CryptoOfficer		Host Name	defaul	Construction of the			
the interaction		Role:	3e-local		Radio Regi	on: United	States			
ystem Configuration		System Con	System Configuration -> Certificate Store							
General Noisy Channel Cor WAN	throit.	Device CS8	Device Certs	Intermediate CA	Hoot CA	Local CRL	OCSP R	esponder		
WAN Enky Enemet VLAN Confiction Store and 5 1 WLAN Mode: AP Phy Setting AP Security AP Security AP Security AP Security AP Weekes Clients Winslaw VLAN Mapping		OCSP Signer C	etificate (* pern, * dee) Click upload	d		~	S	elect the certif Browse	icate file	
		OCSP Signer C	ertificate Issued To	Issu	ied By	Ya	lid From	Valid To		
Psec Tunnel Profilos Status		1 US Maryl ocspo	and Ultra-3eti Testing Locspo@3eti.com	US Maryland Rock rootca chaoxing	ville Ultra-Seti Te .lin@ultra-Seti.co	esting Aug 20	25 22 05 03 116 GMT	Aug 20 22:05:03 2036 GMT	View Delete	
Services Settings DHCP Server SMMP Agent		L		Click to v	iew or delete	certificate		/	d	

Figure 39: OCSP Signer

2.4 3e-520 Series Radio Configuration

2.4.1 Introduction

The radios of the 3e-520 series can be configured as Access Point, Mesh Access Point, Mesh Point or Client modes.

2.4.1.1 Access Point Mode

When a radio is configured as Access Point mode, the radio allows other Wi-Fi client devices to connect to a wired network. The uplink of the AP can be connected to a router for wired network. In Access Point mode, the AP interface can be configured to map up to 8 VLANs. Each VLAN is mapped to one SSID. Packets in the air between AP and wireless clients contain no VLAN tag. Packets from a wireless client associated with a given SSID are tagged by the AP, according to the configured mapping between SSID and VLAN. The VLAN tagging happens after the packets are received by the radio. VLAN tags in packets sent to wireless clients are removed by the AP before packets are transmitted to wireless clients.

2.4.1.2 Mesh Access Point Mode

When a radio is configured as Mesh Access Point mode, the radio acts as a bridge interface which can be bridged into a Mesh network. It also acts as an AP interface which allows Wi-Fi client devices association with the same SSID and encryption key.

2.4.1.3 Mesh Mode

When a radio is configured as Mesh Mode, the bridge interface always acts as VLAN trunk. Packets in and out of bridge are sent unmodified, so there are no VLAN-related configurations for the bridge interface.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

2.4.1.4 Client Mode

When a radio is configured as Client Mode, the radio interface acts as a Wi-Fi client device. It will scan the available APs and connect to the nearest AP which has the same SSID and encryption key as configured.

2.4.1.5 WAN Interface

The WAN Ethernet interface can be configured as VLAN truck or a single VLAN in the 3e-525N series. In 3e-523N series, WAN Ethernet interface is always in VLAN trunk mode. In trunk mode, packets are sent and received unmodified. Refer to Section 2.3.7 for WAN VLAN configuration.

2.4.1.6 Web Management Interface

Web management traffic can come from a Local Management port or a non-local port on the management VLAN. Refer to Section 2.3.7 for management VLAN configuration.

2.4.2 Radio WLAN Mode Configuration

The **Radio – WLAN Mode** screen (Figure 40: Radio – WLAN Mode) allows you to set the operating mode to one of the following:

- Access Point,
- Mesh Access Point,
- Mesh Point,
- Client.

NOTE: For devices like the 3e-525N, which has two radios, each radio needs to be configured separately.

-		3eTI AirGu	ard™ Wireless Access Point			1.200
Ultra	3eTI	Usemame:	CryptoOfficer	Host Name:	default	
ELECTROPHICS.		Role:	3e-local	Radio Region:	United States	
System Configuration General Norsy Channel Con WAN LAN Bridge Ethernet VLAN MAC Address Film Centificant Store Radia 1 MLAN Mode: AP PHTY Setang AP General	en nital rog	Radio 1 - Access Mesh A Mosh P Client Access	> WLAN Mode Point cress Point cint			

Figure 40: Radio – WLAN Mode

2.4.3 Radio PHY Setting Configuration

The Radio – PHY Setting link is used for configuration of the radio's Physical Layer (PHY) parameters.

1000		3eTi AirGu	ard [™] Wireless An	ccess Poin	t i		Lingmat
Ultra	3eTI	Username:	CryptoOfficer		Host Name:	default	
ELECTROPIECS		Role:	3e-local		Radio Region	: United States	
and the Reader		Radio 1 ->	PHY Setting				
General Noisy Channel Cor WAN	and .	MAC Address Wireless Mode:		00:0e.8et	if.cb:64 (SparklanCo)		
LAN		Channel Number:		1 (2412	MHz) V		
Elbernet VLAN		802.11n Featu	re:	Enable:	20MHz Channel 🛩		
MAC Address Filter Contificato Store	mg	Tx Pwr Mode:			Fixed Power Level 2:		
Radio 1 MI AN Mode: AP PHY Setting AP 12 Journal		* Rader signal * Power level of	was seen since last b options vary per chann	oot el and country	v codel		
AP Socurity		Advanced		23			
Wirplass VLAN Ma	s ppng	RTS Threshold	di:	2347	(Range: 1-2347)		
Psec Tunnel Dvolkis	2010	Propagation D	istance:	1	(Range: 1-72) Miles		
Status	Status		Action on Detecting Radar ?:		auto config channel		
Bervices Settings * Applicable only when V DHCP Servic SWMP Agent Senal Port Apply		ity when WLAN mode	is AP				
 Sental Communication 	tion.						

Figure 41: Radio 1 – PHY Setting

2.4.3.1 Wireless Mode

Select the wireless mode from the drop-down list. You can choose from the following options:

- 802.11a,
- 802.11g.

See Section 2.4.3.3 to configure 802.11n.

Note: The 3e-523E-900 product uses only 802.11b mode, restricted channels, translated into the 902-928MHz range, and restricted Tx Pwr Mode settings. It is not possible for the user to select modes, channels, or transmit power settings other than what is described in Appendix D.

See Appendix D. for specific details regarding the 3e-523E-900.

2.4.3.2 Channel Number

Table 8 shows all the channel numbers in frequency band for 802.11g and 802.11a. When multiple APs are used in the same area, it is important not to assign adjacent channels to these APs. For example, assigning channels 1, 6 and 11 to three APs will keep the transit frequencies of the three APs widely separated. This will reduce the interference between the access points.

NOTE: The country code may restrict the channels available for selection.

See Appendix D. for specific details regarding the 3e-523E-900 operation

Table 8:	Table 8: Frequency Channel Numbers					
Wireless Mode	Channel No.					
	1 (2.412 GHz)					
	2 (2.417 GHz)					
	3 (2.422 GHz)					
	4 (2.427 GHz)					
000 11 <i>m</i>	5 (2.432 GHz)					
802.11g	6 (2.437 GHz)					
	7 (2.442 GHz)					
	8 (2.447 GHz)					
	9 (2.452 GHz)					
	10 (2.457 GHz)					
	11 (2.462 GHz)					
	36 (5.18 GHz)					
	40 (5.2 GHz)					
	44 (5.22 GHz)					
	48 (5.24 GHz)					
802.11a	149 (5.745 GHz)					
	153 (5.765 GHz)					
	157 (5.785 GHz)					
	161 (5.805 GHz)					
	165 (5.825 GHz)					
NOTE: Actual available channels vary based on each country's RF regulations. The channels above are only available and do not require DFS detection in the U.S.						

2.4.3.3 802.11n Feature Setting

802.11n can be set to:

- Disabled,
- Enable 20 MHz channel,
- Enable 40- MHz channel,
- Enable 40+ MHz channel.

40- MHz option is to bundle channel with lower frequency band. **40+ MHz** option is bundle channel with higher frequency band.

2.4.3.4 Transmit Power Settings

The Transmit (Tx) Power (Pwr) Mode (**Tx Pwr Mode**) can be set to Off, Fixed and Auto (it set to Off by default). Setting the Tx Pwr mode in Auto giving the largest range of radio transmission available under normal conditions. As an option, the AP's broadcast range can be limited by setting the **Tx Pwr Mode** to fixed and choosing a **Fixed Power Level** from 1 to 6 (1 being the lowest power and shortest distance).

If you want to prevent RF transmission, set **Tx Pwr Mode** to off. This will not turn off RF transmissions from any associated wireless devices, but they will not be able to communicate with the device when the **Tx Pwr Mode** is off.

See Appendix D. for specific details regarding the 3e-523E-900 operation

2.4.3.5 RTS Threshold

The Request to Send (RTS) Threshold is the number of bytes used for the RTS/CTS handshake boundary. When a packet size is greater than the RTS threshold, the RTS/CTS handshaking is performed.

2.4.3.6 **Propagation Distance**

Propagation distance is an estimate of the radio distance to the nearby devices. It is used to estimate propagation delays between devices. It will not affect the device's transmitter power and therefore affect the distance of reachability.

2.4.3.7 Dynamic Frequency Selection (DFS) Configuration

DFS is a spectrum-sharing mechanism that allows wireless LANs (WLANs) to coexist with radar systems. It automatically selects a frequency that does not interfere with certain radar systems while operating in the 5 GHz band. In 3e-520 series, when radar signal is detected, the access point can be configured to either to select a different frequency band or to stop Wi-Fi transmission.

2.4.4 Access Point Configuration

111	3eTI AirGuard™ W	/ireless Access Point		55205
Ultra 3eTi	Username: Cryp Role: 3e-ic	itoOfficer ocal	Host Name: default Radio Region: United States	
System Configuration General Norsy Channel Control WAN Ethernet VLAN MAC Address Fibring Certiticate Store Radio 1 WAN Mode: AP MAY Setting AP General AP Security AP Security	Radio 1 -> AP Gen BSSID: SSID: Broadcast SSID: Beacon Interval: DTIM Period: Client Aging Period: Client Session Lifetime: IEEE 802.11w: Management from Client	teral 02.0e.8e.6ftcb.64 default Disable 200 2 300 0 Disabled Disabled Disabled Disabled Disabled Disabled Disabled Disabled Disabled Disabled Disabled Disabled Disabled Disabled Disable D	(Range: 144-1000) (Range: 1-265) (Range: 60-65635) (Range: 0-2147483647, 0 = D(sabled)	

Figure 42: Radio – AP General

2.4.4.1 AP General

The **AP General** settings options included on this screen are shown in Figure 42 and described further in Table 9.

NOTE: These settings should ONLY be changed by a professional installer.

Table 9: Radio Settings					
Settings	Options	Description			
Broadcast SSID	Enabled/disabled	When disabled, the AP hides the SSID in outgoing beacon frames and stations cannot obtain the SSID through passive scanning. Also, when it is disabled, the AP does not send probe responses to probe requests with unspecified SSIDs.			
Beacon	1//_1000 (TU)	The time interval in time unit (TU) which the 802.11			
Interval	144-1000 (10)	1 TU = 1.024 milliseconds.			
DTIM	1-255	The number of beacon intervals that broadcast and multicast traffic is buffered for a client in power save mode.			
Client Aging Period	0-65535 (seconds)	If the client is inactive for the period specified than it will be dissociated.			
Client Session Lifetime	0-2147483647 (seconds)	The time a client session will last. After a client session lift time, a client will need to re-associate with the AP again.			
IEEE 802.11w	Disable /Optional /Required	Disable: only accept client with 802.11w disabled. Optional: accept client with 802.11w either enabled or disabled Required: only accept client with 802.11w enabled.			
Management from Client	Disabled / Enabled	Disabled: Wireless clients are prevented from accessing the Management (WEB) interface. Enabled: Wireless client can access the Management (WEB) interface. Default is Disabled.			

NOTE: For backward compatibility, when a 3e-525N AP is deployed together with old 3e-525A-3 APs, the 802.11w needs to be disabled.

2.4.4.2 AP Security

		3eTI AirGua	rd™ Wireless Access Poir	Nt .			kogent
ELECTIONICS	3eTI	Usemame: Role:	CryptoOfficer 3e-local	Host M Radio	lame: Region:	default United States	
System Configuration General Network Channel Confi	e i	Radio 1 -> A Security Method	P Security # FIPS-802.11i				
WAN LAN Bridge Etioniet VLAN MAC Address Filter Certificate Store Radio 1	ng	Settings (key O Pre-Shared Pre-Shared Master Key	not set) I Key with Passphrase I Key with Master Key (64 hexadecimal digits)	1			
VILAN Mode: AP Phty Setting AE General AP Security		RADIUS Servi IPsec Tunnel	er /				1
Wrotess VLAN Mac Profiles Status	ping	Primary Rad Radus Serve	ius Server Settings r Type:	3eTI	×		
Services Settings DHCP Server SMMP Agent Servel Fort Servel Commerced Romote Administrat Web Server	on kin	Backup Radu Shared Secre	is Server IP Address 6 (minimum 10 characters)				
Admin User Manager List All User Add New User User Login Policy Remute A&A Setup Two-Fector Auth	ment T	Backend Pas Backend Key Backend Key	sword (minimum 10 characters) (128-bit in 32 hexadecimal digits) Again		_		
Monitoring/Reports System Status Bridging Status Bridging Status Adjacent AP List DMCR Creed Luct		Encryption Se Group Opher Pairwise Oph	ulte and Re-keying er dan Key i Relate	AES-COMP AES-COMP			
Logs System Log Web Access Log Auditing		Acply	nverve) bleane	Lo winnes 💌			

Figure 43: Radio – AP Security

The **Radio – AP Security** screen (Figure 43) displays a default factory setting of no encryption (Key not Set), for security reasons the unit will not communicate with any clients unless the encryption is set.

2.4.4.2.1 FIPS 802.11i

For the 802.11i setting, you must enable either PSK Settings or 802.1x (RADIUS) Settings. If you do not plan on deploying a RADIUS server, PSK mode is the security mode.

2.4.4.2.1.1 Pre-shared Key with Passphrase or Master Key

The pre-shared key can be of the following format:

- Pre-Shared Key with Master key: 64 hexadecimal digits,
- Pre-Shared Key with Passphrase: 8-63 bytes of characters are composed of any combination of upper- and lower-case letters, numbers, and special characters (that include: '!', '@', '#', '\$', '%', '^', '&', '*', '(', and ')').

2.4.4.2.1.2 Encryption Suite and Re-keying

Re-keying time is the frequency in which new encryption keys are generated and distributed to the client. The more frequent re-keying, the better the security. For highest security, select the lowest re-keying interval. Once you have selected the options you will use, click **Apply**.

2.4.4.2.2 802.1X / RADIUS (Common Criteria Compliant Mode)

If a RADIUS Server will be used, select 802.1x and input the Primary RADIUS Server and Request for Comments (RFC) Backend security settings. Use of a RADIUS Server for key management and authentication requires that you have installed a separate certification system, and each client must have been issued an authentication certificate.

Table 10: 802.1X Configuration					
Fields	Directions				
RADIUS Server Type	Select the RADIUS server type between: IETF and 3eTI. IETF is the default				
Primary RADIUS Server IP Address	Enter Primary RADIUS Server IP Address				
Backup RADIUS Server IP Address	Enter Backup RADIUS Server IP Address				
Shared Secret	Enter Share Secret for the RADIUS Sever with minimum 10 characters.				
Backend Password	Enter Backend Password				
Backend Key	Enter Backend Key in 32 hexadecimal digits				
Backend Key again	Re-enter the Backend Key				

The 3e-520 series can provide increased protection of the communication with the RADIUS authentication server by configuring an IPsec tunnel with the authentication server over which the RADIUS protocol will travel. Click the IPsec Tunnel check box and provide an IPsec Tunnel Profile in order to protect RADIUS communication with IPsec.

IPsec Tunnel	Profile auth_profile V Specify IPsec IDs
Primary Radius Server Settings	
Radius Server Type:	IETF 🗸
Primary Radius Server IP Address	192.168.254.11
Backup Radius Server IP Address	
Shared Secret (minimum 10 characters)	
Backend Security Settings	
Backend Password (minimum 10 characters)	
Backend Key (128-bit in 32 hexadecimal digits)	
Packand Kay Appin	

Figure 44: RADIUS Configuration with IPsec Protection

Note: IPsec Tunnel Profiles must be configured before IPsec can be used to protect RADIUS packets. See Section 0 in order to configure an IPsec Tunnel Profile.

Note: Compliance with Common Criteria (WLANEP) requires that all RADIUS traffic be transmitted within IPsec tunnels.

The 3e-520 series initiates the IPsec protocol when building and IPsec tunnel to a RADIUS server. During the authentication phase of the protocol, the 3e-520 series (Initiator) will specify to which of the RADIUS server's (Responder's) identities it wants to communicate with. Most RADIUS servers will have one IPsec identity bound to the RADIUS server's IP address. By default, the 3e-520 series will talk to this identity. There are instances where a RADIUS server may host multiple IPsec identities. In this case, the desired identity can be specified by checking the "Specify IPsec IDs" check box and supplying an IPsec identity.

IPsec Tunnel	Profile auth_profile V Specify IPsec IDs					
Primary Radius Server Settings						
Radius Server Type:	IETF 🗸					
Primary Radius Server IP Address	192.168.254.11					
Primary Radius Server IPsec ID	radius1.ultra-3eti.com					
Backup Radius Server IP Address						
Backup Radius SErver IPsec ID						
Shared Secret (minimum 10 characters)						
Backend Security Settings						
Backend Password (minimum 10 characters)						
Backend Key (128-bit in 32 hexadecimal digits)						
Backend Key Again						

Figure 45: RADIUS Configuration with IPsec ID

Note: When selecting an IPsec profile that uses Public Key authentication, the 3e-520 series will validate the supplied IPsec ID against the RADIUS server's X509v3 certificate. The RADIUS server's X509v3 certificate must contain the identity as either the subject or one of the subject alternative names.

After all the fields are entered correctly, click on the **Apply** button.

2.4.4.3 Access Point Wireless Clients

The **Radio – AP Wireless Clients** screen (Figure 46) shows the clients that are currently associate to the AP for that radio. Please note that the device will support up to 64 wireless clients per AP radio interface.

1000	1	3eTI AirGuard™ Wireless Access Point						
Ultra	3eTI	Username:	CryptoOfficer		Host Name:	default		
electicone ca	and the second s	Role:	3e-local		Radio Regio	on: United State	95	
System Configurati General Noisy Channel Co WAN LAN Bridge <u>Ethernet VLAN</u> MAC Address Fill Certificate Store	on ntrol	Radio 1 -> AP	Wireless Clients	Transmit Rate	RX Packets/Bytes	TX Packets/Bytes	Inactive (msec)	VLAN ID SSID

Figure 46: Radio – AP Wireless Clients

2.4.4.4 Wireless VLAN Mapping

In the 3e-520 series, by enabling **Wireless VLAN Mapping**, VLAN tag would be added to the packets of wireless client and be forwarded to their dedicated VLAN network. Figure 47 shows configuration for **wireless VLAN mapping**.

Ultra ELECTRONICS	3eTI	3eTI AirGuard™ Wireless Access Point						
		Username: Role:	CryptoOfficer 3e-local			Host Name: Radio Region:	default Lipited States	
System Configuration General Noisy Channel Cor WAN LAN Bridge	on Itroi	Radio 1 -> Wireless VLA	Wireless VLAN M N:	Mapping O Enable	Disable			

Figure 47: Radio – Wireless VLAN Mapping

2.4.4.4.1 Enable Wireless VLAN

To enable VLAN mapping, hit the **Enable** button and the **Apply** button. After enabling Wireless VLAN Mapping, you can create multiple wireless VLAN from the lower portion of the **Radio – Wireless VLAN Mapping** screen (Figure 48 shows the VLAN mappings is enabled).

The AP General SSID was configured on the **Radio – AP General** screen (Figure 42) can be set to enable or disable. When enabled, the packet coming in through this SSID will not have VLAN tags.

100	1	3eTI AirGua	ard [™] Wireless A	Access Point						Lugar
Ultra	3eTI	Username	CryptoOfficer			Host Name:	defau	1		
	1.000	Role	3e-local			Radio Region:	United	d States		
Sustem Configurati	10.0	Radio 1 -> 1	Wireless VLAN	Mapping						
General Newy Channel Co	introl	Wireless VLAN	4:	Enable	ODisable					
LAN		AP General SS	SID:	default						
Ethernet VLAN MAC Address Fith	eng			Enable	AP General SS	D O Disable	AP Gene	ral SSID		
Certificate Store		AP Beacon Int	ervat:	200		(Ran	ge: 144-	1000)		
WEAN Mode: AP Phty Setting AP General AP Security AD (Visioner Floor		/culy								
Wreless VLAN M Paec Tunnel	atping	Existing V	LANS Create V	LAW						
Profiles Status										
Services Settings		Delete	Edit							
SMIP Agent Senal Port		No.	1 - ¹⁰ 1 -	SSID	β.	BSSID V	LAN ID	Security Policy	Broadcast Enabled	Note
Remote Administr Web Servic	0001	1								

Figure 48: Radio – Wireless VLAN Mapping – Enable Wireless VLAN

2.4.4.4.2 Create VLAN

Click on the Create VLAN tab to create VLANs for each SSID.

Figure 49 shows the VLAN configuration screen in Create VLAN page. Each SSID can be associated with a different VLAN tag. That VLAN tag will be added to the Ethernet package when coming into the AP and be removed when sent out through the radio. A VLAN tag configuration of "0" will eliminate the VLAN modification for the SSID.

The detail SSID AP configuration is the same as the **Radio – AP General** configuration (Section 2.4.4). The detail SSID security authentication configuration is the same as in **Radio – AP Security** configuration (Section 2.4.4.2) excludes Pre-Shared Key with Passphrase option.

100	3eTI AirGuard"	Wireless Ac	cess Point			Lugout	
Ultra 3e	TI Username: 0	CryptoOfficer		Host Name:	default		
ELECTROPICS .	Role: 3	le-local		Radio Regioro	United States		
System Configuration General Noisy Channel Control WAN LAN	Radio 1 -> W	reless VLAN	Mapping -> Create V	Nireless VLAN	194)		
Ethernet VLAN MAC Address Filtering Certificate Store	SSID:		Note: Enter 0 for "No	802.1Q header*	na N		
Radio 1 ISLAN Mode: AP	BSSID:		2 Der Zer Dit ab Ga	125			
PHY Setting AP General AD Security	DTIM Period:		2	0	Range: 1-256)		
AP Wireless Clients	Client Aging Per	iod.	300	(1	Range: 60-65535)		
Wreless VLAN Mapping	Client Session L	fetime:	0	0	Rampe: 0-2147463647)		
Profiles Status	IEEE 802 11vi		Disabled 9	112			
Services Settings DHCP Server SWMP Asset	Note						
Satul Port Satul Communication Remote Administration	Enable SSID bro	adcast					
Web Server Admin User Managament	Security Method:	FIPS-802.11i					
List Al Users Add New User	Settings (key o	ot set)					
User Login Policy Remote A&A Setup Two-Factor Auth	Pre-Shared I Master Key (Key with Master K 64 hexadecimal d	ey ligits)				
Monitoring/Reports System Status	O 802.1x		1.000				
Bidging Site Map	RADIUS Server	£					
DHCP Clent List	Primary Radiu	s Server Setting		1 m 100			
System Log	Radius Server	Type		3ell 💌			
Web Arcess Log	Primary Radius	Server IP Addres	5				
Audit Configuration	Backup Hadus	Server IP Addres	8				
Audit Log System Administration Email Notification Conf	Shared Secret	(minimum 10 chai	(acters)				
Radio Tx Off Control System Disprado	Backend Secu	rity Settings		-			
Default Configuration	Backend Passy	tord (minimum 10	characters)	5			
Reboot	Backend Key (Backend Key (128-bit in 32 hexadecimal digits)					
On Demand Self-test Periodic Self-test Utilities	Backend Key A	gain		<i>3</i>			
Help	Encryption Suit	te and Re-keying	Li torrada	100			
	Group Cipher Painwise Cinher	r.	AES-CCI AES-CCI	MP			
	Group Encrypti	on Key Lifetime	15 Minu	utes 🗸			

Figure 49: Radio – Wireless VLAN Mapping – Create Wireless VLAN

1) Notes and Tips:

- a) In 3e-520 series, VLAN 1 is always tagged. VLAN 0 can be used for untagged VLAN.
- b) If the WAN Ethernet interface is put in "Single VLAN" mode and its VLAN ID is different than the "Management VLAN", the device is not manageable through the WAN port. In this scenario, manage the device through the LAN port when the user is around the device.
- c) If the WAN Ethernet interface is put in "Single VLAN" mode and its VLAN ID is different than "Sensor VLAN", the video traffic is not reachable through the WAN Ethernet port. In this scenario, configure the AP radio to have one SSID mapped to the "Sensor VLAN" and access video traffic through this SSID when the user is around the device.
- d) In (b) and (c) above, the user can always access the device or video traffic through the wireless bridge remotely.
- e) When the device is deployed in an outdoor environment, it may be desirable to disable/block the Ethernet WAN port. This can be achieved by putting "Ethernet WAN" in "Single VLAN" mode and configuring a non-existing VLAN ID. This way, no traffic will be sent out from this Ethernet WAN port.
- f) It is always recommended to manage the device through the LAN virtual interface (192.168.15.1) when local access is available for the device. The LAN virtual interface is helpful especially when:
 - i) A device IP address is unknown,
 - ii) The device has not obtained an IP address from a DHCP server,
 - iii) The Management VLAN is tagged.

2.4.5 Mesh Point Configuration

2.4.5.1 Wireless Mesh Introduction

In the Mesh mode of 3e–520 series, wireless mesh is used to set up a backhaul mesh network connection. Wireless mesh provides a mechanism for APs to collaborate, which enables extension of the Basic Service Set (BSS) of a standalone AP and connection of two separate LANs without installing any cabling.

The wireless bridge function supports a number of bridging configurations. Some of the most popular settings are discussed in this section:

- Point-to-point bridging of two Ethernet links
- Point-to-multipoint bridging of several Ethernet links.

Meshing is a function that is set up in addition to basic AP setup. If you will be using the unit solely as a bridge, some of the settings you may have selected for AP use will not be necessary.

If setting up as a Mesh during initial setup, you can use the LAN Port directly wired by Ethernet cable to a laptop to set the appropriate settings. The management screens that you may need to modify, regardless of what type of bridging mode you choose, are in the **Radio – Mesh Point** section of the navigation bar. These include:

- Wireless Mesh General,
- Auto Bridge Security.

2.4.5.2 Wireless Mesh General

and a second sec	3eTI Ali Gua	Ind Wireless A	ccess Point			Light
Curra 3eTi	Unermenne. Noie:	CryploOfficer 3e-local		Roat Name Radio Region:	default United States	
System Configuration General	Radio 1 → 1	Wireless Mesh G	lenaral			Nontring
Nexy Charriel Calleri WAN UN	Webiess Vies/	Mode	Auto Bridger			
Ethernel VLAN MAC Address Filtering	858/0		00 Decile Of cit 64			
Cartificate State	\$940		edectamant.			
WILADI Moder Meah Point.	Broadcast SS	0	Cinabia 🐓			
Virelese Mests General	Selecon livery	<i>d</i> .	1000 (Range: 144	1000)		
Page Tatent	Max Auto Brid	ges.	16 (Range 1-	cii		
Profiles Status	- RSSI Window	Sine	6 (Range: 1-	DO) HED		
Services Settings	Signal Shorigh	ti Threshold	10% W Hetel	- Annalised		
Shimp Agent	Link Sensitivity		19% M Into			
Setal Conceptation Setal Conceptation Remote Advantation	ignore Mesh 3	igneture	No 💌 ieti			
Admits Unor Monagement Unit All Unors Add New Unor	Auny					
User Light Paley Remote ALA Bridge Two Party: Auto	Remote Win	tess fieldge Nodes				
and the state of the state of the	NO. 15550	50181389	-Q01 0	Date U	the scalute	RHL /

The **Radio – Wireless Mesh General** screen (Figure 50) contains wireless bridging information. This screen is important in setting up your bridge configuration.

Figure 50: Radio – Wireless Mesh General

From a mesh network, the wireless mesh sniffs for beacons from other wireless mesh nodes and identifies APs that match a policy such as SSID and channel.

Instead of simply adding the APs with the same SSID/channel to the network, a three-way association handshake is performed in order to control network access.

To make a unit the root node, set the bridge priority lower than any other node in the network. This done on the **System Configuration – Bridge** screen (Section 2.3.6).

Table 11: Wireless Mesh General Setting Options						
Settings	Options	Description				
SSID	Numbers and Letters	Can be any set of letters and numbers (up to 32) assigned by the network administrator. This nomenclature has to be set on the wireless bridge and each wireless device in order for them to communicate.				
Broadcast SSID	Disable/Enable	When disabled, the AP hides the SSID in outgoing beacon frames and stations cannot obtain the SSID through passive scanning. Also, when it is disabled, the bridge does not send probe responses to probe requests with unspecified SSIDs.				
Beacon Interval	144-1000 (TU)	The time interval in time unit (TU) which the 802.11 beacon is transmitted 1 TU = 1.024 milliseconds.				
Max Auto Bridges	1-16	The maximum number of auto bridges allowed. This number defines how many neighbor bridges (with the same Basic Service Set Identifier (BSSID)) the radio is allowed to associate with. Based on application, a smaller number will allow higher bridge throughput.				

Table 11 describes the auto bridging general setting options.

	Table	e 11: Wireless Mesh General Setting Options
Settings	Options	Description
RSSI Window Size	1-100	RF signal fluctuates over time and the fluctuation varies in different operating environments. This parameter serves to smooth Received Signal Strength Indicator (RSSI). The RSSI that applications use will be an average of last window-size RSSI samples. The sampling rate depends on the beacon interval of the neighbor mesh node. This helps stabilize the network. For fixed location deployment, higher values are suggested for both window size and beacon interval. Lower value is recommended while adjusting antenna or distributing mobile mesh devices.
Signal Strength Threshold	50% 45% 40% 35% 30% 25% 20% 15% 10% None	On creating a bridge link, if the signal strength is less than this threshold, the link will not be created. After a link is created, it will not be destroyed even after the signal goes below this threshold. This helps to stabilize the network.
Link Sensitivity	50% 45% 40% 35% 30% 25% 20% 15% 10% None	After a link is created, signal strength is mapped to RSTP path cost. Because RF signal fluctuates, path cost needs to be adjusted accordingly. However, adjusting path cost too frequently will cause network instability. This field serves as a threshold to adjust path cost. Path cost is adjusted if signal strength increases/decreases by this value since the last adjustment. If the value is set to none, every link act like 100% signal and there will be no path cost adjustment later on. It is strongly recommended that the same value is set on all other nodes in the same network.
Ignore Mesh Signature	Yes/No	Mesh node puts a signature in Beacon IE (Information Element). Only node with matching SSID and encryption algorithm will have match signature. Select Yes to ignore this signature so that it is backward compatible. Select No if you don't need backward compatibility.
Remote AP's MAC Address	Read Only	Displays the BSSID of remote bridges that were added on the Wireless Bridge - Radio screen.

2.4.5.3 Bridge Monitoring

In the upper right-hand corner of the **Radio – Wireless Mesh General** screen (Figure 50) there is a **Monitoring** button. If you click on this button, a pop-up window will appear (Wireless Bridge Information in Figure 51). If you select **Enable refresh**, you can set the **Bridge refresh interval** from 5 seconds to 30 minutes. Refreshing the screen allows you to see the effect of aiming the antenna to improve signal strength.

Wireless Bridge Information								
🛛 🗛 Apply 🛛 Enable refresh 💿 Disable refresh 🔿 🛛 Bridge refresh interval: 5 seconds 🛛 💌								
Remote Wireless Bridge Nodes								
No. BSSID Signal Strength	Tx Rate	Link Status	Description					
1. 00156D84F8F1 Excellent (-33 dBm/100%)	300.0 MBit/s	linked	BR2					
2. 00156D84F4CC Excellent (-48 dBm/92%)	300.0 MBit/s	linked	BR1					
Copyright © 2012 Ultra Electronics - 3e Technologies International. All rights reserved.								
	-							

Figure 51: Wireless Bridge Information (Monitoring)

2.4.5.4 Auto Bridge Security

The **Radio – Auto Bridge Security** screen (Figure 52) is used to configure static encryption keys for the wireless bridge. This is an important screen to set up to ensure that your bridge is working correctly. In order for communication to occur, the encryption key that you use on this screen must be the same for any bridge radios connected to your Mesh network.

The Encryption Type can be configured as:

- 1) Static AES with 128-bit, 192-bit or 256-bit Encryption Key; or
- 2) Static AES-CCMP with 128-bit, 192-bit or 256-bit Encryption Key.

e sett Aklauri (1	M) - Inte	met Explainer				
	192.3	144.15.1	UNIVERSITY OF TRANSPORT	 D • 9 Certificate error 	(MT) DWUDIA TTSE	×
Ultra	3eTI	JeTI AlrGu Deernemel Role	ard ¹⁰ Wireless Access S CryptoOfficer Je-local	Point Itent Name Radio Region:	default United States	Log
System Configuration General Ninky Channel Com WAN LAN Droge Etwaren VLAN MAC Arctices Filter	4 610	Radio 1 -> Encryption Typ	Auto Bridge Security Static AES Static AES-CCM			
Contention Scient Radio 1 URANI Moder Mesh Physics Mark Llast Market Sciences Profiles Decorption Services Service Services Service Services Service Services Service Services Administrat View Service	Point	© 120-bit Er	Key Again Key Again Key Again Key Again Key Again Key Again	i 32 hexadecimal digits (G-9, a-1, or A-F) i 45 hexadecimal digits (G-9, a-1, or A-F) i 64 hexadecimal digits (G-9, a-1, or A-F)		
Admin User Manaper List 43 Users Add New User	ment	ADRIV				

Figure 52: Radio – Auto Bridge Security

2.4.6 Wireless Client Configuration

The radios in the 3e-520 series can be configured as wireless clients to connect with an AP. The client mode of a radio cannot be combined with either bridge or AP mode.

2.4.6.1 Radio PHY Setting in Client Mode

The **Radio – PHY Setting** screen is shown in Figure 53. It is used to setup physical layer parameters of the radio. The client radio will scan all the channels to find out the available APs. On the **Radio – Client General** screen (Figure 54) the desired AP can be selected. The **wireless mode**, **channel number** and **802.11n Feature** are not configurable and grayed out.

For setting the RTS Threshold and Propagation Distance follow the instructions in Section 2.4.3.

addition in the	3eTI AirGu	3eTI AirGuard " Wireless Access Point						
Ultra 30	Uwwmanne:	CryptoOfficer		filter Nation:	default			
	Rolei	364004		Radio Regium	United States			
Taxan Configuration	Radio 1 ->	PHY Setting						
Deminat Deminat	MAC Address		10 De Je Of	ch:64 (RpanklanCa)				
WAN COMPACT OF COMPACT	Wyeless Mod	es.	802,119					
LAN Deitur	Channel Num	ber	3-(24123)	(Hu) (M)				
Ethemet VLAN	002.11n7eat	ate:	Coulds: P	OMHa Channel 94				
Certificate Nore	Ta Per Mode		- 90 W	Fixed Power Level * 14				
Radac 1 Wi Abi Mada Class Physical Control (1997)	* Radar signal * Power level	I was seen since last t options vary per chan	net and country	codel				
Client Becarity	Advances							
Page Tannel Profiles	RTS Threshol	đ	2347	(Range: 1-2347)				
Stells	Propagation 0	listance	1	(Ranpi: 1-72) Men				
DHOP Server	Action on Deb	comp Rader 1	euto-cordi	e channel * M				
ShinP Agent Serial Part	* Applicable of	nly when WLAN mode	s AP					
Setal Communication	Including .							
Web fames	V6A.							

Figure 53: Radio – PHY Setting

2.4.6.2 Client General

The **Radio – Client General** screen is shown in Figure 54. Radio will automatically scan all the channels to find any available APs in the area. The desired AP's SSID will be put in the **SSID** field that will send association and authentication packets to the nearby Access Point. You can Enable the IEEE 802.11w when the AP requires to use 802.11w to enhance further protection to a set of robust management frames by the Protected Management Frames (PMF) service. The 802.11w is configured on a per-SSID basis and disabled by default.

Clicking on the **Scan** button will refresh the list of all the APs in the area. The SSID does not have to be the one being detected. Some APs may be configured not to broadcast its SSID.

The **Status** field shows status of the connection between the client and the AP. The client and AP will not associate with each other until the **Radio – Client Security** screen (Figure 55) is configured correctly.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

and the second se	3eTI AirGuard ** Wireless Access Point							
Ultre 3eTi	Ukarniamer	CryptoOfficer		Thest N	lative: .	UUT2		
Tarici Marinea	Hole:	3a local		NaRo	Region	United Status		
System Configuration	Radio 1 -> Cli	ent General	o so antera a					
Noise Chamel Control WWW	MAC Address SSID	1	00 0e 0e 81 ca) AP 10 1	19 (SpandanCo)				
LAN Didge Officeral VLAN	IEEE 002 11w	1	Disabled M					
MAC Address Fillering Gerbficate Store	Connect		Decovert					
WLAN Mude Client	Status						1 3	
Clert Security	Association Statu Authentication St BSSID	ti duk	Associated Authentication 1	Successitul				
Profiles Situtos	Access Point IP A Frequency, MHz	ddress:	192 168 254 25 5240 (channel)	4 4				
Services Settings DHCP Teacer SMIP Agent Remote Advancementation Well Service	Signal Level Signal Strength Tx Rate:	1	33 dBm 100% 70.6 Mbit/s	270				
Admin Usar Management List Al-Usars Add free Usar	Behrsite .							
Unit Logis Policy Remote AdA Solida Tara Parter Ada	Site Servey basid	frequency(chu	(lenna)	signal level		flags	said	
Aunitacing/lingests Scature Status	18 1b eb 4c 8c f	3 2412 (channe 1 2402 (channe	48.832	-51	MPA MPA	2 PSK CCMP	38636580-2-0878F MBQ40	
Rodging Blatus Bodging Blat Map Adjacert AP Lat	02 0e 8e 87 ca d	E 2852 (chanse 3 5240 (chanse	# 11) 4 40)	-36 -33	WPA	2.PSK.CCMP	AP101	
Opt Channel Last	AP Court: 4							
Web Access Log NodRing	Stan							
Huas Corregulatory								

Figure 54: Radio – Client General

2.4.6.3 Client Security

The **Radio – Client Security** screen is shown in Figure 55. The client authentication security is carried out on this screen. The **Security Type** can be configured as **WPA2-PSK-PASSPHRASE**, **WPA2-PSK-CCMP** or **WPA2-EAP-TLS-CCMP**.

WPA2-PSK-PASSPHRASE: The 8-to-63 characters should be entered in the Passphrase field.

WPA2-PSK-CCMP: The 64 hexadecimal digits should be entered in the Master Key field.

1 floor	3eTI AirGuard	™ Wireless Access Point			Legast
Ultra 3eTI	Username:	CryptoOfficer	Host Name:	default	
ELECTROPICS	Role:	3e-local	Radio Region:	United States	
System Configuration General Noisy Channel Control WAN LAN Bridge	Radio 1 -> Cli Security Type: W Encryption (key	ent Security PA2-PSK-PASSPHRASE PA2-PSK-CCMP PA2-EAP-TLS-CCMP not set)			
MAC Address Filtering Certificate Store	Master Key (64 h	exadecimal digits)			
WLAN Mode: Client PHY Setting Client General Client Security	Radio Tx power in	s off. Turn it on before client securit	y can be set.		

Figure 55: Radio – Client Security Type

WPA2-EAP-TLS-CCMP: As shown in Figure 56, by selecting client security type WPA2-EAP-TLS-CCMP, use a pre-loaded certificate from the list and provide a **Login Name** can be configured on this screen.

and the second se	3eTI AirGue	rd ** Wireless Access Point		2210	Light
Ultro 3eTi	Hustriams:	CryptoOfficer	Host Nerve:	default	
ILL HONCE	Rafe	De-local	Radio Region	United States	
System Configuration Density Non-Channel Control WAN LAN DMps Elimited VLAN	Radio 1 -> C Security Type:	Client Security WPA2 EAP-TLS-OCMP			
Certificate Store	Certificate	US Maryland Ultra-Suti Testing	device1 device1@0et.cc	101 V	
Radio 1 WLAN Michel Cillent DHY Setting	Login Name:	device 1			
Cliente Rectatho	Approv				

Figure 56: Radio – Client Security – WPA2-EAP-TLS-CCMP

A device certificate and root CA must be loaded from Certificate Store (details in Section 2.3.9 and Figure 57). Then click on **Apply** to save the client security.

and the second	const.	JeTI AirQuard	I ^{7#} Wireless Acco	ess Point	1.000 (100	super-		0.0
BUCTBONICS	3eTI	Unormation: Role:	CryptoOfficer 3e-local	source wee	Hast Name: Radio Region:	default Onited States		
System Configuration Conversi Noisy Channel Carth 214N	6	System Confi	guration -> Certif	Icate Store	Host CA. Loud	CRL DCSP Re	-pooler	
UNI Bistyle Ethernet VUAN Carthease Ether Eadlert WUAN Mode: Clever Bitt Sattag David General	9	Continued Certific Private Key Pasa	ale i Piikale Key File (iord	".pem, ".pfx, "p12)	1		Боуная	
Part Security Part Turnel Profiles Statut		Device Certificat						
Service Service DVCP Serve SM/P Agent Service Port Thread Communication Remote Advantation		1 US Maryla device3	et Ubis-Set Testing Jevics 1@Set.com	US Manysent Roc rootza chakke	kvile Utus-Bet Testing g.ingultus-Bet coni	Aug 1 21 10 22 2016 GALT	Valid To JA 37 21:18/22 2036 GMT	View Delete
Test Started Advant User Metagom Lan AP Grant Add New User User Loger Polity Benetik AdA Schap Test-Factor Add Meelity begrösports System Status Diriging Status Diriging Status	met	Contributed Motor Notify Email Notification Email Actives:	Earlinn (83 O'Eir	i days before cert expl uble 🖲 Disable	res and every: 7 da	wa thereafter.		

Figure 57: Certificate Store – Loading Client Certificates for WPA2-EAP-TLS-CCMP

2.5 IPsec Configuration

IPsec tunnels can be configured to protect communication between the 3e-520 series and external services such as a RADIUS authentication server, NTP server or Remote Audit Log server. IPsec tunnels are used for a trusted channel to external servers in Common Criteria configuration. In order to use IPsec, at least one IPsec Tunnel Profile must be configured. Each profile specifies a cipher suite, authentication method, credentials and rekey timers. Once an IPsec Profile is configured it can be applied to a service. IPsec Security Policy Database (SPD) is dynamically configured, based the trusted paths IPsec is used to protect. For example, IPsec tunnel is configured to protect a remote syslog trusted path. In this instance, records are written into the SPD to protect packets passing between the 3e-525/523 and the remote syslog server based on source address, destination address, protocol and port number. When protecting remote syslog trusted path, the SPD will have one record matching ingress UDP traffic with source address and port corresponding to the remote syslog server. Additionally, the SPD will have one record matching egress traffic, with destination address and port, corresponding to the remote syslog server. Traffic passing through the security boundary and matching either of these two records will be classified as "PROTECTED" using IPsec transport mode. Traffic that does not match any records in the SPD will be allowed to "BYPASS" the security boundary unperturbed. Additional records are written into the SPD when additional trusted paths are configured for IPsec protection (i.e., RADIUS server, NTP ...). Traffic that does not match any records in the SPD will be "DISCARDED".

The device supports IPsec tunnel and transport modes which allows packet payloads to be encrypted. The device requires no administrative configuration and negotiates either depending on the peer.

2.5.1 IPsec Tunnel Profiles Configuration

An IPsec Tunnel profile must be configured in order to enable secure communications between the 3e-520 series device and a remote log server or an NTP server. Once configured, the profile will then be available from NTP Server and the Audit Configuration sections (see Figure 23: NTP Time Source with IPsec Protection and Figure 85: Remote Audit Logging). The 3e-520 series device supports up to 8 tunnel profiles. To create an IPsec Tunnel profile, click on the 'Profiles' menu under 'IPsec Tunnel' on the left of the screen. This page shows the IPsec Profiles configuration options and provides a list of all configured profiles.

2.5.1.1 Profile Settings

Add a new IPsec tunnel profile by filling out the following profile settings and clicking the "**Add**" button. Refer to Table 12: IPsec Tunnel Profile Settings for more setting details.

New Profile Name: Enter profile name, string must be between 3 and 16 alphanumeric characters, _ and. is allowed.

Cipher Suites: Select from Auto Negotiated, Suite B GCM 128, Suite B GCM 256, AES CBC 128, or AES CBC 256 (see Section 2.5.1.2 and Table 13 for more detail).

Authentication: Select Pre-Shared or Public Key. If the "Public Key" is selected, then a pre-loaded certificate should be selected from the drop-down list. Please note that the certificate list is populated by the certificates loaded via "Certificate Store" which is covered in Section 2.3.9.

Pre-Shared Key: if using Pre-Shared Authentication, enter key here.

Rekey Timers: Enter rekey time in minutes for IKE SA and IPsec SA (20-144min range). Enter IPsec SA life rekey timer in kilobytes and kilo packets within the ranges listed (0 for unlimited).

Party and	-	JeTI AirGu	ed ^{re} Western	Access Point				A support
ELECTRONICS	3eTI	Usarbatea: Refe:	GryptoOfficer Se-local		Hoat A Histiis	loma: Rogiún:	default Gedeel States	
Instant Configuration General Noney Channel Con WAN EVAN Entrone VLAN MAC Address Filter Configuration	evel evel	IPsec Tunn New Poste N Optier Solar Authenticatio Pre-Shared K	el -> Profiles anvi ey:	Aun	n Negotizend		-	
Radio 1 WEAN Marke Cline PHY Setting Client Deneral Client Deneral Client Deneral Problem British British Shift Deneral Shift De	t En	ReleviTines IRE SA Lifeti IPSec SA Life IPSec SA Life IPSec SA Life	ne(nonutes) etime(minutes) et Moytes: e MPackets	180 80 0	26 1440 20 1440 98-2047 192-209	min men Mörykes: 0 fo P151 Könscker	r Untimitied to. O for Unterrited	
Ristgics Status Endging Stat Mag Adjacent AP List DECP Claim List app Noth Accounting Mich Accounting Audit Configuration Audit Configuration Audit Long	LON.	Configured P Deter Padis 1 Servi 2 Devic 3 Pak	noffles Name Col r1 Aut e1 Aut Aut	er Sute n Negolisted n Negolisted n Negolisted	AutherOcason Public Key Public Key Pre-Shured Key	Configurer Configurer Configurer Configurer	m Times a 100-0000 a 100-0000 a 100-0000 a 100-0000	în Une

Figure 58: IPsec Tunnel Profiles

Configured Profiles: this section lists all configured profiles and allows the user to select a profile for deletion.

Table 12: IPsec Tunnel Profile Settings								
Settings	Options	Description						
New Profile Name	String from 1 to 16 characters	Free formed string to name the profile. Up to 8 profiles can be added and each name must be unique. There is not default value.						
Cipher Suite	Auto Negotiated Suite B GCM 128 Suite B GCM 256 AES CBC 128 AES CBC 256 Suite B GMAC 128 Suite B GMAC 256	Cipher suites used during IKEv2 Phase 1 as well as ESP Phase 2 cipher negotiation. See Table 13: Cipher Suites for details. The default value is "Auto Negotiated".						
Authentication	Pre-Shared Key Public Key	Authentication method used during IKEv2 Phase 1 Security Association establishment. The default value is Pre-Shared Key						

Table 12: IPsec Tunnel Profile Settings									
Settings	Options	Description							
Pre-Shared Key	 String from 16 to 32 characters long enclosed within double quotation marks Hex string of 8 to 63 characters 	The Pre-Shared Key used during the IKEv2 Phase 1 Security Association establishment. The key can be entered as a hex string or a double quoted string that is hashed into the resulting key. The device conditions the text-based pre- shared keys using the SHA1,SHA-256, and SHA384 hash algorithm.							
IKE SA Lifetime	From 20 to 1440 minutes	The number of minutes an IKEv2 Phase 1 Security Association can be active before it is automatically re- authenticated. The default value is 180 minutes							
IPsec SA Lifetime	From 20 to 1440 minutes	The number of minutes an IPsec ESP Child Security Association (Phase 2) can be active before it is automatically rekeyed. The default value is 60 minutes.							
IPsec SA Life Kbytes	From 96 to 2047 Kbytes. 0 indicates Unlimited	The number of bytes that can be transmitted over an IPsec ESP Child Security Association (Phase 2) before it is automatically rekeyed. The default value is 0 indicating Unlimited.							
IPsec SA Life kPackets	From 192 to 2097151 kPackets. 0 indicates Unlimited	The number of packets that can be transmitted over an IPsec ESP Child Security Association (Phase 2) before it is automatically rekeyed. The default value is 0 indicated Unlimited.							

2.5.1.2 IPsec Cipher Suites

The 3e-520 series supports a number of different encryption standards and protocols, including the four Suite B cryptographic suites as defined in RFC 6379.

The order in which the protocols appear gives the preference order of use within the 'Auto Negotiated' mode. Otherwise, if one of the Suite B modes is chosen, the denoted protocols are used for that given mode.

Table 13: Cipher Suites

Mode	IKEv2		ESP (Hardware encryption)			
	Encryption	Integrity	Pseudo Random Function	Diffie Hellman Group	Encryption	Integrity (where applicable)
Standard (Auto Negotiated)	aes256cbc aes128cbc	sha512 sha384 sha256 sha1	Same as negotiated integrity	ecp521 ecp384 ecp256 modp2048 modp1536 modp1024	aes256gcm128 aes256gcm96 aes256gcm96 aes256gcm64 aes256ccm64 aes256cbc aes128gcm128 aes128gcm128 aes128gcm96 aes128gcm96 aes128gcm64 aes128ccm64	sha512 sha384 sha256 sha1

Mode	IKEv2		ESP (Hardware e	ncryption)		
	Encryption	Integrity	Pseudo Random Function	Diffie Hellman Group	Encryption	Integrity (where applicable)
					aes128cbc	
Suite B GCM 128	aes128cbc	sha256	sha256	ecp256	aes128gcm128	-
Suite B GCM 256	aes256cbc	sha384	sha384	ecp384	aes256gcm128	-
AES CBC 128	aes128cbc	sha256	sha256	ecp256	aes128cbc	sha256
AES CBC 256	aes256cbc	sha384	sha384	ecp384	aes256cbc	sha384
Suite B GMAC 128	aes128cbc	sha256	sha256	ecp256	NULL	aes128gmac
Suite B GMAC 256	aes256cbc	sha384	sha384	ecp384	NULL	aes256gmac

2.5.2 IPsec Tunnel Status

To check the status of IPsec tunnels, click on the 'Status' menu under the IPsec heading on the left side of the screen. This page (Figure 59) lists current status of all IPsec Tunnels configured in the 3e-520 series device.



Figure 59: IPsec Tunnel Status

2.6 Services Settings

2.6.1 DHCP Server

The **Services Settings - DHCP Server** screen (Figure 60) is used to configure the DHCP server function accessible from (only) the Local LAN port. The DHCP server function is used for initial configuration of the management functions.

DHCP Server is Enabled by default. You can disable the DHCP server function, if you wish, but it is not recommended. You can also set the range of IPv4 addresses to be assigned; **Starting IP Address** defaults to 192.168.15.10, **Ending IP Address** defaults to 192.168.15.240.

The **Lease Period** (after which the dynamic address can be reassigned) can also be varied; the default setting is 1 Day.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com
and the second se	3eTI AirGuard** Wireles	is Access Point		14000
Ultra 3eTi	Hermann CryptoOffer Note: 3e-local	*	Hool Name da Radio Region: U	efacet roted States
System Configuration General Roley Charmal Control WAN LAV Boltys Effammer (CAN MAC Advance Filling Certification Since Harlies Y VicAdr Moder Chart Fifty Sectory Chart (General Chart Security Profiles Status Status Efforts Security Effamilies Status Status Security Security Security Security	Services Settings -> De Exating IP Address Ending IP Address Lease Period Apply	ICP Server		

Figure 60: Services Settings — DHCP Server

2.6.2 SNMP Agent

The **Services Settings – SNMP Agent** screen (Figure 61) allows you to configure the Simple Network Management Protocol (SNMP) Agent for the device. By default, the SNMP Agent is set to disable (both SNMPv1/SNMPv2c and SNMPv3 protocol are **unchecked** shown in Figure 61). The SNMP Manager function interacts with the SNMP Agent to execute applications to control and manage object variables in the AP. Common forms of managed information include number of packets received on an interface, port status, dropped packets, and so forth. The SNMP is a simple request and response protocol, allowing the manager to interact with the agent to either:

- Get Allows the manager to Read information about an object variable,
- Set Allows the manager to Write values for object variables within an agent's control.

Section 2: Device Configuration WiFiProtect User's Guide 3e-520 Series

ULTRA

4.45	3eTI AirGuard	[™] Wireless	Access Point			Logout
Ultra 3eTi	Username:	CryptoOfficer		Host Name:	default	
ELECTROPIECS	Role:	3e-local		Radio Region:	United States	
	Services Sett	ings -> SNMI	P Agent			
System Configuration General			-			
Noisy Channel Control	SNHP General G	onliguration				
LAN	SNUP Protococ		USNMPV1,SNMPV2: USN	MPv3		
Bridge	System Location	C. C	Detault Location			
Ethemet VLAN MAG Address Piltering Certificate Store	System Contact		Default Contact			
Radio 1 WLAN Mode: Client	SNHPv1/SNNP	r2c Configuratio	on (Community Settings, Trap	Receivers)		
PHY Setting	Community		Source		Access Control	
Client General Client Security	1				None 🕑	
IPsec Tunnel	2				None	
Status	3				None	
Services Settings	4				None	
SNMP Agent						
Secial Communication	Trap Receiver	Address	Community		Type	
Remote Administration	1				SNMPv1 Trap	
Web Server Admin Liser Management	2				SNMPv1 Trap	
List All Usors						
Add New User User Login Policy	SNHPv3 Config	iration (SNMPv	3 Users, Trap Receivers, Engin	e 10)		
Remote A&A Setup	SNMP Engine II			(10+64 boxed	acimal dicite)	
Monitoring/Reports				(10-04 Hexad	ocinia organi	
System Status Brideiras Status	2111 AT 2 11					
Bridging Sile Map	SNMMAT USET	arrie 5	MA Autoenecation Password	VES Encryption (Access Control	
Adjacent AP List DHCP Client List	1				None	
Logs	2				None	
System Log Web Access Lon	3				None 🔽	
Auditing Audit Configuration Audit Log	4				None	
System Administration	Trap Receiver	Address	SNMPv3 Usemame	SHA Authenticat	ton Password AES Encryption Passwo	bed
Radio Tx Off Control	1					
System Upgrade	2					

Figure 61: Services Settings - SNMP Agent

WiFiProtect 3e-520 series products support multiple versions of SNMP protocols: SNMPv1, SNMPv2c and SNMPv3. You can disable any SNMP protocols by unchecking the SNMP protocol version.

The **SNMP General Configuration** section shown in Figure 61 allows you to configure what versions of SNMPs to be enabled in the device, and enter **System Location** and **System Contact** information to help identify the device. Only SNMPv3 supports secure SNMP connections to the device, as described below.

When the SNMPv1, SNMPv2c protocol option is checked (enabled), you can enter the **Community Settings** and **Trap Receivers** in the **SNMPv1/SNMPv2c Configuration** section:

1) SNMPv1/SNMPv2c Community Settings:

- a) **Community** Authentication of clients is performed (only) by a "community string", in effect a type of password. It is transmitted in plain text, and so provides only limited security.
- b) **Source** The source IP addresses from which the SNMP manager is allowed to access this device.
- c) Access Control Defines the level of management interaction permitted. Select between None, Read Only (Get), and Read Write (Set). The default setting is None.

2) SNMPv1/SNMPv2c Trap Receivers:

- a) **Trap Receiver Address** The IP address of the trap receiver that receives SNMP traps from this device.
- b) **Community** The community string to be used in reporting SNMP traps.
- c) **Type** The type of SNMP traps. Select between **SNMPv1 Trap** and **SNMPv2 Trap**. The default setting is **SNMPv1 Trap**.

When the SNMPv3 protocol option is checked (enabled), you can set up secure SNMPv3 access in the **SNMPv3 Configuration** section.

3) SNMPv3 Engine ID:

- a) SNMPv3 requires that an application know the unique identifier (snmpEngineID) of the remote SNMP protocol engine in order to retrieve or manipulate objects maintained on the remote SNMP entity. You can enter a unique hexadecimal string of 10 64 characters for the Engine ID or leave this field to be blank so that the default Engine ID is used. The default Engine ID is automatically generated based on the device MAC address in accordance with RFC 2571:
 - i) First 4 octets The first bit is set to 1. The rest is the Internet Assigned Numbers Authority (IANA) Enterprise number 8433 assigned for 3eTI.
 - ii) Fifth octet Set to 3, indicating the MAC address is specified in the next 6 octets.
 - iii) Last 6 octets MAC address of the device WAN port.

4) SNMPv3 Secure Users:

- a) **SNMPv3 Username** The username for the secure SNMPv3 connection. A minimum of eight characters in length.
- b) SHA Authentication Password Only Secure Hash Algorithm (SHA) authentication is supported. Enter a cryptographically strong password, one that contains characters from all of the following 4 groups, and at least 2 of each group: uppercase letters, lowercase letters, numerals, and symbols found on the keyboard.
- c) AES Encryption Password Only Advanced Encryption Standard (AES) encryption is supported. Enter a cryptographically strong password, one that contains characters from all of the following 4 groups, and at least 2 of each group: uppercase letters, lowercase letters, numerals, and symbols found on the keyboard.
- d) Access Control Defines the level of management interaction permitted. Select between None, Read Only (Get), and Read Write (Set). The default setting is None.
- 5) SNMPv3 Trap Receivers:
 - a) **Trap Receiver Address** The IP address of the trap receiver that receives SNMPv3 traps from this device.
 - b) **SNMPv3 Username** The username for sending SNMPv3 traps. A minimum of eight characters in length.
 - c) **SHA Authentication Password** The SHA authentication password for sending SNMPv3 traps. Enter a cryptographically strong password, one that contains characters from all of the

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com



following 4 groups, and at least 2 of each group: uppercase letters, lowercase letters, numerals, and symbols found on the keyboard.

d) AES Encryption Password – The AES encryption password for sending SNMPv3 traps. Enter a cryptographically strong password, one that contains characters from all of the following 4 groups, and at least 2 of each group: uppercase letters, lowercase letters, numerals, and symbols found on the keyboard.

Click **Apply** to save configuration changes.

NOTE: Please contact 3eTI support for SNMP Management Information Base (MIB) information.

2.6.3 Serial Port (3e-523N Only)

The **System Configuration – Serial Port** screen (Figure 62) allows you to control the type and format of the serial data to be transmitted and received, if the serial port is available, enabled and configured. Refer to section Appendix C. Serial I/O Interface Board for wiring definition of the signals.

NOTE: To enable this serial port, you must enable serial communications under **Services Settings – Serial Communication** (Figure 63).

System Configuration ->	Serial Port
Interface Type	RS-232 •
Duplex (RS485 only)	Full-Duplex 💌
Data Rate	115200 💌
Data bits	8 💌
Parity	None 💌
Stop bits	1 💌
Flow control (RS232 Only) Apply	None 💌

Figure 62: System Configuration – Serial Port

Table 14: Service Settings – Serial Communication						
Settings	Options	Description	Default			
Interface Type	RS-232 RS-422 RS-485	Select the interface type for the serial I/O port.	RS-232			
Duplex (RS485 only)	Full-Duplex Half-Duplex	In full duplex mode data is transmitted and received simultaneously. In half duplex mode data is transmitted or received but not at the same time.	Full Duplex			
Data Rate (bits per second)	115200 57600 38400 19200 9600 4800 2400 1200	Select the data rate required.	115200			
Data bits	8 7 6 5	Select the number of data bits to be transmitted or received.	8			
Parity	None Odd Even	Select parity to be used.	None			
Stop bits	1 2	Select number of stop bits to be used.	1			
Flow control (RS232 only)	None Hardware	When hardware flow control is selected, RTS and CTS are used.	None			

2.6.4 Serial Communication (3e-523N Only)

The **Services Settings - Serial Communication** section (Figure 63) displays the status and configuration of the current serial port mode of operation. You can choose to **Enable** or **Disable** (default) serial communications and select a serial **Port Profile**.

NOTE: See System Configuration – Serial Port Section 2.6.3 to configure the serial interface.

2.6.4.1 Raw Socket

This is the default setting, and allows serial devices connected to two 3e–523 devices to communicate across the network. It supports bidirectional, multiple unicasting and peer-to-peer communications. Enter MAC addresses (formatted as 00:11:22:33:44:55) for the radios of the remote nodes that will communicate via the Raw Socket.

Services Settings -> Seria	Communication
Configuration	
C Enable O Disable	
Low Latency. Enable 💌	
Port Profile: Raw Socket	
Destination radio MAC addreses.	MAC format is 00:11:22:33:44:55
Remote Node Mac 1:	
Remote Node Mac 2:	
Remote Node Mac 3:	
Remote Node Mac 4:	
Remote Node Mac 5:	
Remote Node Mac 6:	
Remote Node Mac 7:	
Remote Node Mac 8:	
Remote Node Mac 9:	
Remote Node Mac 10:	
Apply	

Figure 63: Services Settings – Serial Communication – Raw Socket

2.6.4.2 TCP Socket

This Port Profile provides a direct IP connection using Transmission Control Protocol (TCP). When using TCP sockets your serial server **Operation Mode** can be configured as a **Server** (default) or **Client** (Figure 64).

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

If the 3e–520 series device is configured as a TCP **Server**, other network devices can initiate a TCP connection with the serial device connected to the serial port. Network devices initiating connections must be configured with the IP address of the serial device and the TCP port number associated with its serial port. The **TCP Port** can be configured as needed in this screen (default is 18000).

If the 3e–520 series device is configured as a TCP **Client**, it will automatically establish a bidirectional TCP connection between the serial device and a server (or other networked device); enter the associated **Remote IP Address** (default 192.168.254.254) and **TCP Port** to configure this connection.

Client	
lowing network service:	
192.168.254.254	
18000	
	Client Nowing network service: 192.168.254.254 18000

Figure 64: Services Settings – Serial Communication – TCP Socket

2.6.5 Remote Administration

The Services Settings – Remote Administration Access Control screen (Figure 65) allows you to set up access control policies for remote administration via HTTPS, SNMP and Internet Control Message Protocol (ICMP) protocols. In the factory default configuration, the **Remote Administration Access Control** option is set to **Disable Filtering** and hence remote administration is allowed for any source IP address and MAC address.

When the **Remote Administration Access Control** option is set to **Enable Filtering**, the device validates the source IP and/or MAC addresses of administrative queries and control requests to ensure that the message request is from an approved node. This enables secure remote administration from selected IP and/or MAC addresses.

Contractory of Contra	3eTI AirGuard	" Wireless Acce	ss Point		
Ultro Jori	Usemate: Role:	CryptoOfficer 3e-locat	in R	ost Name: adlo Regioe:	detaut United States
System Configuration General Noty Charnel Control SIAN	Services Setta Renole Administra	ngs -> Remote Ac	Iministration Access C CEnable Fittering ®Drad	Control le Filenny	Californians.
Entration VI AN	Policy	Protocol	Management FC IP		Management PC MAC
MAC Address Filenag Cardificate Store	: .	A8 💟	and the second second		
Redio 1	2 4	A# 🛩			
PRV Setting	3	14 1			_
Cent General Cent Securiti	4	All 🖌			
Pher, Tannell	5	Al 🖌			
Balas	4. 🖌	Ad 🖌			
DHOP Service	7	All M			
SMMP Agent	8	Al V			
Serve Communition	100000000000000000000000000000000000000			and second second	
YYER DUTYN	Minutes:	CODELIE, "ALL, X WE SHAR	The became increased are built	PS, ShuP or	CAVP protocors from the specified P* and/or WAC.
Admin. Unar Management Last All Users Add New User User Legis Policy Discrete 414 States	If built IP and MAC	aldresses are not to	extind in the access cantrol y	othey it is a m	dots all filler and applies to all sequent packets.
Two Factor Auth	100000000000000000000000000000000000000				12
NontoringStyperm Bysten Bates	A social and partial of	Initial Pulicy	Delice D De Access Control	and associated	shows, true by one statistic from the policy 1 P.4
Bridging Status Bridging Stin Mag Adjatest AP List DHCP Client List	matches to a polic match to any polic	y, an action is taken to y, the default access of	arcept or drop the packet. If ontrol policy is applied	f passes down	through the Access Control List and doesn't
Logs System Log Web Assess Log	Default Access Co	ntral Policy: ACCEI	1		
Auditing					
Audit Log System Administration	4444				

Figure 65: Services Settings — Remote Administration Access Control

Up to 8 Access Control policies can be specifically defined in the Access Control List, with the following fields for each entry:

- **Policy ACCEPT** or **DROP** a request packet if it matches this access control policy.
- **Protocol** The protocol field can be **HTTPS**, **SNMP**, **ICMP** or **All**. If the protocol field is "All" (default), the access control policy is used to match for all HTTPS, SNMP and ICMP protocols.
- **Management PC IP** This is the actual IP address of a management Personal Computer (PC), which shall be an IPv4 unicast IP address in the form xxx.xxx.xxx, such as 192.168.202.100.
- Management PC MAC This is the actual MAC address of a management PC, which shall be a
 unicast MAC address in the form XX:XX:XX:XX:XX, such as 00:23:69:24:BF:80. You can also
 enter a MAC address in the form XX-XX-XX-XX-XX as it is displayed in the output of the
 Windows "ipconfig /all" command; it will be automatically converted to the form
 XX:XX:XX:XX:XX:XX.

NOTE: The management PC and the device must be connected to the same LAN if MAC address filtering is used.

For each access control policy entry, both **Management PC IP** and **Management PC MAC** fields are optional. You can leave either or both of the fields blank, with the following application cases:

• Management PC IP Only – This is an IP address filter. Only the source IP of request packets is checked.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

- Management PC MAC Only This is a MAC address filter. Only the source MAC of request packets is checked.
- Management PC IP and MAC This is an IP address and MAC address filter. Both the source IP and source MAC of request packets are checked.
- Both Management PC IP and Management PC MAC are unspecified This is a match-all filter. The access control policy applies to any request packet.

You can select the Default Access Control Policy to be one of these two values:

- **ACCEPT** Always accept a request packet when it does not match any access control policies defined in the Access Control List. This is the default value.
- **DROP** Silently drop a request packet when there is no specific access control policy which matches it.

The ordering of access control policies is important. A request packet is checked against each policy in the Access Control List in turn, starting at policy 1, and if it matches that access control policy, then an action is taken to accept or drop the request packet. If it passes down through the Access Control List and does not match to any policy, the **Default Access Control Policy** is applied.

When you click the **Apply** button to save your setup, a confirmation dialog box will pop-up reminding you to double check your setup to ensure that your configuration allows remote administration from selected IP and/or MAC addresses, including the IP/MAC of your current management PC.

2.6.6 Web Server

The **Services Settings – Web Server** screen (Figure 66) allows you to use the uploaded certificate for a secured Web Server access.

A Web Server Certificate can be uploaded from **Certificate Store** under **System Configuration**. After loading a certificate, from the **Web Server** page, then you can select the preloaded web server certificate from the dropdown list and click on **Apply**. The default web server certificate cannot be deleted.

1110-00		3eTI AirGu	ard [™] Wireless Access P	oint		Logad
Ulfra	3eTI	Usemame:	CryptoOfficer	Host Name:	default.	
ELECTROVACS.	and the second s	Role:	3e-local	Radio Region:	United States	
System Configurat General Noisy Channel Co WAN LAN Bridge Ethamet VLAN MAC Address Filt Cerificate Store Radio 1 WLAN Mode Clie PHY Satting Client General Client Security Place Tunnal Profiles Status Servicas Settings DHCP Server Status Servicas Settings DHCP Server Serial Port Serial Communica	ion introl ering nt	Services So Web Server (Apply	ettings -> Web Server Certificates: Device Default C US Maryland Uit	iertificate ra-3eti Testing device1 device1@4	3eti.com	

Figure 66: Services Settings — Web Server

2.7 Admin User Management

Users can access the device through either dedicated local Ethernet port or the WAN Ethernet port. The Management UI treat users identically regardless of the access ports.

2.7.1 List all Users

The **Admin User Management – List All Users** screen (Figure 67) lists the CryptoOfficer and any accounts configured for the unit. You can edit or delete users from this screen. You can also unlock the account which was locked.

Contraction of the second	3eTI AirGuar	d ¹⁴ Wireless Access Point			La Maria			Logat		
Ultra 30TI	Usersene: Role:	CyptoOfficer 3e-local		Host Name: Radio Region	UVT2 United States					
Summer Configuration	Admin User	Admin User Management -> List All Users								
General Note Channel Control	Over 10	Nete	Status	Reason	Pellup	-				
80204	C/ypioOfficier	Default Crypto Office	Active	Normal	N/A	Edt	Detete	UTRICK .		
Even	administrator		Locked	Bed passwd	N/A	Edit	Delotar	Unice		
Elfornet/CAN	gichel/98		Active	Normal	N/A	Cdt-	Delete	Untock		
Carlificate Store	Testanier1		Licked	Bad paidwell	74/A	Edt	Delete	UNRAS.		
Haito 1	Testuser2		Active	Normal	NEA.	Edit	Coleta	Unicell		
Weeken Mach Germal Autr Britge Becarts Protect Tarmel Protect Structure Settings OHCP Terrier Structure Settings OHCP Terrier Structure Settings Results Advances Web Server Acting Users Chart Login Policy Research Auto, Starty Auto, Start Setting Cher Login Policy Research Auto, Starty Auto										

Figure 67: Admin User Management — List All Users

If you click on **Edit**, the **Admin User Management - Edit User** screen (Figure 68) appears. On this screen you can edit the **User ID**, **Password**, and **Note** fields. Default values are "CryptoOfficer", "CryptoFIPS", "Crypto Officer" and "Default Crypto Office" respectively. If the password complexity is enabled, your password should come from all of the following 4 groups, and at least 2 of each group: uppercase letters, lowercase letters, numerals, and symbols found on the keyboard.

The Password Complexity Check and the Minimal Password Length are established on the **Admin User Management – User Login Policy** screen (Figure 71).

a stance	3eTI AirGuard	I™ Wireless Acc	ess Point			Logost
Ultra 3eTi	Usemane:	CryptoOfficer		Host Name:	default	
ELECTROPECS	Role:	3e-local		Radio Region:	United States	
System Configuration General Noisy Channel Control WAN LAN Breige Ethemet VLAN MAC Address Filtering Certificate Store Redio 1 WLAN Mode: Client PHY Setting Client General Client Security IPsec Tunnel Physikes	Bele: Admin User N User ID minimum Password comple Definition of a con uppercase letters, Please change th User ID: User ID: User New Passwo Password Lockow Maximum Bad Pa	3e-local lanagement -> Ed length: 8 http://deck.OFF splex password: a pas lowercase letters, nu le password followin xd. xd. Again: LPeriod: ssword Attempts:	dit User sword that contains charact merais, and symbols found ig the password complexit CryptoOfficer	Radio Region: ters from all of the t on the keyboard ty rule.	United States following 4 groups, and at least 2 of eac	h group:
Struce Services Settings DHCP Server SIMIP Agent Serial Communication Remote Administration Web Server	Note: Password Aging: Update Rese	٤	Default Crypto Office O Enable			

Figure 68: Admin User Management — Edit User

2.7.2 Add New User

The Admin User Management – Add New User screen (Figure 69) allows you to add new users with 3elocal role and configure the associated User ID, Password, and Note fields. To ensure password complexity, your password should come from all of the following 4 groups, and at least 2 of each group: uppercase letters, lowercase letters, numerals, and symbols found on the keyboard.

1.11	3eTI AirGuar	d™ Wireless Acco	ess Point			Logost
Ultra 3eT	Usernome:	administrator		Host Name:	default	
ELECTRONICS	Role:	3e-local		Radio Region:	United States	
System Configuration General Noisy Channel Control WAN LAN Bridge Ethemet VLAN MAC Address Filtering Certificate Store Radio 1 WLAN Mode: Client PHY Setting Client General Client Security Proc. Turnel Profiles Status Services Settings DHCP Server SMMP Agent Services Settings DHCP Server SMMP Agent Service Server SMMP Agent Service Administration Remote Administration Remote Administration Remote Administration Web Server Admin Univer Nanagement List All Lines Add New Univ Discr Login Policy Remote Administration Fundas	Admin User N User ID minimum Password comple Detintion of a cor- uppercase letters Please change the User ID: User Password Lockou Moximum Bad Pa Note: Password Aging: Add Reset	Management -> Ac length: 8 suty check: OFF molex password a pas lowercase letters, nur he password followin gain: if Period: isoword Attempts:	Id New User sword that contains charact merals, and symbols found g the password complexi g the password complexi 0 (0 ~ 90)Minutes 3 Attempts 0 Enable I Disable	ers from all of the on the keyboard by rule.	following 4 groups, and	l at least 2 of each group:

Figure 69: Admin User Management — Add New User

- User ID: User-defined unique user identification string
- User Password: User-defined user authentication string. Password can be composed of any combination of upper- and lower-case letters, numbers, and the following special characters: "!", "@", "#", "\$", "%", "^*, "&", "*", "(" or ")".
- **Password Lockout Period:** The password lockout period defines the number of minutes the user must wait after failing the maximum bad password attempts before he/she can try logging in again. The default value is 10 minutes
- **Maximum Bad Password Attempts:** The maximum bad password attempts define the number of failed authentication attempts before the lockout period is triggered. The default value is 3 attempts.
- **Password Aging:** By enabling the password aging feature (Figure 70), there will be a maximum time limit for the user's password after which the user will be required to change it. The default value is disabled.

1 Marcola 1	3eTI AirGuar	d ¹⁴ Wireless Acce	ss Point		Legast
Olfra 3eTl	Username:	administrator		Host Name:	default
turchonecs	Role:	3e-local		Radio Region:	United States
	Admin User M	lanagement -> Ad	d New User		
System Configuration General	Door ID minimum	locath 9			
Noisy Channel Control	Payword comple	why check: OFF			
WAN	Definition of a cor	nplex password: a pass	word that contains charact	ters from all of the t	following 4 groups, and at least 2 of each group:
Bridge	uppercase letters	lowercase letters, nur	erals, and symbols found	on the keyboard	
Ethernet VLAN	Please change t	he password following	the password complexity	ty rule.	
MAC Address Pittering Certificate Store	11			_	
Radio 1	User ID:				
WLAN Mode: Client	User Password				
Client General	User Password A	gain			
Client Security	Password Lockou	t Period:	10 (0 - 50)Minutes		
Profiles	Movimum Bad Pa	seword Attempts:	3 V Attempts		
Status	Linter		a 🔄 rassides	_	
Services Settings	NOTE:				
SNMP Agent	Password Aging:		Enable O Disable		
Serial Port	Maximum Passw	ord Age:	30 (30 ~ 90) Days		
Remote Administration	Warning Before I	assword Expiration:	7 (2 ~ 90) Days		
Admin User Management	Password Grace	Period:	3 V Days or 3	 Logins 	
Add New User User Login Policy	Password Unique	mess Depth:	3 V Passwords		
Remote A&A Setup Two-Factor Auth	Add Reset				

Figure 70: Password Aging

- **Maximum Password Age:** If this account is inactive for this number of days, the account will be considered as disabled. The user will be forced to change his password to re-enable the account. The default password age value is 30 days.
- **Warning Before Password Expiration:** The user will be warned this number of days before his/her password expires. The default value is 7 Days.
- **Password Grace Period:** The user will have a grace period of this number of day or successful logins after the password expires in order to change the password. After the grace period, the user interface will be locked except for the Edit User page forcing the user to change the password. The default grace period is 3 Days or 3 Logins.
- **Password Uniqueness Depth:** The user will be prevented from using this number of pervious passwords. The default value is 3 previous passwords.

2.7.3 User Login Policy

The **Admin User Management - User Login Policy** screen (Figure 71) allows you to configure options governing user login policy and associated device functionality. The "User Login Policy" applies to all local users.

1000 COL	3eTI AirGuard	Wireless Access Point	t	Los
Girco 3eTi	University :	admenistrator	Hoat Name:	default .
and the second second	Hote:	lie-local	Redio Region:	United States
System Configuration Centeral	Admin User Ma	inagement -> User Logi	n Policy	
Noley Glarmel Centrel WAN	Pasaword Comple	anty Chura		
LAN Bridge Etheren VLAN MAC Address Filtering Conflictle Three	Definition of a comp opportable letters, in O Enable @ Dv	tex parswort a password that overcase letters, numerals, are sable	contains characters from all of the Engenbols found on the keyboard	blowing 4 groups, and at least 2 of each grou
Radio 1	6.44 Contractor 10	42.3		
WLAN Model Client PHY Setting	Additional Option	li.		
Gert Brouty	Mamon Parisett	t Length	8 (0 - 30) charact	F
Waey Tuning Profiles	Login Session Tim	sout	50 (3 - 60) minutes	
Natus	Osolav Successful	Login Count in Past	7 (1 + 395) data	
DITCF Server		and an inclusion of the second	b. As constants	
ShillP Agent Genid Pret Senid Communication	Account Lockout	Email Notification		
Remote Alteriolatization Web Server	O Enable: @ E	satie		
Admin Uker Management List Al Users	E-mail address.	1		.0.
Mary Looke Pelicy Herrican Adv. Settig	Appy			
Teb-Fatter Auto				

Figure 71: Admin User Management — User Login Policy

You can choose to Enable or Disable (default) the Password Complexity Check. The definition of a complex password is a password that contains characters from all of the following 4 groups and at least 2 of each group: uppercase letters, lowercase letters, numerals, and symbols found on the keyboard.

Additional Options can be configured:

- **Minimum Password Length:** Enter a value between eight (8) and 30 characters; the default is 8. Note that password lengths of up to 30 characters are supported. Passwords that are longer than 30 characters will be truncated when creating a user or changing passwords for an existing user,
- Login Session Timeout: Enter a value between three (3) and 60 minutes; the default is 10. If the admin user session is inactive for more than the timeout amount, then the login session automatically terminates.
- **Display Successful Login Count in Past:** Enter a value between one (1) and 365 days; the default is 7 days. The successful login count will be displayed within the period of the time is set after successful login to the Web GUI.

NOTE: If you make changes to the password rules, all users will be required to change their passwords, even those users' passwords that already meet the new password rules. Because the password information is encrypted, the device cannot determine whether any existing passwords meet the new criteria.

The default for the **Account Lockout Email Notification** is set to "Disable". If enabled, the system will send an email to the email address listed to inform that person that a user has been locked out of the system. To configure the email notification settings, go to the **System Administration – Email Notification Configuration** screen (Figure 87).

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

2.8 Remote Authentication and Authorization

The 3e-520 series devices offer both local and remote user authentication. Local authentication is done via username and password. When the user accesses the 3e-520 series device URL, he/she inputs username and password, the 3e-520 series device authenticates the user with its local on-device username/password then authorizes the user based on local user privilege policy.

When there are multiple 3e-520 series devices in a deployment, it becomes cumbersome to manage user add/delete/change password locally on each device. For this type of scenario, the 3e-520 series devices offer a centralized authentication and authorization framework. User's authentication and privileges can be centrally managed at enterprise level and uniformly enforced. The 3e-520 series device uses the Lightweight Directory Access Protocol (LDAP) to authenticate username and password against an Active Directory Server.

The 3e-520 series uses LDAP over TLS protocol (LDAPS) for a secure connection to the remote server and verifies the server identity via a trusted root CA. The LDAP server's trust anchor must be loaded into the 'Certificate Store' as shown in Section 2.3.9. Additionally, an IPsec tunnel can be setup to further secure the connection between the 3e-520 series devices and the Active Directory server.

2.8.1 Remote A&A Setup

To reach the 3e-520 series 'Remote Authentication & Authorization Setup' page click on the 'Remote A&A Setup' menu under the Admin User Management sub-heading on the left panel of web GUI. (See Figure 72) Click the 'enable' option on the 'Remote Authentication & Authorization Feature' section. Next, under the 'LDAP Server Configuration' section, enter the LDAP server URI (hostname or IP address), and optional port number if LDAP server is running on non-standard port. e.g. ldaps://openldap.ultra-3eti.com:999

If an IPsec tunnel connection to the LDAP server is required, click on the 'IPsec Tunnel' checkbox. Select an IPsec tunnel profile from the drop-down window if a profile had been created (See Section 0 **Error! R eference source not found.**). Also, if required, check the 'Specify IPsec ID' checkbox and enter the ID in the next field down.

Next, enter User Search Format, the search format consists of the Fully Qualified DN as a flexible regular expression. The 3e-520 series device binds to the LDAP server using this distinguished name after substituting the '*' character with the username provided on the login screen. For example,

```
"uid=*,ou=people,dc=ultra-3eti,dc=com"
```

The 3e-520 series device would replace the "*" in this search string with the username input from web GUI login page to construct a bind DN.

"uid=john,ou=people,dc=ultra-3eti,dc=com"

LDAP server will return "*Invalid Credential*" if the username and password combination does not match any record on server. Or it will return the group membership that user belongs to.

e.g. *memberOf:* cn=3e-local,ou=Group,dc=ultra-3eti,dc=com

Click **Apply** to save your selections.

and a second second	JeTI AirGuard	Wireless Access Point	155122	1.55M
Citro 3eTi	Username: Rojej	CryptoOfficer 3e-focal	Rost Name: Radio Region	UUT1 United States
System Configuration Demonit Natus Channel Gambol Natus Channel Gambol Natus Channel Gambol Dimension Configuration District State Hardin 1 District State Hardin 1 District State Hardin 1 District State Hardin 1 District State District State Dist	Role: User Managen Enable O Note: Remote A&A Note: Remote A&A The LDAP server certificate LDAP Server UR Phas: Turnel Unar Search For	Series Serie	Radio Region on & Authorization Set al MOMT* port when "Local le. Don't forget to load trusted hostname o e.g. cn=",au	United States Inp MOMT* e in default 192,108,15.0 network, Froot GA in "Certificate Store" to verify server r IP: Enclose IPv6 (n.)] I=people;dc=example;dc=coti
Web Barvet Atmite User Management Lef Al Doors And New User Last Loop Colleg Receive Ad-Sales Receive Ad-Sales	Activ			

Figure 72: Admin User Management – Remote A&A Setup

2.8.2 Remote A&A User Groups

As discussed in Section 1.4.1 'User Roles', any of three roles can be assigned to Remote A&A users: '3e-local', '3e-CryptoOfficer', and '3e-administrator'. Users/roles created at the centralized management server are independent of users created locally at the 3e-520 series device.

Three role groups can be created at the centralized management server, users can then be assigned to the appropriate group. Table below describes User Groups/Roles and shows a list of major privileges.

LDAP Group	Privilege			Note	
	Local User Database Manipulation	LDAP Server Setting	Encrypt Algorithm/Key, etc. Security Related Setting	Other	
3e-local	Y	Y	Y	Y	Small number of users, e.g., device owners.
3e-CryptoOfficer	N	N	Y	Y	Most device maintainers fall in this group.
3e-administrator	N	N	Ν	Y	Users no "need-to-know" security related setting

Table 15: User Group Role & Privilege in LDAP Server

2.8.3 LDAP Server Configuration

An Active Directory / LDAP server can be configured to the remote management of user authentication. To set up the Active Directory / LDAP server, you need to create three (3) user groups in Active Directory:

- 3e-local,
- 3e-CryptoOfficer, and
- 3e-administrator.

The remote user will be associated with one of these three (3) groups in the 'memberOf' attribute. The 3e-520 series device would authenticate the remote user's username & password against the LDAP server by querying the 'memberOf' attribute of the user. The group info ('memberOf' attribute) of the user will be returned from the LDAP server to the 3e-520 series device to grant the privilege of the user for remote management access.

2.9 **Two-Factor Authentication**

2.9.1 **Two-Factor Authentication Overview**

The remote management of 3e-520 series devices can be configured with Two-Factor authentication by utilizing CAC (Common Access Card) authentication and Remote LDAP Server authentication. The first factor authentication uses CAC certificate and pin authentication via trusted chain CAs and OCSP responder server. The second factor is to use CAC username & password to authenticate via LDAPS to the remote LDPS server.



Figure 73: Two-Factor Authentication Overview

User with a valid CAC is authenticated to gain access to the 3e-520 series device web management interface. The 3e-520 series device optionally uses the Online Certificate Status Protocol (OCSP) to query OCSP Responder for CAC revocation status. Only if OCSP responder replies with a "good" certificate status would 3e-520 series device allow the authentication process to continue.

The following components are required to implement and operate a PKI network providing Two-Factor authentication of 3e-520 series device, see Figure 73 for the components that make up a PKI network.

2.9.1.1 Web Client

To Access Web GUI remotely with CAC authentication, the following requirements should be configured in the Web Client and 3e-520 series device.

- Only Windows Internet Explorer / Edge Web Browser supports CAC authentication.
- ActiveCard software and SmartCard reader driver are installed in the Web Client.
- The SmartCard service is enabled and running (from Windows Services).
- The Certificate Propagation service is enabled & running (from Windows Services).

2.9.1.2 OCSP Responder

If "Enforce OCSP check" is enabled on 3e-520 series device, an OCSP responder is required to be installed and setup in the network. Software such as Tumbleweed Validation Authority running on Windows Server platform can be configured as an OCSP responder. DoD CA-23 certificate and any corresponding CRL can be loaded to the OCSP Responder from a local file. Current DoD certificates and CRLs can be obtained from https://crl.chamb.disa.mil/.

2.9.2 First-Factor Authentication—CAC and OCSP

Before enabling Two-Factor Authentication, the Remote A&A needs to be enabled first. You also need to upload the certificate trust chain CAs (intermediate CA and Root CA) of the CAC to the Certificate Store of the 3e-520 series device. If the OCSP Responder is used, you need to upload the OCSP Signer Certificate to the device's Certificate Store as well.

To reach the 3e-520 series 'Two-Factor Authentication Setup' page click the '**Two-Factor Auth**' under the **User Management** sub-heading on the left panel of web GUI. See Figure 74: Admin User Management – **Two-Factor Authentication** below. Click the 'enable' option on the 'Two-Factor Authentication' Section. Next, under the 'CAC OCSP Responder Configuration' Section, select either Enforce or Skip OCSP Check. If Enforce OCSP Check is selected, select either Nonce is been used or not. Then Enter Primary URL and/or Secondary URL of the OCSP Server. Select a previous uploaded OCSP Responder Certificate from the list then click on Apply.

Labora -	3eTI AirGuard	* Wireless Access Point		Legeut			
Olfra 3eTi	Username:	CryptoOfficer	Host Name:	default			
ELECTRONICS.	Role:	3e-local	Radio Region:	United States			
System Configuration General Noisy Channel Control	Services Settin	gs -> Two-Factor Authenticatio	n				
WIAN	Two-Factor Authe	ntication: CAC + Remote LDAP					
LAN Bridge Ethernet VLAN	Enable	O Disable					
MAC Address Filtering Certificate Store	Note: CAC and Ren network.	kote: CAC and Remote LDAP are ignored in login session from "Local MGMT" port when "Local MGMT" is in default 192.168.15.0 where the second s					
Radio 1							
PHY Setting	CAC OCSP Respon	der Configuration					
Client General Client Security	Enforce OCSP	Check O Skip OC	SP Check				
Psec Tunnel Profiles	O Use Nonce	Do Not Use Nonce					
Status	Primary URL:	10.210.39.101					
Services Settings DHCP Server	Secondary URL:	20.220.19.99					
SNMP Agent Serial Port Serial Communication	Responder Certificate: US Maryland Ultra-3eti Testing ocspd ocspd@3eti.com						
Remote Administration Web Server	Apply						
Admin User Management List All Users Add New User User Login Policy Remote A&A Setup Two-Factor Auth							



2.9.3 Second-Factor Authentication—LDAP Server

After the CAC authentication is granted, the username in the CN attribute of the CAC certificate along with its password will be sent to the LDAP server for authentication. If the authentication succeeds, the user group membership is retrieved from the LDAP server and used as privilege information for the remote

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

management. An IPsec tunnel can be used to further secure the LDAPS connection. Please refer to Section 2.8 Remote Authentication and Authorization for more detail.

Note: Two-Factor Authentication configuration change will take effect after device reboot.

2.10 Monitoring/Reports

This section gives you a variety of lists and status reports. Most of these are self-explanatory.

2.10.1 System Status

The **Monitoring/Reports - System Status** screen (Figure 75) displays the status of the unit, system status information include:

- System Uptime,
- Total Usable Memory size,
- Free Memory,
- Current Processes,
- Door Status (Open/Close status is detected by serial port connection),
- CPU Info,
- WAN Ethernet MAC address,
- LAN Ethernet MAC address,
- Radio 1 & 2 MAC address, and
- Routing Table.

1 mile	3eTI AirGua	rd™ Wireless Access Poin	nt		Lington
Ultra 3eT	Username:	CryptoOfficer	Host Name:	0072	
ELECTRONICS	Role:	3e-local	Radio Region:	United States	
Section Configuration	Monitoring	Reports -> System Status			
General	Device Statu				71
Noisy Channel Control	System Uptime	EX 257	0:820		
LAN	Total Usable N	Aemory Size:	261169152 bytes		
Endge Ethorent M. (A)	Free Memory		130633728 bytes		
MAC Address Filtering Certificate Store	Door Status	3969.	Open		
Radio 1	1200302000	1.000			
WLAN Wode: AP PHY Setting AP Departs	Other Informat	ion: CPU Info			
AP Security	Network Inte	erface Status			12
AP Wireless Clients Wireless VI AN Montres	WAN Ethernet	MA/G address	00 07 D5 04 BE 5C		
Psec Tunnel	LAN Ethernet	MAC address:	00.07.D5.04:BE 28 00.0F 8E 6E CA:D9		
Pioffes	nation i meno i	autii 610	00.00.00.01 (M.D.)		
Services Settings	Rosting Tabl	8			
DRCP Server	Dest, LAN IP	Subret Mask	Default Gateway	Hop Count	Interface
Remote Administration Wah Server	192.168 15.0	255 255 255 0	S. 1997 (1997)	0	eth0
Admin User Management					
Add New User					
User Login Policy Remote ARA Setup					
Two-Factor Auth					
Monitoring/Reports					
Blidging Status					
Bidging Site Map					
DHCP Client List					

Figure 75: Monitoring/Reports — System Status

2.10.2 Bridge Status

The **Monitoring/Reports - Bridge Status** screen (Figure 76) displays the Spanning Tree Protocol (STP) status for each port, as well as associated **Bridge Information** for the device.

Ethernet Link STP Status: This is for the Uplink Ethernet port of the device. This screen displays current STP parameters for this port.

Bridge Information: This screen displays current STP parameters for the bridge link.



Figure 76: Monitoring/Reports — Bridge Status

2.10.3 Bridge Site Map

The **Monitoring/Reports – Bridging Site Map** screen (Figure 77) shows the spanning tree network topology of both wired and wireless nodes connected to the network. The root STP node is always on top and the nodes of the hierarchy are displayed below it. Wired links are double dotted lines and wireless links are single dotted lines.

NOTE: This map does not update dynamically. You must press the Update button to refresh the map.



Figure 77: Monitoring Reports — Bridging Site Map

2.10.4 Adjacent AP List

The **Monitoring/Reports – Adjacent AP List** screen (Figure 78) shows all the APs on the network. These APs are detected by the device's radio(s). The list of APs is only within the band that can be seen from a particular channel. For example, if the AP is on channel 1, it will display APs on channels 1-3.

a sta	3eTI AirGua	rd™ Wireless Acces	s Point						Logout
Ultra 3eTi	Username:	CryptoOfficer		Host Name:	UUT2				
ELECTRONICS	Role:	3e-local		Radio Region:	United State	15			
System Configuration General	Monitoring/R	eports ->Adjacent A	P List						
Noisy Channel Control									
LAN	Radio 1		OCT D	Channel	Sanal(day)	Turne	toologi	MED	Match
Bridge	1 00:0e:8e	(6f ca d3/SparklanCo)	3510	36	-20	AP	Age(105)	Y	Y
MAC Address Filtering	2 5c:b0:66	40.9d f3(UNKNOWN)	SBG6580-2-0878F	1	-39	AP	14	Y	N
Certificate Store Radio 1									
WLAN Mode: Mesh Point									
PHY Setting Wireless Mesh General									
Auto Bridge Security									
Profiles									
Status									
Services Settings DHCP Server									
SNMP Agent									
Web Server									
Admin User Management									
Add New User									
User Login Policy Remote A&A Setup									
Two-Factor Auth									
Monitoring/Reports System Status									
Bridging Status									
Bridging Site Map Adjacent AP List									
The Prest at									

Figure 78: Monitoring/Reports — Adjacent AP List

2.10.5 DHCP Client List

The **Monitoring/Reports - DHCP Client List** screen (Figure 79) displays all clients currently connected to the unit via the DHCP server, including their reference **Index**, **Hostname**, **IP Address**, and **MAC Address**. It also indicates an **Expired at Date** for the client, based on the displayed DHCP server lease period.

The DHCP Client list constantly collects entries. To remove entries from the list, check mark the **Revoke Entry** selection and click **Remove** to confirm the action.

a star	3eTI AirGu	ard™ Wireless	Access Point			ما	gout
Ultra 3eTI	Username:	CryptoOfficer		Host Name:	UUT2		
ILICTROVICS	Role:	3e-local		Radio Region:	United States		
System Configuration	Monitoring	/Report -> DHC	P Client List				
General	DHCP server	lease period: 1 Day					
Noisy Channel Control	Index I	lostname	IP Address	MAC Address	Expired at	Remove	
LAN							
Bridge							
MAC Address Filtering							
Certificate Store							
Radio 1 VII AN Mode: Meab Point							
PHY Setting							
Wreless Mesh General Auto Bridge Security							
Psec Tunnel							
Profiles							
Status Services Settings							
DHCP Server							
SNMP Agent Remote Administration							
Web Server							
Admin User Management							
Add New User							
User Login Policy Departs ALA Salars							
Two-Factor Auth							
Monitoring/Reports							
Bridging Status							
Bridging Site Map							
DHCP Client List							
Derior Greek Link							

Figure 79: Monitoring/Reports — DHCP Client List

2.11 Logs

2.11.1 System Log

The **Logs – System Log** screen (Figure 80) displays System Facility Messages with a Date-Time stamp. These are messages documenting functions performed internal to the system, based on the system's configured functionality. The system log records all the general system activity, operation and alerts information during the device running. For example, the system log records the device boots up, software version, management access and self-test record, etc. Generally, the administrator would only use this information if trained as, or working with, a field engineer or if gathering information to be provided to technical support.

The System Log (syslog) continues to accumulate listings until it reaches its 256KB size limit. The log rotates by overwriting the first 1/3. For security reasons, the system log cannot be cleared.

If you choose to **Export** the syslog, it can be opened or saved as an HTML document.

1.11	3eTI AirGuar	rd™ Wirel	less Access Point			Logost
Olfra 3eT	Username:	CryptoOf	ficer	Host Name:	UUT2	
ELICTRONICS	Role:	3e-local		Radio Region:	United States	
	Logs -> Syst	em Log				Export
System Configuration						
Noisy Channel Control	Date-Time		System-Facility-Message			
WAN	May 10 17 03 11	2020	EST5 523NBR212 user.err master0	aemon: md_write_re	esponse id 64 rw /tmp/mdc	hildstatus61: -2, 0.000
LAN	Mar 19 17:03:11	2020	273 sec			
Bridge	Mar 19 17:03:11	2020	EST5 523NBR212 user.err master0	Daemon: error: childS	tatusUpdate 0 PID=636 <i< td=""><td>klogd></td></i<>	klogd>
Ethernet VLAN	Mar 19 17:03:11	2020	EST5 523NBR212 user err master0	aemon: error -2 writ	ing to /tmp/mdchildstatus6	1, closing
MAC Address Filtering Cartificate Store	Mar 19 17:03:11	2020	EST5 523NBR212 user info sysinit:	start_stop_sntp stop		
Radio 1	Mar 19 17:03:11	2020	EST5 523NBR212 user err sysinit:	start_stop_sntp_asyn	no stop no input conf	
WLAN Mode: Mesh Point	Mar 19 17:03:11	2020	EST5 523NBR212 user info auditor	gd: An email on audit	t log alert to guan-jung che	n@ultra-3eti.com has
PHY Setting			been successfully queued for delive	ry		
Wireless Mesh General		2020	EST5 523NBH212 user err auditlog	d: Failed to send trap	o data to subagent1: <tra< td=""><td>p><oid>.1.3.6.1.4.1.8</oid></td></tra<>	p> <oid>.1.3.6.1.4.1.8</oid>
Auto Bridge Security	Mar 19 17:03:11	5050	433.5.1.0.5-Joid>-data>-item>-oid	P.1.3.6.1.4.1.8433.5	1.0.5.0-void=-value_type	var-value_type>=valu
Psec Turnel			e-Audit logging has reached the file	size mreshold. 70%	NAME AND A DOM: NOT A DOM: NOT	4004007005 400
Status	Mar 19 17:03:11	2020	ES15 523NBH212 user notice kern 408720E onstands advantage in the	et: type=1305 audit(1	564055391,284.14); auid-	4294967295 985-429
Services Settings			4957295 op= add rule key=(null) li ESTE 523MBD313 user police kern	st-4 res-u al: hose 1305 audit/1	684665301 204 151 million	emphasized added and
DHCP Server	Mar 19 17:03:11	2020	d=4304067305 pag=4304067305 pa	er, type=1303 augutt	1004000081.204.10), auan	_enabled=1.0id=1.avi
SNMP Agent			ERTE 533NBP313 user notice kern	al konst 1305 audit/1	594655301 294 16V audit	nida 770 obdaŭ anida
Remote Administration	Mar 19 17:03:11	2020	4204087205 next4204087205 rest	er, gipe-raco acontri 1	1004000081.204.10), auan	"Dig-11.9 ord-0 and-
Web Server	Mar 19 17:03-12	2020	EST5 523NBB212 user info sysinit	init in based function	005	
Admin User Management	Mar 19 17:03:12	2020	EST5 523NBR212 user info sysinit	start stop sntp Star	t	
Add New Liter			EST5 523NBR212 daemon notice r	tod/7961 INIT: ntod	ntosec-1.1 0+419 2018-03	-14T12-03-57-0700: S
User Login Policy	Mar 19 17:03:12	2020	tarting	shafe and many rules .		
Remote A&A Setup	Mar 19 17:03:12	2020	EST5 523NBR212 daemon info ntp	d[796]: INIT: Comma	ind line: ntpd -g -c /tmp/ntp	conf
Two-Factor Auth	Mar 19 17:03:12	2020	EST5 523NBR212 daemon info ntp	d[797]: INIT: precisio	n = 1.910 usec (-19)	
Monitoring/Reports	Mar 19 17:03:12	2020	EST5 523NBR212 daemon info ntp	d[797]: INIT: success	stully locked into RAM	
Bridging Status	Mar 19 17:03:12	2020	EST5 523NBR212 daemon info ntp	d[797]: CONFIG: rea	dconfig: parsing file: /tmp/i	ntp.conf
Bridging Site Map	Mar 19 17:03:12	2020	EST5 523NBR212 daemon notice r	tpd[797]: LOG: swite	thing logging to file /tmp/ntp	p.log
Adjacent AP List	Mar 19 17:03:15	2020	EST5 523NBR212 user info sysinit.	setSerialHw line 496	3 RS232 mode	
DHCP Client List	Mar 19 17:03:15	2020	EST5 523NBR212 user.err kernel: (drivers/tty/serial/8250	0/8250.c _3eTI_loctl RS232	2 selected
Loos .	Mar 19 17:03:15	2020	EST5 523NBR212 user err kernel: (drivers/tty/serial/8250	08250.c _3eTI_ioctl full du	plex selected
System Log	Mar 19 17:03:18	2020	EST5 523NBR212 user info sysinit	Magic Version Num	ber: 520 5.1.0.dbg3.258	
Web Access Log	Mar 19 17:03:18	2020	ES15 523NBH212 user.en sysinit	mean=16/8		
Audit Configuration	Mar 19 17:03:18	2020	ES15 523NBR212 user.err sysinit	vanance=61793		1.0
Audit Log	Mar 19 17:03:18	2020	ES15 523NBH212 user warn kerne	t sysinit (61) used gr	eatest stack depth: 5/76 b	lytes left
System Administration	Mar 19 17:03:19	2020	ES15523NDH212 user into kemel:	priyustal deauthensi	caung itom c2.0e.oe.38:34	Ha by local choice (re
Email Notification Conf	Mar 40 47 02 40	3030	ERTE 520000040 upor info knowski	station of the order	been block at the under the AGR	stanu damala

Figure 80: Logs – System Log

2.11.2 Web Access Log

The **Logs – Web Access Log** screen (Figure 81) displays System Facility Messages with a Date-Time stamp for any actions involving web access. All configuration operation access via web GUI will be logged. For example, web access log records when you set encryption mode, change operating mode, etc., using the web browser. It establishes a running record regarding what actions were performed and by whom.

The Web Access Log (Weblog) will continue to accumulate listings until it reaches its 10KB size limit. The log rotates by overwriting the first 1/2. For security reasons, the Weblog cannot be cleared.

To remind the administrator that the log will be overwritten soon, this device will send a notification email when the Weblog size reaches 50%. The device also provides a configurable "Weblog alert point" and "Weblog alert email" address. When the Weblog grows over the Weblog alert point, an email is automatically sent to the configured Weblog alert email address. Note that an email will be sent only once at the alert point before the Weblog rotates and begins overwriting data.

NOTE: The Weblog alert email address and email server need to be properly configured. Refer to Section 2.13.1 Email Notification Configuration for more information on server configuration.

If you choose to Export the weblog, it can be opened or saved as an HTML document.

1.11. ····		3eTI AirGuan	d™ Wireless Access Poi	nt			Logout
Ultra	3eTI	Usemame:	CryptoOfficer	Host Nar	ne:	UUT2	
ELECTRONICS		Role:	3e-local	Radio Re	gion:	United States	
Rystem Configuration General Noisy Channel Cor WAN LAN Bridge Ethernet VLAN MAC Address Filte Certificate Store Radio 1 WLAN Mode: Mesi PHY Setting	ntrol ring h Point	Logs -> Web When weblog size system will send before weblog rot Weblog alert poin Weblog alert ema	Access Log e reaches maximum size, it will notification email. You can confi ates t: it: it:	rotate and the first 1/2 part v gure another alert point. No	will be ove te that the	rwritten. When weblog size rea email will be sent only once at	ches 50%, the the alert point
Auto Bridge Securi Psec Tunnel Profiles Status	neral Ry	Appy					Export
Services Settings DHCP Server SNMP Agent Remote Administra Web Server	ation	Date-Time Jan 1 00:00:12 2 Jan 1 00:00:12 2	System-Facility-H 005 User-CryptoOfficer 005 User-CryptoOfficer "Z"	ossage Operation Changed admin- Operation Changed note for	user passa admin-us	word for CryptoOfficer in Crypt ser CryptoOfficer from "Default	o Officer role Crypto Office" to
List All Users Add New User User Login Policy Remote A&A Setu Two-Factor Auth Monitoring/Reports System Status Bridging Status	ement p	Jan 1 00:01:05 2/ Jan 1 00:00:30 2/ Jan 1 00:08:38 2/ Jan 4 23:42:54 2/ Jan 4 23:42:56 2/	005 User:Admin Operation 005 User:CryptoOfficer 005 User:CryptoOfficer 006 User:CryptoOfficer 005 User:CryptoOfficer	ion:System Upload Operation: Set configuration Operation: Changed admin- Operation: Wireless oper mo Operation: Bridge1 Broadca Operation: Bridge1 Beacon Operation: Bridge1 Ressl_W	to the fac user passi ide chang anged: old st_SSID c Interval cl indow_Siz	tory default configuration word for CryptoOfficer in Crypti ed: old=AP, new=Bridge I=default, new=D00E8E9FCAD hanged: old=000, new=1 hanged: old=1000, new=20 te changed: old=5, new=20	o Officer role 19
Bridging Site Map Adjacent AP List DHCP Client List ogs System Lop Web Access Log]	Jan 4 23:42:59 2 Jan 4 23:43:11 2 Jan 4 23:44:46 2	005 User: CryptoOfficer 006 User: CryptoOfficer 006 user: CryptoOfficer 006 new=Disable 006 User: CryptoOfficer 005 User: CryptoOfficer 005 User: CryptoOfficer 005 User: CryptoOfficer 005 User: CryptoOfficer	Operation: Bridge1 AES-CC Operation: PHY 1 channel cl Operation: PHY 1 802.11 fex Operation: PHY 1 Tx Pwr M Operation: PHY 1 Tx Pwr Le Operation: PHY 1 wireless n	M Key chi hanged: o ature chan ode chang vel chang node chan	inged. Id=1 new=6 iged: old=Enable: 20 MHz Cha red: old=Off new=Fixed ied: old=1 new=8 ged: old=802.11g new=802.11	nnel

Figure 81: Logs – Web Access Log

2.12 Auditing

The unit collects audit data and provides an interface for authorized administrators to review generated audit records. It generates records for two separate classes of events: authentication/access to the system, and actions taken directly on the system. All audit records include the date/time of the event, the identity associated with the event (such as the service, computer or user), the success/failure of the event and a definition of the event (by code or explanation).

Every start and stop of the audit service are noted in the audit record. For audit events resulting from actions of identified users, the unit associates each auditable event with the identity of the user that caused the event. The unit includes or excludes auditable events from the set of audited events based on object identity, user identity, subject identity, host identity, and event type.

The Auditing screens contain auditing functions for the system. The screens and functions are detailed in the following subsections.

2.12.1 Audit Configuration

The **Auditing – Configuration** screen (Figure 82) is used to configure the auditing settings. You can enable or disable the auditing function on this screen. You can select which audit event types you wish to log. Table 16 lists event types and descriptions. The figure below describes the attributes for audit log selection, the user can also specify audit logs for a particular administrator by inputting the user ID in the text box.

NOTE: Many of these event notifications cannot be disabled for security reasons. They are displayed for reference only.

System Configuration	Auditing -> Audit Configuration						
General	The following audit events cannot be disabled.						
Notey Channel Control WAN	Event Type (Auditing on Success/Failure)	Success	Falure				
LAN	Audit Startup/Shutdown	2	2				
Ethernet VLAN	Audit Configuration Change		R				
MAC Address Filtering Certificate Store	Key Transfer Error	NA					
Radio 1	Key Zemization						
PHY Setting	Kay Canaration						
AP General AP Security	Sustem Time Change		2				
AP Wreless Clients Wireless VLAN Mapping	System time onange	-	80				
IPsec Tunnel	Enable or disable the following audit events bas	ed on the event type and outcome.					
Profiles Status	Event Type (Auditing on Success/Failure)	Success	Falure				
Services Settings	Self Test	2					
SNMP Agent	Encryption/Decryption Error	N/A	2				
Serial Port Serial Communication	Cryptographic Signature Error	N/A	2				
Remote Administration Web Server	Cryptographic Hashing Error	N/A	V				
Admin User Management	Random Bit Generation Error	NA	2				
Add New User	Certificate Validation Error	N/A	V				
Remote A&A Setup	IPsec Security Policy		2				
Two-Factor Auth Monitoring/Reports	Psec Security Association		V				
System Status Britaino Status	Management Connection		2				
Bridging Site Map	Data Connection	2	V				
DHCP Client List	802.1X Port Authentication	2	2				
System Log	Admin User Authentication	2	2				
Web Access Log	System Configuration	2	2				
Audit Configuration	Software Update	8	V				
System Administration	Wreless Client Authentication	2	2				
Email Notification Conf Radio Tx Off Control	Wireless Client Association	8	V				
System Upgrade Default Configuration	Encryption Algorithm Change	2	2				
Remote Logging	Mac Address Filter Change	2	V				
On Demand Self-test	Resource Quota Exceeded	N/A	2				
Periodic Self-test Utilities							
Help	Audit administrative actions taken by the specifi	ied admin user ID.					
	Admin User ID:	(Enter "*" for all administrative users)					
	Select the action when local audit storage is full						
	Action on Storage Full: O Stop Logging @ Ro	otate Log File					
	Remote auditing: transmit audit records to an audit server over a trusted channel.						
	Remote Auditing: O Enable Disable						
	IPsec Tunnel						
	TLS Protocol: TLS 1.2						
	Audit Server Address:						
	Audit Server Port: 514 (default port	: 514)					

Figure 82: Auditing – Configuration

Table 16: Auditing – Configuration Event Types and Description					
Event Type	Description				
Audit Startup Shutdown	Audit record generated when the audit service is started or stopped. This record cannot be disabled.				
Audit Configuration Change	Any modification to the audit log configuration (enable/disable, recorded event types, etc.) will trigger the creation of an audit record. Cannot be disabled.				
Key Transfer Error	Any error detected during the dynamic key exchange, either to the station or the authentication server. Cannot be disabled.				
Key Zeroized	The keys are zeroized including:1. Transitioning from static key to DKE (and vice versa)2. Transitioning to bypass mode Individual log messages appear from the application and driver since keys are held in both locations.Cannot be disabled.				
Key Generation	These events are logged anytime a newly generated key is introduced in AP either via web GUI or internally. Cannot be disabled.				
System Time Changed	Whenever the time is changed via the GUI or at bootup if the time is within two minutes of 11/30/1999, 0hr, 0min. Cannot be disabled.				
Self-test	Self-test Results. Default is to log success and failure.				
Encryption Decryption Error	Failure of encryption or decryption. Default is to log failure.				
Cryptographic Signature Error	Failure or cryptographic signature. Default is to log failure.				
Random Bit Generation Error	Failure of randomization process. Default is to log failure.				
Certificate Validation Error	Failure of certificate validation. Default is to log failure.				
IPsec Security Policy	Success or failure of a packet crossing the IPsec Security Boundary. Default is to log failure only.				
IPsec Security Association	IPsec Security Association success or failure. Default is to log success and failure.				
Management Connection	Managed WEB TLS connection to the device. Default is to log success and failure.				
Data Connection	No events are generated for wireless device.				
802.1X Port Authentication	Wireless client authentication. Default is to log success and failure.				
Admin User Authentication	Administrative user authentication, logout and lockout. Default is to log success and failure.				
System Configuration	System configuration changes through the WEB management interface. Default is to log success and failure.				
Software Update	System firmware updates. Default is to log success and failure.				
Wireless Client Authentication	A station's authentication request is audit logged on success or failure. Default is to log both success and failure.				
Wireless Client Association	A station an attempt to associate. Default is to log success and failure				
Encryption Algorithm Changed	The encryption algorithm is changed, including bypass mode. Default is to log success and failure.				
MAC Filter Changed	The MAC address filter is changed including adding/deleting, enable/disable, and changing filter type. Default is to log success and failure				
Resource Quota Exceeded	Wireless client denied association due to resource limitation. Default is to log client denial.				

2.12.1.1 Audit Records Limited by Admin User ID

In addition to limiting audit records by event type, the device can further limit records based on administrative user ID. To limit audit logging to only events associated with an administrative user, add the username to the 'Admin User ID' attribute.

Audit administrative actions taken by the specified admin user ID.					
Admin User ID:		(Enter "*" for all administrative users)			

Figure 83: Audit Records Limited By User ID

2.12.1.2 Audit Record Local Storage Action

Audit log records are stored locally on the device in 256K of persistent memory. Once this memory is full, new audit record will rotate out the oldest records and the old records will be lost. Alternatively, the audit service can be configured to stop logging when the local storage is full. The default behavior is to rotate the audit log.

Select the action when k	al audit storage is full.	
Action on Storage Full:	Stop Logging Rotate Log File	

Figure 84: Audit Record Local Storage Action

2.12.1.3 Remote Audit Log Configuration Details

The remote logging feature is similar to remote sys logging except it optionally provides a TLS session over which to communicate. This feature will work with syslog servers supporting TLS 1.2. When remote audit logging is enabled, all audit log records will be stored locally, and duplicate records will be sent to the remote audit log server.

Note: Only the user with 3e-Local or 3e-CryptoOfficer role may change the audit log configuration.

Remote auditing: trans	mit audit records to an audit server over a trusted channel.
Remote Auditing:	Enable O Disable
IPsec Tunnel	Profile Remote_Audit_Log V Specify IPsec IDs V
TLS Protocol:	TL5 1.2
Audit Server Address.	192.168.200.100
Audit Server IPsec ID:	AuditLog
Audit Server Port:	514 (default port: 514)
1	
Apply	

Figure 85: Remote Audit Logging with IPsec Tunnel

Remote Auditing: Enable or disable the remote logging of audit events.

IPsec Tunnel: When checked will further protect the remote audit log records by sending them within an IPsec tunnel. Check the box and select an IPsec Tunnel Profile. An IPsec Tunnel Profile must be configured before IPsec can be used to protect the audit log packets. See Section 0 in order to configure an IPsec Tunnel Profile. For Common Criteria compliance, the user must select this option to use IPsec tunnel to connect with audit log server. If the "Profile" drop-down list is empty, the user shall go to the IPsec

configuration screen to create a "Profile", in which the user will specify the peer IP address, certificates used for peer authentication and ESP encryption options. To view the IPsec tunnel status, user should refer to Section 2.5.2. In case of IPsec tunnel connection failure, the user can see the IPsec status and the device will continue to try establishing the tunnel will save the log to local storage. Once the tunnel is reestablished, the device will try to sync its local audit log with the remote server.

During the authentication phase of the IPsec protocol, the 3e-520 series (Initiator) will specify to which of the remote audit server's (Responder's) identities it wants to communicate with. Most remote audit servers will have one IPsec identity bound to the server's IP address. By default, the 3e-520 series will talk to this identity. There are instances where a remote audit server may host multiple IPsec identities. In this case, the desired identity can be specified by checking the "Specify IPsec IDs" check box and supplying an IPsec identity. If PKI certificate-based authentication is configured for the IPsec tunnel, the "IPsec IDs" field will match with the peer certificate's entire DN or A SAN:DNS name.

TLS Protocol: When the protocol is left un-checked, the audit log records will be transmitted over a TCP socket. By checking the TLS 1.2 protocol, that socket will be authenticated and encrypted using the TLS 1.2 protocol. The device will negotiate with the configured Audit Server for a secured TLS protocol.

Audit Server Address: User-entered string giving the remote server's address. String entry must be either a standard IPv4/v6 address or a valid domain name (see RFC 2181 for more details). Enclose IPv6 addresses in square brackets.

Audit Server Port: User-entered number string for the port number from which the remote logging server is accepting TLS sessions.

2.12.2 Audit Log

The **Auditing – Log** screen (Figure 86) provides a listing of all the audit records. Audit Log records all security related activities and events which are configured in the Audit Configuration setting (see Section 2.12.1).

The Audit Log (Audit-log) will continue to accumulate listings until it reaches its 256KB size limit. The log rotates by overwriting the first 1/3. For security reasons, the Audit-log cannot be cleared.

To remind the administrator that the log will be overwritten soon, this device will send a notification email when the Audit-log size reaches 50%. The device also provides a configurable "Audit-log alert point" and "Audit-log alert email" address. When the Audit-log grows over the Audit-log alert point, an email is automatically sent to the configured Audit-log alert email address. Note that an email will be sent only once at the alert point before the Audit-log rotates and begins overwriting data.

NOTE: The Audit-log alert email address and email server need to be properly configured. Refer to Section 2.13.1, Email Notification Configuration, for more information on server configuration.

If you choose to Export the Audit-log, it can be opened or saved as an HTML document.

1.11		3eTI AirGuard™ Wireless Access Point Logor							
Ultra	3eTI	Username	c CryptoOfficer		Host Name:	UUT2			
ELECTRONICS		Role:	3e-local		Radio Region:	United States			
	· ·								
Sestern Configuratio		Auditin	g -> Log						
General									
Noisy Channel Cor	lorth	When aux	fit-log size reaches maximu	m size, it will rotate and the fir	st 1/3 part will be o	verwritten. When audit-log size reaches 5	50%, the		
WAN		system wi	Il send notification email. Yo	ou can configure another alert	point. Note that the	e email will be sent only once at the alert	point		
Bridge		before au	ait-log rotates						
Ethernet VLAN									
MAC Address Filte	ring	Audit-log	alert point:	90% 🛩					
Certificate Store		Audit-log	alert email:			_			
WLAN Mode: Mesi	h Point			1					
PHY Setting									
Wreless Mesh Ger	neral	Apply							
Auto Bridge Securi Paec Turnel	tγ								
Profiles							Export		
Status							-		
Services Settings		No	Date-Time	System Facility Messa	ge				
SNMP Agent		3165	Mar 26 20:01:23 2020	1 EVT_STA_ASSOC,	802.11 mgmt assoc	ciation OK, , 00:0e:8e:6f.ca:d3,			
Remote Administra	ition	3166	Mar 26 20:01:23 2020	1 EVT_STA_ASSOC, ,	MLME-ASSOCIATE	E, , 00:0e:8e:6f.ca/d3,			
Web Server		3107	Mar 26 20 01 23 2020	1 EVT_KET_ZERUIZA	TION, MEME-DED	ETEREYS, UNDEBEDICEDS,			
Admin User Manage	ment	3169	Mar 26 20 01 23 2020 Mar 26 20 01 23 2020	1 EVT_KEY_ZEROIZAT	TION, PTK is zero	ized, 00 0e 8e 6t card3			
Add New User		3170	Mar 26 20:01:23 2020	1 EVT KEY ZEROIZAT	TION, PTK is zero	ized , 00 0e 8e 6f ca d3			
User Login Policy		3171	Mar 26 20:01:24 2020	1 EVT_KEY_GENERAT	ION, Successfully	derived PTK from PMK, . 00:0e:8e:6f.ca	aid3,		
Remote A&A Setup	p	3172	Mar 26 20:01:24 2020	1 EVT_SELF_TEST, , m	nac80211 AES-CCI	M self test passed, , ,			
Two-Pactor Auth		3173	Mar 26 20:01:32 2020	1 EVT_KEY_ZEROIZA	TION, , PTK is zero	ized, , 00:0e:8e:6f.ca.d3,			
System Status		3174	Mar 26 20:01:32 2020	1 EVT_KEY_ZEROIZAT	TION, PTK is zero	ized, , 00:0e:8e:6f.ca.d3,			
Bridging Status		31/5	Mar 26 20:01:32 2020	1 EVT KEY ZEDO(ZA)	DOM MIME DELL	ETEKEVS 00 0a be Browd3	1.03,		
Bridging Site Map		3177	Mar 26 20:01:32 2020	1 EVT_KEY_ZEROIZAT	TION PTK is zero	ized 00.0e/8e/8trad3			
DHCP Client List		3178	Mar 26 20:01:32 2020	1 EVT KEY ZEROIZAT	TION. mac80211 k	key zeroized			
Ogs		2470	Mar 26 20 04 22 2020	1 EVT_KEY_ZEROIZAT	TION, PMK/PTK a	re zeroized when removing STA,			
System Log		31/9	Mar 26 20.01.32 2020	00.0e.8e.6f.ca.d3,					
Web Access Log		3180	Mar 26 20:01:32 2020	1 EVT_STA_AUTH, M	LME-AUTHENTICA	ATE, 00:0e:8e:6f.ca.d3,			
Auditing Audit Conferenties		3181	Mar 26 20 01 32 2020	1 EVT_KEY_ZEROIZAT	TION, MUME-DEL	ETEREYS, 00:0e:8e:6f.ca.d3,			
Audit Log		3182	Mar 26 20:01:32 2020	1 EVT_STA_AUTH, 80	12.11 mgmt authen/	tication OK, , 00 De Se 6f ca d3,			
			The second second second		The second second second				

Figure 86: Auditing – Log

2.13 System Administration

The **System Administration** screens contain administrative functions. The screens and functions are detailed in the following section.

2.13.1 Email Notification Configuration

All system notification emails need to be set up using the **System Administration – Email Notification Configuration** screen (Figure 87). Your email server must support SMTP protocol.

If you email server does not require authentication to send email, then leave the username/password fields blank. If your email server does not support TLS 1.2 then disable TLS 1.2. You may also test your email setup using the test feature on this screen.

NOTE: Check your connection to the mail server. Emails sent from the 3e-520 series product family may be queued for a short period if the connection fails temporarily, but it will give up if the connection continues to fail.

1.00	3eTI AirGuard	™ Wireless	Access Point			Logent
Olfra 3eT	Username:	CryptoOfficer		Host Name:	default	
ELECTROPICS	Role:	3e-local		Radio Region:	United States	
System Configuration	System Admi	nistration ->	Email Notification Configu	uration		
General Noisy Channel Control WAN	All system email r	notification will u	se the following setup to send ema	ail,		
LAN Bridge	E-mail Server:					
Ethernet VLAN	Port:		587			
Certificate Store	TLS 1.2:		Enabled V			
Radio 1 WLAN Mode: AP	Usemame:					
PHY Setting AP General	Password:					
AP Wreless Clients Wreless VLAN Mapping	Apply					
IPsec Tunnel Profiles	You may use the	following to test	your email setup now.			
Status Services Settings	Put your destinati	on email addres	s below and hit 'Test'			
DHCP Server SNMP Agent	Test					
Serial Port Serial Communication						
Remote Administration Web Server						

Figure 87: System Administration – Email Notification Configuration

To test the email function, the email address needs to be filled in. Afterwards click on the **Test** button and a pop-up window such as the one shown in Figure 88 will indicate the result of email test.

https://192.168.15.1/cgi-bin/sgateway?PG=87&EMAIL_ADDR=3eTIte	st@gmail.com 💷 💷 🔀
https://192.168.15.1/cgi-bin/sgateway?PG=87&EMAIL_ADDR=3eTItes	st@gmail.com 😵 Certificate Error
Email has been successfully queued for delivery. Please check y	your mail box.
	-

Figure 88: Email Test Result

2.13.2 Radio TX Off Control

The radios can be programmed to turn off and turn on during a specified time in the future as shown in Figure 89. This is typically executed via a management system like WiFiProtect Manager.

a diaman		3eTi AirGuard™ Wireless Access Point Logout						
Ultra	3eTI	Username:	CryptoOfficer	Host Name:	default			
electromacs		Role:	3e-local	Radio Region:	United States			
System Configuration General Noisy Channel Control WAN LAN Bridge Ethemet VLAN MAC Address Filtering Certificate Store Radio 1 WLAN Mode: AP DHY Sestion		System Administration -> Radio Tx Off Control Radio Tx Off Control: O Enable ® Disable Current System Time: 03/13/2020 17:19:23 Radio 1 TxPowerMode: Auto Radio Tx Off Control Status: Idle Start Time: Duration:						
AP General AP Security AP Wireless Client Wireless VLAN Ma	s pping	Start Time and D All radio transmiss	uration Setup: ion of this device will be turned off at the s	pecified date and time an	d for the set duration.			
IPsec Tunnel Profiles Status		Start Date: Start Time:	/ / (month	/day/year) ute:second)				
DHCP Server SNMP Agent Serial Port Serial Communicat Remote Administra Web Server	tion tion	Duration: Submit	(minimum 10 second	5)				

Figure 89: System Administration – Radio TX Off Control

2.13.3 System Upgrade

The **System Administration – System Upgrade** screen (Figure 90) gives you the ability to upload updates to the device's firmware as they become available. When a new upgrade file becomes available, you can perform a firmware upgrade from the Firmware Upgrade window. Normally, the user will first get email notification from 3eTI's product support team. If the user desires to download the new firmware release, a secure download link together with temporary username/password will be provided to the user. The user can download the firmware into local storage, then follows instruction in the section below to tell the device's web UI the local of the firmware.

There is also a configuration file transfer option which allows the system configuration file from one AP to be transferred to another AP, in order to minimize the administration of the APs. Only configuration parameters that can be shared between APs are downloaded in the configuration file. WAN IP address and hostname are not transferred in the configuration file.

Only the user with 3e-Local or 3e-CryptoOfficer role can access this function.

2.13.3.1 Firmware Upgrade

The CryptoOfficer can update the device's firmware. The device uses the public key to verify the digital signature. Upon successful verification, the device will load the new update upon reboot. The update will be rejected if the verification fails.

On the **System Administration – System Upgrade** screen (Figure 90), the **Firmware Upgrade** tab is the default view.

Click on **Browse** and select the firmware file to be uploaded. The device will only accept firmware files that have been digitally signed by 3eTI Click on the **Upload Firmware** button.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com



Concerning of the second		JeTI AirGuard [®] Wireless Access Point						
Ultrol BUCTRONICS 30T	3eTI	literrane:	CryptoOfficer	and the second s	Alost Nation	detaid		
		Wole:	Selocal		Radio Region:	United States		
System Configuration Cleveral Noisy Disaved Control 9040 LAN		System Ac	Iministration -> System L Oppoide Configuration and select a Ne	lpgrade (apert/limpert				
Ethornal VLAN MAC Astrony Film Cartificane State	1			Browse				
Radio 3 WLAN Hinder AP DVV Setting AP Genata		Lipkoad Ferry	1010					



2.13.3.2 Configuration Export/Import

On the **System Administration – System Upgrade** screen (Figure 91) click on the **Configuration Export/Import** tab to upload and download configuration files to APs connected to the network.

To upload a configuration file, select the file using the browse button and enter the passphrase for that file. The passphrase protects the file from unauthorized users. It prevents unauthorized users from applying the system configuration file to an unauthorized AP to gain access to the network. Before downloading the system configuration file to a local computer, the user must enter a passphrase to protect the file. Before the system configuration file can be uploaded onto another AP, the passphrase must be entered on the remote AP.

1.11	3eTI AirGua	ırd™ Wire	less Access Point				Logost
Ultra 3eTI	Username:	CryptoO	ficer		Host Name:	UUT2	
DUCTIONICS	Role	3e-local			Radio Region:	United States	
	System Ad	ministrativ	on -> System Lingrad	ie .			
System Configuration	ayotem Au		ni - aystein apgrad				
General Noise Channel Control	Firmware	Upgrade	Configuration Export	/Import			
WAN	Import Con	Bauration					
LAN	Choose a fi	is to import					
Ethernet VLAN	010000 010	e w mpon	0	COLUMN D			
MAC Address Filtering				Owner			
Certificate Store	Import						
WLAN Mode: Mesh Point	Export Con	figuration					
PHY Setting	Export						
Auto Bridge Security							
Psec Tunnel							
Profiles							
Services Settings							
DHCP Server							
SNMP Agent Remote Administration							
Web Server							
Admin User Management							
Add New User							
User Login Policy							
Remote A&A Setup Two-Easter Auth							
Monitoring/Reports							
System Status							
Bridging Status Bridging Site Map							
Adjacent AP List							
DHCP Client List							
System Log							
Web Access Log							
Auditing Audit Configuration							
Audit Log							
System Administration							
Radio Tx Off Control							
System Upgrade							
Remote Logaina							
The same copping							

Figure 91: System Administration – System Upgrade – Configuration Export/Import

1) Notes and Tips:

- a) When Exporting configuration files, keys are NOT downloaded.
- b) When Importing configuration files to a device, if the device currently is configured to the same security options as those in the uploaded file, the keys are reused. Otherwise, the keys are zeroized and marked "key not set" from the web GUI.

e.g.: The current device has the 802.11i-PMK option on AP security. The configured files to be uploaded will use 802.11i-PMK for AP security. The existing 256bit PMK is reused. Otherwise, the AP security is marked "key not set".

c) In a VLAN scenario, VLAN ID (NOT the SSID) is the index to find a matching security option.

e.g.: Current device has 3 VLAN configured as follows:

- i) VLAN ID = 1, SSID=area-1, security=802.11i pmk,
- ii) VLAN ID = 2, SSID=area-2, security=802.11i-dot1x,
- iii) VLAN ID = 3, SSID=area-3, security=static AES.

The configuration file to be uploaded has 4 VLAN configured as follows:

- iv) VLAN ID = 1, SSID=test-1, security=802.11i pmk,
- v) VLAN ID = 2, SSID=area-1, security=802.11i-dot1x,
- vi) VLAN ID = 3, SSID=area-2, security=802.11i pmk,
- vii) VLAN ID = 4, SSID=area-3, security=static AES.

The device will have 4 VLANs configured as follows:

- viii) VLAN ID = 1, SSID=test-1, security=802.11i pmk (key set),
- ix) VLAN ID = 2, SSID=area-1, security=802.11i-dot1x (key set),
- x) VLAN ID = 3, SSID=area-2, security=802.11i pmk (key not set),
- xi) VLAN ID = 4, SSID=area-3, security=static AES (key not set).

2.13.4 Default Configuration

The **System Administration – Default Configuration** screen (Figure 92) is used to reset the AP to its factory settings.

The **Restore** button is a fallback troubleshooting function that should only be used to reset to last saved general configuration.

The **Save** button can save the current general configuration to be restored in the future.

The **Reset** button can be used to reset and restore the configuration to the factory default settings.

Only the user with 3e-Local or 3e-CryptoOfficer role has access to the Default Configuration page.



Figure 92: System Administration — Factory Default

2.13.5 Remote Logging

The **System Administration – Remote Logging** screen (Figure 93) allows you to forward the *syslog* data from each machine to a central remote logging server. In the unit, this function uses the *syslogd* daemon. If you enable Remote Logging, input a System Log Server IP Address and System Log Server Port. Click **Apply** to accept these values.

	3eTI AirGuard™ Wireless Access Point Los							
Ultra 3eTI	Username:	CryptoOfficer		Host Name: UUT2				
ELECTRONICS	Role:	3e-local		Radio Region:	United States			
	Evetem Adm	inistration - Permote L	oning					
System Configuration	aystem Aum	Inistration - Remote Lo	999 mg					
Noisy Channel Control	Remote Logging		O Enable Dis	able				
WAN	System Log Sen	ver Hostname: (Enclose IPv6						
Bridge	Addresses with [D						
Ethernet VLAN MAC Address Ethering	System Log Sen	ver Port:	514 (defaul	It port is 514)				
Certificate Store								
Radio 1 WLAN Mode: Meab Point	Apply							
PHY Setting								
Wireless Mesh General Auto Bridge Security								
Psec Tunnel								
Profiles Status								
Services Settings								
SNMP Agent								
Remote Administration								
Admin User Management								
List All Users Add New User								
User Login Policy								
Remote A&A Setup Two-Factor Auth								
Aonitoring/Reports								
Bridging Status								
Bridging Site Map Adjacent AP List								
DHCP Client List								
Sustem Lon								
Web Access Log								
Auditing Audit Configuration								
Audit Log								
System Administration Email Notification Conf								
Radio Tx Off Control								
System Upgrade Default Configuration								
Remote Logging								
On Demand Self-text								

Figure 93: System Administration — Remote Logging

2.13.6 Reboot

The **System Administration – Reboot** screen (Figure 94) allows you to reboot the unit without changing any preset functionality.

a alta-	3eTI AirGuard™ Wireless Access Point Lorent							
Ultra 3eTI	Username:	CryptoOfficer		Host Name:	UUT2			
ELECTRONICS.	Role:	3e-local		Radio Region:	United States			
	System Adm	inistration -> Reboot						
System Configuration General Noisy Channel Control WAN	Click 'Reboot' be	atton to reboot device.						
LAN Bridge Ethernet VLAN MAC Address Filtering Cartificate Store	Reboot							
Radio 1 WLAN Mode: Mesh Polet PHY Setting Wireless Mesh General Auto Bridge Security								
Psec Tunnel Profiles Status								
Services Settings DHCP Server SNMP Agent Remote Administration Web Server								
Admin User Management List All Users Add New User User Login Policy Remote A&A Setup								
MonitoringReports System Status Bridging Status Bridging Status DHCP Client List DHCP Client List								
Logs System Log Web Access Log								
Auditing Audit Configuration Audit Log								
System Administration Email Notification Conf Radio Tx Off Control								
Default Configuration Remote Logging Reboot								
COLUMN TWO IS NOT THE								

Figure 94: System Administration — Reboot

2.13.7 Self-Test

Self-tests are run to verify the correctness of cryptographic related functions. The 3e-520 series performs the following types of tests:

- Know Answer Tests for all supported encryption algorithms,
- Know Answer Tests for all supported hashing algorithms,
- Know Answer Tests for all supported HMAC algorithms,
- DRBG Tests,
- Sign/Verify Tests,
- Key Derivation Tests,
- Key Distribution Tests,
- Firmware and Bootloader Integrity Tests.
Test results are written to the system log. The platform should not pass secure data while self-tests are executing therefore network interfaces are disabled during self-tests. The platform is halted if any self-test fails. There are many possible factors can lead to a self-test error. For example, faulty hardware components in the noise/entropy generator could lead to DRBG and key generation failures. User can attempt the reboot the device in case of single self-test failure. If the device shows consistent self-test failures as indicated by the LED status, the user shall return the device to 3eTI for service.

These tests are run during power up, on demand or periodically. All of the above tests are executed automatically when the platform is powered up. Links to initiate on-demand or periodic self-tests are available on the platform's web page under "System Administration" if the user is logged in as a Crypto Officer.

2.13.7.1 On Demand Self-test

Selecting the **On Demand Self-test** link (Figure 95) and clicking on **Start Test** executes each self-test except the firmware and bootloader integrity checks. A web page will be displayed indicating if the tests passed.

	System Administration -> On Demand Selftest
System Configuration General	Click 'Start Test' button to start the self test.
Noisy Channel Control WAN	
LAN	Start Test
Bridge	
MAC Address Filtering	
Certificate Store	
Gedio 1 WLAN Mode: Mesh Point	
PHY Setting	
Wireless Mesh General	
Page Tunnel	
Profiles	
Status	
DHCP Server	
SNMP Agent	
Remote Administration	
Admin User Management	
List All Users	
Add New User	
Remote A&A Setup	
Two-Factor Auth	
System Status	
Bridging Status	
Bridging Site Map	
DHCP Client List	
ogs	
System Log Web Access Log	
Auditing	
Audit Configuration	
Audit Log System Administration	
Email Notification Conf	
Radio Tx Off Control	
Default Configuration	
Remote Logging	
On Demand Self-test	
Fenade Let test	

Figure 95: System Administration — On Demand Self-test

2.13.7.2 Periodic Self-test

Selecting the **Periodic Self-test** link (Figure 96) allows the user to enable/disable periodic tests. One test iteration executes each self-test except the firmware and bootloader integrity checks. The **Periodic Test Interval** is the time between test iterations

	System Administration -> Pariodic Solftest		
System Configuration	System Administration -> Periodic Sentest		
General Noisy Channel Control WAN	○ Enable		
LAN Bridge	Start Date (MM/DD/YYYY):	0 / 0 / 0	
Ethernet VLAN MAC Address Filtering	Start Time (Hour/Minute):	0 / 0	
Certificate Store	Number of Iterations (<0:forever):	1	
Radio 1	Periodic Test Interval (hours):	0 Hours 🗸	
PHY Setting Wireless Mesh General Auto Bridge Security	Periodic Test Interval (minutes):	1 Minutes	
IPsec Tunnel Profiles Status	Current System Date & Time:	Date: 03/27/2020 Time:17:57	
Status Bervices Settings DHCP Server SNMP Agent Remote Administration Web Server	Submit		
Admin User Management List All Users Add New User User Login Policy Remote A&A Setup Two-Factor Auth			
Monitoring/Reports System Status Bridging Status Bridging Status Bridging Status DHCP Client List			
Logs System Log Web Access Log			
Auditing Audit Configuration Audit Lon			
System Administration Email Notification Conf Radio Tx Off Control			
System Upgrade Default Configuration Remote Logging			
Reboot On Demand Self-test Periodic Self-test			
Utilities			

Figure 96: System Administration — Periodic Self-Test

Since network interfaces are disabled during self-tests, a message highlighted in red is displayed on the common banner informing the user that periodic self-tests are enabled.

NOTE: If the Number of Iterations is set to be negative, the self-test will run forever. Test processes will be logged in **Auditing – Log.** When **Periodic Test Interval** is set to a short period, such as 1 minute, log events will fill up the audit log very quickly and generate an email alert as configured.

2.13.8 Utilities

The **System Administration – Utilities** screen (Figure 97) gives you ready access to two useful utilities: Ping and Traceroute. Simply enter the IP Address or hostname you wish to ping or traceroute and click either the **Ping** or **Traceroute** button, as appropriate.

System Configuration	System Administration - Utilities	
General Noisy Channel Control WAN	IP address or hostname:	Ping
LAN Bridge Ethermet VLAN MAC Address Filtering Certificate Store	IP address or hostname:	traceroute
Redio 1 WLAN Mode: Mesh Point PHY Setting Wireless Mesh General Auto Bridge Security		
IPsec Tunnel Profiles Status		
Services Settings DHCP Server SNMP Agent Remote Administration Web Server		
Admin User Management List All Users Add New User User Login Policy Remote A&A Setup Two.Factor Auth		
Monitoring/Reports System Status Bridging Status Bridging Status Bridging Ste Map Adjacent AP List DHCP Client List		
Logs System Log Web Access Log		
Auditing Audit Configuration Audit Log		
System Administration Email Notification Conf Radio Tx Off Control System Upgrade Default Configuration Remote Logging Reboot On Demand Self-test Periodic Self-test		
Utilities Help		

Figure 97: System Administration — Utilities

2.13.9 Help

The **System Administration – Help** screen (Figure 98) displays detailed hardware and software version information when you click on **Help**.

	Hardware and Software Version Information		
System Configuration General			
Noisy Channel Control WAN LAN Bridge Ethernet VLAN MAC Address Filtering Certificate Store	Software version: 5.1.0.00 Build 260 Software version on flash: 5.1.0.00 Build 260 Uboot-1 version: 1.06 Uboot-2 version: 1.06 Rescue version on flash: 1.0.0.R3 Build 8 Model Name: 3e-523N		
Radio 1 WLAN Mode: Mesh Point PHY Setting Wireless Mesh General Auto Bridge Security	Product serial number: F23N18W0200456 Product part number: 32000966-005_I dev: None Processor serial number: 7121U0569 Processor part number: 32000913-004_E dev: None		
Psec Tunnel Profiles Status	EEPROM format: 4		
Services Settings DHCP Server SNMP Agent Remote Administration Web Server			
Admin User Management List All Users Add New User User Login Policy Remote A&A Setup			
Two-Factor Auth			
Monitoring/Reports System Status Bridging Status Bridging Site Map Adjacent AP List DHCP Client List			
Logs System Log Web Access Log			
Auditing Audit Configuration Audit Log			
System Administration Email Notification Conf Radio Tx Off Control System Upgrade Default Configuration			
Remote Logging Reboot On Demand Self-test Periodic Self-test			
Help	s - 3e Technologies International. All rights reserved.		

Figure 98: System Administration — Help: Hardware and Software Version Information

3. 3e-523N Hardware Installation

3.1 Preparation for Use

This section describes installation of the 3e–523N unit; see the following sections for 3e-525N series information. The 3e–523N requires physical mounting and installation on the site, following a prescribed placement design to ensure optimum operation.

NOTE: The 3e–523N is designed for indoor use but can be deployed outdoors within a suitable enclosure. Contact 3eTI for details.

The 3e–523N package includes the following items:

- The 3e–523N unit,
- Power cable adapter,
- Quick Start Guide.

The available 3e-523N accessories are shown in Table 17.

Table 17: 3e-523N Accessories		
Model Name	Description	Comments
3e-A523WA1	3e-523N Wiring Accessory	Serial I/O Interface Board with 20 cm cable and DIN Rail mounting hardware
3e-ANT-2303	3e-523N Antenna Set	Three RP-SMA Antennas
3e- A523PP1	3e-523N Power Pigtail	Cable with custom connector for providing existing 5-12 VDC power to 3e-523N
3e-A523DPK	3e-523N DIN-Rail Mounted Power Kit	DIN Rail mounted 120/240 VAC -> 9 VDC power supply and cable w/custom connector for 3e-523N
3e-A523DMNT	3e-523N Desktop Power Kit	Desktop 120/240 VAC -> 12 VDC power supply and cable w/custom connector for 3e-523N
3e-CBL-CAT6-2	2 Meter CAT6 Interconnection Cable	GigaBit capable Ethernet cable - 2 Meter Length
3e-CBL-CAT6-5	5 Meter CAT6 Interconnection Cable	GigaBit capable Ethernet cable - 5 Meter Length

3.1.1 Installation Instructions

It is intended that the user not open the unit. Any maintenance required is limited to the external enclosure surface, cable connections, and to the management software only.



WARNING: To comply with FCC RF exposure compliance requirements, the antennas used with the 3e-523N product family must be installed with a minimum separation distance of 20 cm, 35 cm for any approved directional antenna, from all persons and must not be co-located or operated in conjunction with any other antenna or transmitter. Installation should be accomplished using the authorized cables and/or connectors provided with the device or available from the manufacturer/distributor for use with this device. Changes or modifications not expressly approved by the manufacturer or party responsible for this FCC compliance could void the user's authority to operate the equipment.

3.2 Device Installation

3.2.1 DIN Rail Mounting

The 3e-523N includes a standard locking DIN Rail mount as shown in Figure 99.



Figure 99: 3e-523N DIN Rail Mounting

3.2.2 Rear Mounting

The DIN Rail mount can be remoted to allow the 3e-523N directly mounts to a panel or other flat surface, using metal screws to attach the unit at four mounting holes as shown in Figure 100 and Figure 102. Note that these screws cannot penetrate more than 0.25 inches into the enclosure. This provides easy access to the front panel connectors and grounding screw.



Figure 100: 3e-523N Rear Mounting

3.2.3 Base Mounting

The 3e-523N can be mounted directly to a panel or other flat surface on its side, using metal screws to attach the unit at four mounting holes as shown in Figure 101 and Figure 102. However, this may obscure the label that provides important product information as shown below. Note that these screws cannot penetrate more than 0.25 inches into the enclosure.





3.2.4 3e-523 Mounting Dimensions



Figure 102: 3e-523N Mounting Dimensions

3.3 Accessory Kit Installation

Table 17: 3e-523N Accessories describes the Ethernet, power, ground, and serial I/O connections associated with these kits. Detailed installation instructions are provided with each accessory kit. Please contact 3eTI Product Support for additional information.

4. 3e-525N Hardware Installation

4.1 Installation Preparation

The 3e-525N product family requires physical mounting and installation on the site, following a prescribed placement design to ensure optimum operation and roaming.

FCC Regulations require that the 3e-525N product family be professionally installed by an installer certified by the National Association of Radio and Telecommunications Engineers or equivalent institution.

3e-525N product operates with PoE which requires the use of a separate power injector that "injects" DC current into the Ethernet cable. These PoE devices should comply with the 802.3at or 802.3af specifications. Standard versions of the 3e-525N product family draw a maximum 14 Watts of power.

The 3e-525N product family package includes the following items:

- The 3e-525N unit,
- 3e-CPLR-1 weatherproof coupler (Qty 1),
- Quick Start Guide.

The available 3e-525N accessories are shown in Table 18.

Table 18: 3e-525N Accessories			
Part Number	Description	Comments	
3e-ANT-N	Single N-type Antenna		
3e-ANTMNT3N	3e-525N Triple Cable Mount	Mounting Bracket with 3 type-N Adapters without antenna cables	
3e-ANT-DOME-1	Indoor (Ceiling-Mount) Dome Antenna		
3e-ANT-DOME-2	Outdoor Dome Antenna		
3e-CPLR-1	Bayonet Coupler	Coupler to waterproof connectors	
3e-CONN-A1	AUX Cable Connector	Auxiliary connector for the 3e-525N	
3e-POE-GB1	Gigabit POE Injector		
3e-CBL-CAT6-1	1 Meter CAT6 Interconnection Cable	GigaBit-Capable Ethernet Cable - 1 Meter Length – RJ45 Connectors on Both Ends	
3e-CBL-CAT6-2	2 Meter CAT6 Interconnection Cable	GigaBit-Capable Ethernet Cable - 2 Meter Length – RJ45 Connectors on Both Ends	
3e-CBL-CAT6-3	3 Meter CAT6 Interconnection Cable	GigaBit-Capable Ethernet Cable - 3 Meter Length – RJ45 Connectors on Both Ends	
3e-OPK-N	Outdoor Protection Kit	Kit for the 3e-525N Family	
3e-PMK-005	Pole Mount Kit		

The 3e–525N product family can be mounted outdoors on a high post to achieve the best bridge result. If mounted outdoors, the outdoor protection kit must be used to prevent lightning damage.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com





WARNING: To comply with FCC RF exposure compliance requirements, the antennas used with the 3e-525N family products must be installed with a minimum separation distance of 20 cm, 35 cm for any approved directional antenna, from all persons and must not be colocated or operated in conjunction with any other antenna or transmitter. Installation should be accomplished using the authorized cables and / or connectors provided with the device or available from the manufacturer/ distributor for use with this device. Changes or modifications not expressly approved by the manufacturer or party responsible for this FCC compliance could void the user's authority to operate the equipment.

4.2 Installation Instructions

The 3e–525N product family is intended to be installed as part of a complete wireless design solution.

This section deals only with the 3e–525N product family and its accessories. The purpose of this section is the description of the device and its identifiable parts so that the user is sufficiently familiar to interact with the physical unit. Preliminary setup information provided below is intended for information and instruction of the wireless LAN system administration personnel.

It is intended that the user not open the unit. Any maintenance required is limited to the external enclosure surface, cable connections and to the management software (as described in previous sections) only. A failed unit should be returned to the manufacturer for maintenance.

4.2.1 3e-525N Ethernet Cable Assembly

The 3e-CPLR-1 is a weather-proof coupling accessory for connecting the standard RJ-45 cables to the I/O plate of the 3e-525N family products. Assembly requires no tools and this accessory can be attached to any standard Ethernet (RJ-45) cable assembly.

First, unpack parts from the assembly kit. Feed the Ethernet cable through the nut, sealing ring, and threaded sleeve. Place the two pieces of the locking ring over the RJ-45 connector, making sure these two pieces fully mate (you will hear an audible click). Pull the RJ-45 up to the edge of the threaded sleeve. Ensure the RJ-45 connector is oriented for Code A installation shows below. Tighten the housing and nut onto the threaded sleeve. (Figure 103).



Figure 103: 3e-525N Ethernet Cable Assembly

4.2.2 Pole Mounting

To mount the unit outdoors, you should choose a suitable post to mount the unit high in the air. You can purchase a pole mounting kit from 3eTI. Use the U-ring, screws and nuts to attach the mounting plate to the post. Next attach the unit to the mounting plate with screws.



Figure 104: 3e-525N Pole Mount Installation

NOTE: Pole Mount Kit is designed for a Pole of Ø 2.5".

5. 3e-523E-900 Hardware Installation

5.1 Installation Preparation

The 3e-523E-900 radio-communications product requires physical mounting and installation on site, following a prescribed placement design to ensure optimum operation and roaming. By design, the 3e-523E-900 will only transmit and receive the intended direct-sequence spread spectrum modulated 900MHz through the use of automatic keying and code deciphering. Reference FCC ID QVT-523N-900, which is unique to this specific model.

FCC Regulations require that the 3e-523E-900 product be professionally installed by an installer certified by the National Association of Radio and Telecommunications Engineers or equivalent institution. The product uses an industry standard N-type RF output port on the exterior of the product. Installation is rated for commercial and military environments only (non-residential).

5.1.1 Specifications

ELECTRICAL SPECIFICATIONS:

Operating Voltage:	110/220VAC, 50/60Hz
Power Requirements:	15 Watts, max
External Interfaces:	Ethernet, 10/100/1000, standard RJ-45 interface
	N-type female jack antenna connector
Radio Transmission:	902-928 MHz band

ENVIRONMENTAL SPECIFICATIONS:

Operating Temperature:	-40 to +50°C External Ambient (-40 to +122 °F)
Storage Temperature:	-40 to +70°C (-40 to +158 °F)
Relative Humidity:	90%, non-condensing

MECHANICAL / MOUNTING SPECIFICATIONS:

Enclosure:	Polycarbonate, polyurethane gasket, stainless steel hardware
Weight :	8.8 lbs
Features:	Pad-lockable, outdoor deployment (weatherproof)
Dimensions:	12.08" W x 13.57" H x 6.95" D



5.1.2 Mounting Pattern



5.1.3 Installation Instructions

Install cable gland or conduit hub, as

data cables and/or weather seal.

for

appropriate



! WARNING !

Do not attempt to install any outdoor equipment during hazardous conditions such as a thunderstorm, where lightning could strike the equipment or installer. Failure to follow this warning could result in injury or death.

Mounting feet (and screws) are supplied with the unit, attached inside of the cover. Installer must attach these to the enclosure prior to mounting.

Drill holes per mounting pattern drawing and fasten with ¹/₄" screws or bolts (provided by the installer), as appropriate for the installation.

AC power cabling and cable gland (or conduit and hub) are to be provided by the installer as appropriate. Two cutouts provided, one for signal wire, one for power wiring, each at 1.109" diameter. Both holes must have appropriate cable glands or conduit hubs added for weatherproof installation.



Install AC wiring per label. Ground to an available location on the busbar, Line (black) to "L", Neutral (white) to "N" terminals.

Secure wires with gland/hub.

Install power cable gland or conduit hub, as appropriate.

Figure 105: 3e-523E-900 Installation

External grounding wire is not provided with the unit. Protection for the user and unit require a safety ground wire attached to the threaded stud at the bottom of the unit to a secure earth bonded surface. The length of this grounding wire should be kept to a minimum, Ultra-3eTI recommends less than 3 feet.

NOTE: The installer is required to ensure that the connection to a proper earth ground is made by properly certified and authorized personnel and must conform to all applicable codes and regulations. The materials required to connect to a proper ground are defined by local conditions and must be procured locally to ensure the correct safety environment is achieved. The cable used to connect to a proper ground must be AWG 10 or heavier. This cable should be kept as short as possible.

5.1.4 RF Connections

RF lightning arrester, and antenna are not provided with the unit.

- The installer should provide a suitable lightning arrestor that can attach directly to the Nfemale RF port at the bottom of the unit. Ground the lightning arrestor to the same earth bonded ground attachment as used for the unit.
- The installer should provide a suitable antenna, maximum +6dBi gain omni-directional dipole or directional Yagi style, to meet the RF coverage needs per application.
- A 25-foot length of LMR-240 cable is provided with the unit to connect the user-provided antenna to the unit. Ensure this cable is routed and tied down to avoid undue mechanical stress on the lightning arrestor attached to the unit.

5.1.5 **RF Safety Information**



FCC: To comply with FCC RF exposure compliance requirements, the antennas used with the 3e-523E-900 product must be installed with a minimum separation distance of 20 cm from all persons and must not be co-located or operated in conjunction with any other antenna or transmitter. Installation should be accomplished using the authorized cables and / or connectors provided with the device or available from the manufacturer / distributor for use with this device. Changes or modifications not expressly approved by the manufacturer or party responsible for this FCC compliance could void the user's authority to operate the equipment.

NOTE: This equipment has been tested and found to comply with part 15 of the FCC Rules. These limits are designed to provide reasonable protection against harmful interference when the equipment is operated in a commercial environment. This equipment generates, uses, and can radiate radio frequency energy and, if not installed and used in accordance to the instruction manual, may cause harmful interference to radio communications. Operation of this equipment in a residential area is likely to cause harmful interference in which case the user will be required to correct the interference at their own expense.

Appendix A. Common Criteria

A.1 Overview

Common Criteria for Information Technology Security Evaluation is an international standard for computer security certification. The 520 series claims compliance with CPP_ND_V20 and PP_WLAN_AS_EP_V1.0.

This appendix provides details of how to administer the Device to be compliant with the requirements specified in the protection profiles. In this User Guide, CryptoOfficer is equivalent to the Security Administrator in the Security Target document while Administrator is equivalent to the Non-Security Administrator role in the Security Target. The operational environment has to satisfy the conditions listed in the table below.

Operational Environment Objective from Security Target		User Guidance Reference	Applicable User Role(s)
OE.NO_GENERAL_PURPOSE	There are no general-purpose computing capabilities (e.g., compilers or user applications) available on the TOE, other than those services necessary for the operation, administration and support of the TOE.	N/A	None
OE.PHYSICAL	Physical security, commensurate with the value of the TOE and the data it contains, is provided by the environment.	Add this in the Section 3 Configuration Management section of the user guide. (Add some wording to reflect this in the user guide.	None
OE.TRUSTED_ADMIN	TOE Administrators are trusted to follow and apply all administrator guidance in a trusted manner.	Add this in the Section 3 Configuration Management section of the user guide. Add some wording to reflect this in the user guide.	None

Table 19: Operational Environment Objective

A.2 Common Criteria Compliant Steps

User shall follow the steps detailed below from A.3 to A.9 to put the device into Common Criteria Compliant mode of operation.

A.3 Identifying Secure Delivery of the Device

See 3eTI "Delivery Procedure" document.

A.4 Running Compliant Firmware

The following organizational assumption shall be met to operate the device in Common Criteria compliant mode.

Operational Function	Description
Banners	The device will display a customizable banner prior to the login process.
General Purpose	Only those services that are necessary for the operation, administration and support of the device are required.
Security	The physical security for the device is provided by the environment.
Admin	The security administrators should follow and apply the guidance in a safe manner.
TOE_BYPASS	Information cannot flow between the wireless client and the internal wired network without passing through the TOE.

Table 20: Device Organizational Assumption

The device must be running firmware version 5.1 in order to be compliant with the CPP_ND_V20 and PP_WLAN_AS_EP_V10. The device comes delivered with the appropriate firmware for compliance. The following procedure describes how to verify the firmware is version 5.1. Please note that the device's cryptographic engine always runs in NIST FIPS mode; there is no configuration necessary.

- Supply power to the device via a DC power supply or via POE (see Section 0 or 1.2.2.2).
- Configure the Ethernet interface on the PC with a static IP address (of 192.168.15.100 and netmask 255.255.255.0).
- Connect the interface on the PC to the Designated Management Interface on the device (see Section 1.2.1 and Section 1.2.2 for details).
- Open a supported browser on the PC and navigate to https://192.168.15.1. Log into the device with user name 'CryptoOfficer'. The default password for the CryptoOfficer user is 'CryptoFIPS'. The device will require you to change the password.
- Verify firmware 5.1 is installed by checking the 'Software version' on the 'Help' screen described in Section 2.13.9.

Software version 5.1.0.0 Build 260 for 3e-523N:

ULTRA

System Configuration 5.10.00 Build 200 UAN Software version: 1.06 Energie Software version: 1.06 Energie Software version: 1.06 Cartical State Product senial number: 7120106569 Product senial number: 7120106569 Software version: 7120106569 Product senial number: 7120106569 Software version: 7120106569 Software version: Software version: 4 Software version: Software version: Software version: Software version: Software version: Software version: Product senial number: Software version: Software version: Software v		Hardware and Software Version Information		
Noise Channel Control Software version: 5.10.00 Build 200 UNA Bridge Software version: 1.08 Environt VLAN Uboot-1 version: 1.08 MAC Address Filtering Execute version: 1.08 Certificate State 1.08 3e-523N Bails 1 Nodel Number: 258 Product part number: 258 Product part number: 723N110/0200458 Product part number: 723N110/0200458 Product senial number: 721/0569 Product senial number: 723N110/0200458 Status Software version: 4 Status Software version: 4 Wineless Media Band General Auto Environment 4 Status Software version: 51.0.00 Build 200 Software version: Software version: 51.0.00 Build 200 Software version: Software version: 50.0001/0001 </td <td>System Configuration General</td> <td></td> <td></td>	System Configuration General			
Proves Status Services Settings DNCP Server SMMP Agent Romote Administration Web Server Main User Management List All Users Add New User User Login Policy Remote Ad. Setup Two-Factor Auth Bankoning/Reports System Status Bindiging Status Bindigi	Noiny Channel Control WAN LAN Bridge Ethernet VLAN MAC Address Filtering Certificate Store Tadio 1 WLAN Mode: Mesh Point PHY Setting Wineses Mesh General Auto Bridge Security Pisor Trannel Envilse	Software version: Software version: Uboot-1 version: Uboot-2 version: Rescue version on flash: Model Number: Product serial number: Product serial number: Product serial number: Processor serial number: Processor serial number: Processor serial number: EEPROM format:	5.1.0.00 Build 200 5.1.0.00 Build 200 1.06 1.0.0 R3 Build 8 3e-523N 258 F23N18W0200456 S200068-065_1 dev: None 7121U0569 32000913-004_E dev: None 4	
Services Settings DirCP Server ShMP Agent Remote Administration Web Server List All User List All User List All User User Login Policy Remote AdA Setup Two-Factor Auth Remote Logging Rebot Default Configuration Remote Log	Fromes			
Admin User Management List All Users Add New User User Login Policy Renote AdA Setup Two-Factor Auth Banikoring/Reports System Status Bindiping Status Bindipin	Services Settings DHCP Server SNMP Agent Remote Administration Web Server			
Anikoring/Reports System Status Bindiging Status Bindiging Status Diriciping Status Diriciping Status Diriciping Status Diriciping Status System Log Web Access Log System Log Web Access Log System Configuration Audit Configuration Email Notification Confi Radio Tx Off Control System Upgrade Default Configuration Remote Logging Reboot On Demand Self-test Periodic Self-test Periodic Self-test Periodic Self-test	Idmin User Management List All Users Add New User User Login Policy Remote A&A Setup Tan, Farthy Juth			
Justem Log System Log Web Access Log Audit Log System Administration Email Notification Readio Tx Off Control System Upgrade Default Configuration Remote Logging Reboot On Demand Self-test Periodic Self-test Periodic Self-test Periodic Self-test Periodic Self-test Periodic Self-test	Ronkoring/Reports System Status Bridging Status Bridging Site Map Adjacent AP List DHCP Client List			
Auditing Audit Configuration Audit Log System Administration Email Notification Conf Radio Tx Off Control System Upgrade Definit Configuration Remote Logging Reboot On Demand Self-Aeat Periodic Self-Aeat Periodic Self-Aeat Periodic Self-Aeat Periodic Self-Aeat	.ogs System Log Wab Access Lop			
System Administration Email Notification Conf Radio Tx Off Control System Upgrade Default Configuration Remote Logging Reboot On Demand Self-test Periodic Self-test Help	Audit Configuration Audit Configuration			
enviet © 2020 Like Electronics . To Technologies International All rights sessioned	System Administration Email Notification Conf Radio Tx Off Control System Upgrade Default Configuration Remote Logging Reboot On Demand Self-test Periodic Self-test Unities			
	secretable (0.2020) Libra Flactbacker	- Terbecholes International All rights array	and a second	

Figure 106: Software Version

The following requirements are satisfied by running the device with firmware image version 5.1.

Table 21: Firmware Requirements

PPWLAN Requirement	Description
FCS_CKM.1 (1) - Cryptographic Key Generation (symmetric Keys for WPA2 Connections.	The firmware provides symmetric key generation using a FIPS approved cryptographic engine
FCS_CKM.1(2) - Cryptographic Key Generation (for asymmetric keys)	The firmware provides asymmetric key generation using a FIPS approved cryptographic engine
FCS_CKM_EXT.4 - Extended: Cryptographic Key Zeroization	The firmware zeroizes all key material when not in use. No further configuration is necessary.
FCS_COP.1 (1) - Cryptographic Operation (for data encryption/decryption)	The firmware uses a FIPS approved cryptographic engine.
FCS_COP.1 (2) - Cryptographic Operation (for cryptographic signature)	
FCS_COP.1 (3) - Cryptographic Operation (for cryptographic hashing)	
FCS_COP.1 (4) - Cryptographic Operation (for keyed- hash message authentication)	
FCS_COP.1 (5) - Cryptographic Operation (WPA2 Data Encryption/Decryption)	

PPWLAN Requirement	Description
FCS_TLS_EXT.1 - Extended: TLS	 The firmware provides a WEB interface that supports TLS version 1.1 or 1.2 with ciphers AES-CBC-128 or AES-CBC-256. The cipher suite is auto-negotiated on each web request and requires no further configuration. The following cipher suites are supported by the device: TLS_RSA_WITH_AES_128_CBC_SHA as defined in RFC 3268, TLS_DHE_RSA_WITH_AES_256_CBC_SHA as defined in RFC 3268, TLS_DHE_RSA_WITH_AES_128_CBC_SHA as defined in RFC 3268, TLS_RSA_WITH_AES_128_CBC_SHA as defined in RFC 3268, TLS_DHE_RSA_WITH_AES_256_CBC_SHA as defined in RFC 3268, TLS_RSA_WITH_AES_128_CBC_SHA256 as defined in RFC 5246, TLS_RSA_WITH_AES_256_CBC_SHA256 as defined in RFC 5246, TLS_DHE_RSA_WITH_AES_128_CBC_SHA256 as defined in RFC 5246, TLS_DHE_RSA_WITH_AES_256_CBC_SHA256 as defined in RFC 5246, TLS_DHE_RSA_WITH_AES_256_CBC_SHA256 as defined in RFC 5246,
FCS_HTTPS_EXT.1 - Extended: HTTPS	The firmware provides a WEB interface that is based on the TCP/IP and HTTPS protocols. The WEB interface does not support unsecure HTTP requests.
FCS_RBG_EXT.1 - Extended: Cryptographic Operation (Random Bit Generation)	The firmware produces random numbers for all cryptographic functions using a FIPS approved generator.
FPT_TUD_EXT.1 - Extended: Trusted Update	The device verifies all uploaded firmware is digitally signed by 3eTl before it is written to flash (see Section 2.13.3.1 Firmware Upgrade for details). The current version of TOE firmware is displayed by viewing the Help screen in Section 2.13.9.
FPT_TST_EXT.1 - Extended: TSF Testing	The device automatically runs a self-test during initial start-up. Results of the test are logged in the Audit Log. No further configuration is necessary.
FPT_FLS.1 - Fail Secure	In the case of a power on test failure, the system will halt and will be left in a secure state with network interfaces disabled. See Section 2.13.7 for self-test details.
FTP_TRP.1 - Trusted Path	The WEB interface is the trusted path between the device and remote administrators. No further configuration is necessary.

A.5 Adding Administrative Users

Users that are allowed to make changes to the device are termed "Authorized Administrators". An Authorized Administrator has either read or read/write privileges over key operational components of the device based on their configured role. Each user is assigned one of two supported roles: 'Crypto Officer' or 'Administrator' (see Section 1.4.1 for a detailed description of the 'Crypto Officer' and 'Administrator' roles).

Add administrative users to the device and set login policies in order to comply with the PPWLAN.

- Log into the WEB GUI over the Designated Management Interface by pointing your browsing to https://192.168.15.1 and logging in as the default user (CryptoOfficer).
- Create user accounts by following the guidance in Section 2.7.2 Add New User. Set each user's "Maximum Bad Password Attempt" and Password Lockout Period" attributes to lock the user out for a period of time after successive bad password attempts. Please note that "Password Lockout Period" will keep the user locked out from device access for that specified time period. The default lockout period is 10 minutes and is configurable from 0-90 minutes. In order to ensure that remote administrator authentication failures do not lead to a situation where no administrator access is available, an administrator account with local access should be configured with a lockout period of "0" minutes. When the 'Password Lockout Period' value is set to 0, the administrator won't be locked

out even after the login threshold is reached. Note that the 'Password Lockout Period' should not otherwise be set to 0.

- Enable the "Password Aging" attribute for each use and set the "Maximum Password Age" to force the user to change his password on a regular basis.
- Set the 'Minimum Password Length' to 15 characters in the User Login Policy page by following the guidance in Section 2.7.3. Set the 'Login Session Timeout' to a value between 3 and 60 minutes.
- Add a login banner as described in Section 2.3.2.

The following requirements are satisfied once administrative users have been added and the login policy has been set.

PPWLAN Requirement	Description
FIA_AFL.1 - Authentication Failure Handling	See Section 2.7 for details on how to add users and set their "Maximum Bad Password Attempt" and "Password Lockout Period" attributes.
FIA_PMG_EXT.1 - Extended: Password Management	See Section 2.7 for details on how to set passwords and password length requirements. Passwords shall be able to be composed of any combination upper and lower case letters, numbers, and the following special characters: '!', '@', '#', '\$', '%', ''', '&', '*', '(', ')'
FIA_UIA_EXT.1 - Extended: User Identification and Authentication	The device responds to ICMP echo requests and displays a login banner before initiating the authentication process. See Section 2.3.2 for details on setting the login banner.
FIA_UAU.7 - Protected Authentication Feedback	The device only provides obscured feedback while the authentication is in progress.
FIA_UAU_EXT.5 - Extended: Password-based Authentication Mechanism	The device provides a local password based authentication mechanism. See Section 2.7 for details on creating administrative users.
FIA_UAU.6 - Re-authenticating	The device requires a user to reenter his or her current password when changing that password. See Section 2.7.1 for detail on changing a user's password.
FMT_MTD.1 - Management of TSF Data (for general TSF data)	The device limits the management of the Security Functions to authorized administrators. The WEB GUI, through which the device is managed, prevents the reading of password-based authentication data, pre-shared keys, symmetric keys or private keys.
FMT_SMF.1 - Specification of Management Functions	The device provides management functionality through the WEB interface.
FMT_SMR.1 - Restrictions on Security Roles	The device maintains user roles and assigns local users to those roles. Access to security functions is limited by the roles. See Section 1.4 for details on security roles.
FTA_SSL.3 - TSF-initiated Termination	WEB interface sessions are terminated when inactive for a configurable amount of time. See Section 2.7.3 for detail on 'Login Session Timeouts'.
FTA_SSL.4 - User-initiated Termination	The device's WEB interface provides a 'Log Out' button on every web page.
FTA_SSL_EXT.1 - Extended: TSF-initiated Session Locking	WEB interface sessions are terminated when inactive for a configurable amount of time. See Section 2.7.3 for detail on 'Login Session Timeouts'.
FTA_TAB.1 - Default TOE Access Banners	A login banner is displayed before establishing a WEB interface session. See Section 2.3.2 for details on setting the login banner.
FMT_MOF.1 - Management of Security Functions Behavior	Management of security functions is limited to authorized administrators through the WEB interface. See Section 1.4 for details on security roles.

Table 22: Administration and Login Policy Requirements

A.6 Configuring System Time

The device requires a reliable time source for proper operation. The device is equipped with a real-time clock that is used as the local time source. Additionally, the device can be configured to synchronize its real-time clock with an external NTP time source. The real-time clock must be accurately set and synchronized with an external time source through an IPsec tunnel in order to operate in a Common Criteria environment.

- Add an IPsec Tunnel Profile to the device by following the guidance in Section 0. Ensure that the Cipher Suite selected is <u>NOT</u> "Suite B GMAC 128" or "Suite B GMAC 256". These Cipher Suites do not provide confidentiality and are not compliant with the PPWLAN.
 - a) The IPsec tunnel SA re-key timer shall be configured to the range within 24 hours (1440 minutes, refer to Section 0),
 - b) The IPsec tunnel child SA re-key timer shall be configured to the range within 8 hours (480 minutes, refer to Section 0).
- Configure the System Time to synchronize with an external NTP time source by following the guidance in Section 2.3.2. Check the IPsec Tunnel check box and select the IPsec profile configured above.

The following requirements are satisfied once the time has been accurately set.

Table 23: Time Stamp Requirements

NDPP Requirement	Description
FPT_STM.1 - Reliable Time Stamps	The device provides reliable timestamps using a local real time clock optionally synchronized with an external source. See Section 2.3.2 for setting the system time.

A.7 Configuring Access Point Mode

The device supports multiple modes of wireless communication, including Access Point, Mesh Access Point, Mesh Point and Client mode. In order to operate in a Common Criteria environment, the device must be configured in Access Point mode.

- Follow the guidance in Section 2.4.2 and place the device in Wireless Access Point mode.
- Complete the PHY settings in Section 2.4.3.
- Complete the Access Point General Settings in Section 2.4.4. By default the device does not allow management sessions from wireless clients. Update the "Management from Client" attribute on the AP General page if wireless clients need access to the WEB interface.
- Complete the Wireless VLAN Mapping configuration by making sure the "Wireless VLAN" attribute is disabled in Section 2.4.4.4.

In a Common Criteria environment, the Access point must use 802.1x when authenticating wireless clients. Additionally communications with the Authentication Server (RADIUS) must be protected using IPsec Tunnels.

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com



- Add an IPsec Tunnel Profile to the device by following the guidance in Section 0. Ensure that the Cipher Suite selected is <u>NOT</u> "Suite B GMAC 128" or "Suite B GMAC 256". These Cipher Suites do not provide confidentiality.
 - a) The IPsec tunnel SA rekey timer shall be configured to the range within 24 hours (1440 minutes, refer to Section 0),
 - b) The IPsec tunnel child SA rekey timer shall be configured to the range within 8 hours (480 minutes, refer to Section 0).
- 2) Complete the Access Point Security Settings in Section 2.4.4.2. Select 802.1x as the authentication method. Check the IPsec Tunnel checkbox and select the IPsec Tunnel profile added in the step above. Configure the 'Radius Server Type' to be IETF and fill in the RADIUS server address and shared secret.

The following PPWLAN requirements are satisfied once the device is configured in Access Point mode as described above.

NDPP Requirement	Description
FIA_8021X_EXT.1 - Extended: 802.1X Port Access Entity (Authenticator) Authentication.	The device performs the 802.1x authenticator role and does not provide wireless clients access until a successful authentication exchange is completed. Communication between the device and the RADIUS server conforms to RFC 2865 and 3579. See Section 2.4.4.2 for Access Point Security settings.
FCS_IPSEC_EXT.1 - Extended: Internet Protocol Security (IPsec) Communications	The device tunnels RADIUS traffic using IPsec ESP protocol. IKEv2 is used for authentication and key exchange. NAT traversal is supported. See Section 0 for building IPsec Tunnel Profiles.
FIA_PSK_EXT.1 - Extended: Pre-Shared Key Composition	The device can use Pre-Shared keys for IPsec authentication. The keys can be entered as text or a bit string. See Section 0 for building IPsec Tunnel Profiles.
FIA_X509_EXT.1 - Extended: X509 Certificates	The device supports X509v3 certificates used in IPsec and TLS authentication. The device protects certificates from unauthorized deletions and modification through limiting access to authorized entities using the WEB interface. See Section 0 for building IPsec Tunnel Profiles. See Section 2.6.6 for configuring the WEB Server Certificate
FTP_ITC.1 - Inter-TSF trusted channel	All communication between the device and the Authentication Server , NTP server and remote log server is through one or more IPsec tunnels. See Section 0 and Section 2.4.4.2 for configuration details.
FCS_CKM.2 (1) - Cryptographic Key Distribution (PMK)	All RADIUS based key distribution is done through an IPsec tunnel and does not expose the cryptographic keys.
FCS_CKM.2 (2) - Cryptographic Key Distribution (GTK)	All RADIUS based key distribution is done through an IPsec tunnel and does not expose the cryptographic keys.

Table 24: Access Point Security Requirements

A.8 Configuring Wireless Client Session Establishment

The device can deny establishment of wireless clients based on MAC address, time, day and per radio interface.

• Follow the guidance in Section 1.3.3 "MAC Address Filtering" to limit wireless client access.

The following PPWLAN requirements are satisfied.

Table 25: Access Point Session Establishment

NDPP Requirement	Description
FTA_TSE.1 - TOE Session Establishment	The device is able to deny establishment of a wireless client session based on MAC address, time or day.

The "MAC Address Filtering" is per wireless interface based, so that a wireless client can be added to the "deny list" on one interface while be on the "allowed list" on the other interface. To make this happen, first follow Section 2.4 to configure each radio into "Access Point" mode, then follow Section 1.3.4 to configure the "MAC Address Filtering" for each radio

A.9 Configuring Audit Event Loggin

The device generates audit log records for the following security related events:

- All use of the identification and authentication mechanism.
- Any attempt at unlocking an interactive session.
- The termination of a remote session by the session-locking mechanism.
- The termination of an interactive session.
- Changes in system time.
- Initiation or Failure of a firmware update.
- Initiation, Termination or Failure of a trusted channel.

Each audit log record is stored locally on the device and can be viewed by an authorized administrator with a 3e-local role (see Section 2.12.2 for viewing audit log records). In addition to storing records locally, the device must be configured to transmit audit log records to a remote auditing server over a trusted channel using an IPsec tunnel. Since the volume of audit log records can become excessive, an authorized administrator with a 'Crypto Officer' role can choose not to log categories of audit events—IPsec Security Policy success audit records, for example, can be muted and not logged.

Audit log records are stored locally on the device in 256K of persistent memory. Once this memory is full, new audit record will rotate out the oldest records and the old records will be lost.

In order to operate in a Common Criteria environment the following audit records must be enabled:

- 'Self-Test' Success and Failure audit records.
- 'Management Connection' Success and Failure audit records.
- 'Admin User Authentication' Success and Failure audit records.
- 'System Configuration' Success and Failure audit records.
- 'Software Update' Success and Failure audit records.

See Section 2.12.1 in order to verify the above audit records are enabled. Additionally, the device must be configured to transmit audit records to a remote auditing server through an IPsec tunnel.

- Add an IPsec Tunnel Profile to the device by following the guidance in Section 0. Ensure that the Cipher Suite selected is <u>NOT</u> "Suite B GMAC 128" or "Suite B GMAC 256". These Cipher Suites do not provide confidentiality and are not compliant with the.
 - a) The IPsec tunnel SA rekey timer shall be configured to the range within 24 hours (1440 minutes, refer to Section 0),
 - b) The IPsec tunnel child SA rekey timer shall be configured to the range within 8 hours (480 minutes, refer to Section 0).

2) Enable the Remote Auditing and provide the 'Audit Server Address' and 'Audit Server Port' described in Section 2.12.1. Be sure to check the IPsec Tunnel check box and select the IPsec Tunnel profile add in the above step.

The following PPWLAN requirements are satisfied once the audit logging is configured.

PPWLAN Requirement	Description
FAU_GEN.1 - Audit Data Generation	The device audit logs startup and shutdown of the audit function.
FAU_GEN.2 - User Identity Association	The user ID is included in all audit log records resulting from the action of administrative users.
FAU_SEL.1 - Selective Audit	See Section 2.12.1 for details on selecting auditable events.
FAU_STG.1 - Protected Audit Trail Storage (Local Storage)	The device stores the last 256KB of audit records and prevents their removal. No further configuration is necessary.
FAU_STG_EXT.1 - Extended: External Audit Trail Storage	See Section 2.12.1 for details on configuring a remote audit log server.
FAU_STG_EXT.3 - Action in Case of Loss of Audit Server Connectivity	The device stops passing packets and logs a communication error when Audit Server connectivity is lost. No further configuration is necessary.
FAU_STG_EXT.4 - Prevention of Audit Data Loss	See Section 2.12.1 for configuring the audit service to either stop logging or rotate the logs when the local audit trail is full.
FAU_SAR.1 - Audit Review	See Section 2.12.2 for details on reviewing audit records.
FAU_SAR.2 - Restricted Audit Review	Read access to audit records is limited to authorized administrators with the 'CryptoOfficer' role. See Section 1.4.1 for details on Management User Roles.

Table 26: Audit Log Requirements

A.10 Understanding Audit Events

A.10.1 Audit Event Record Structure

The following is a sample Audit Event message:



Table 27 describes the fields in an Audit Event message. If the Audit Event logging utility cannot determine the value for a particular field, a space character appears followed by the comma delimiter.

Table 27: Fields in Audit Event Message

Field	Description
Index Number	The Audit Event logging utility prepends each message with an index number that is incremented with every message. The index is set to '1' whenever the system is rebooted. The index wraps back around to '1' after its maximum value of '99999'.
Timestamp	Time when the message was generated in the form 'MMM DD HH:mm:SS YYYY' MMM - Abbreviated month name DD - Day of month as 1 or 2 characters HH - Hour (0-23) mm - Minute (0-59) SS - Second (0-60) YYYY - Year Time is in local time.
Success/Failure Flag	Flag indicating if the audit record was generated from a successful event or an event that failed. 1 - The event was successful 2 - The event failed
TAG	3eTI-enumerated message tag which uniquely identifies the message type.
User Name	Authorized administrator that initiated the event.
Message-text	Description of the event.
Source Address	Source IP or MAC address of the network packet that initiated the event.
Destination Address	Destination IP or MAC address of the network packet that initiated the event.

A.10.2Audit Event: Audit Startup and Shutdown

The Audit Event logging utility is started automatically during the system bootup. The utility cannot be stopped other than by rebooting the system. When the device is restarted, the audit log contains the following information:

- 69 Sep 11 14:11:44 2017 1 EVT_AUDIT_START_STOP, , Audit Log is shutting down, , ,
- 1 Sep 11 14:13:12 2017 1 EVT_AUDIT_START_STOP, , Audit Log is enabled on startup, , ,

A.10.3 Audit Event: Audit Coverage Change

Audit log messages are generated when the audit configuration is changed. The following messages are generated as a result of successful audit configuration change through the WEB GUI:

- 307 Sep 11 15:01:05 2017 1 EVT_AUDIT_CONFIG_CHANGE, CryptoOfficer, Audit configuration has been updated, , ,
- 309 Sep 11 15:01:05 2017 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Updated audit configuration, , ,

The following messages are generated as a result of failed audit configuration change through the WEB GUI:

 318 Sep 11 15:03:48 2017 2 EVT_AUDIT_CONFIG_CHANGE, CryptoOfficer, Invalid audit server address, , ,

PROPRIETARY INFORMATION: Use or disclosure of this data is subject to the restrictions on the title page of this document. Ultra Electronics, 3eTI • 12410 Milestone Center Drive, Germantown MD 20876 • 800.449.3384 • www.ultra-3eti.com

 319 Sep 11 15:03:48 2017 2 EVT_SYSTEM_CONFIG, CryptoOfficer, Failed to update audit configuration, , ,

A.10.4 Audit Event: Loss of Connectivity to Server

Audit log messages are generated when the link to the external audit server is not available. The following messages are generated as a result of losing connectivity to the audit server 192.168.205.9:

- 454 Sep 11 15:30:45 2017 2 EVT_MANAGEMENT_CONNECTION, , Connection Error: Audit data waiting to be sent out, , 192.168.205.9:11515,
- 612 Sep 11 15:41:11 2017 2 EVT_MANAGEMENT_CONNECTION, , Failed to initiate a TCP connection, , 192.168.205.9:11515,

A.10.5Audit Event: Administrative Actions Event

The following are some examples of audit log messages recording administrative actions:

- 1) User Login and Logout:
 - a) 344 Sep 11 15:16:35 2017 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, httpd User CryptoOfficer from 192.168.15.100 logged in successfully, , ,
 - b) 352 Sep 11 15:17:34 2017 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, httpd User CryptoOfficer from 192.168.15.100 logged out, , ,
- 2) Configure User:
 - a) 364 Sep 11 11:37:44 2017 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, Added admin-user TestUser1 role=Administrator, , ,

3) Modify User Password:

- a) 372 Sep 11 11:41:46 2017 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, Changed admin-user password for TestUser1 in Administrator role, , ,
- 4) Configure AP 802.1x Security Method:
 - a) 432 Sep 11 12:03:33 2017 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i sub-algorithm is changed from PSK Passphrase to 802.1x, , ,
 - b) 436 Sep 11 12:03:33 2017 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i Primary Radius Server IP changed: old=, new=192.168.205.9, , ,
 - c) 437 Sep 11 12:03:33 2017 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i Radius Shared Secret changed, , ,
 - d) 438 Sep 11 12:03:33 2017 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Updated AP security configuration, , ,
- 5) Configure AP PSK with Master Key:
 - a) 647 Sep 11 15:44:17 2017 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i sub-algorithm is changed from 802.1x to PSK Master Key, , ,
 - b) 648 Sep 11 15:44:17 2017 1 EVT_KEY_GENERATION, CryptoOfficer, AP1 FIPS-802.11i PSK Master Key changed., , ,

- c) 649 Sep 11 15:44:17 2017 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i PSK Master Key changed., , ,
- d) 651 Sep 11 15:44:17 2017 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Updated AP security configuration, , ,

6) Configure AP PSK with Passphrase:

- a) 654 Sep 11 15:45:15 2017 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i sub-algorithm is changed from PSK Master Key to PSK Passphrase, , ,
- b) 655 Sep 11 15:45:15 2017 1 EVT_KEY_GENERATION, CryptoOfficer, AP1 FIPS-802.11i PSK Master Key changed., , ,
- c) 656 Sep 11 15:45:15 2017 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i PSK Master Key changed., , ,
- d) 657 Sep 11 15:45:15 2017 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Updated AP security configuration, , ,

7) Login banner change:

 a) 7 Apr 12 18:48:18 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Terms & Condition changed: old={This is first banner message second line} new={This is second banner message second line third line}, ,

8) Configure session inactivity and timeout:

a) 10 May 13 09:52:05 2018 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, Login session timed out., , ,

9) Configure IPsec functions:

a) Add IPsec profile:

,

- i) 14 May 13 09:53:23 2018 1 EVT_KEY_ZEROIZATION, , Zeroized IPsec Profile PSK, ,
- ii) 15 May 13 09:53:23 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, IPsec Profile first_profile added, , ,
- iii) 16 May 13 09:53:23 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Add VPN profile has been successful, , ,

10) Update IPsec profile:

- a) 18 May 13 09:54:36 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, IPsec Profile first_profile Updated, , ,
- b) 19 May 13 09:54:36 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Update VPN profile has been successful, , ,

11) Use IPsec profile [NTP]:

- a) 13 Apr 12 19:02:11 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, NTP server 1 IP changed, old= new=192.168.2.1, , ,
- b) 14 Apr 12 19:02:11 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, NTP server 1 IPsec profile changed, old=, new=second_profile, , ,

12) Generating/import of, changing, or deleting of cryptographic keys WiFi PSK change:

- a) 31 Apr 12 19:05:34 2018 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i sub-algorithm is changed from PSK Master Key to PSK Passphrase, , ,
- b) 32 Apr 12 19:05:34 2018 1 EVT_KEY_GENERATION, CryptoOfficer, AP1 FIPS-802.11i PSK Master Key changed., , ,
- c) 33 Apr 12 19:05:34 2018 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i PSK Master Key changed., , ,
- d) 34 Apr 12 19:05:34 2018 1 EVT_ENCRYPT_ALG_CHANGED, CryptoOfficer, AP1 FIPS-802.11i Group Key Life Time changed from 600 to 900 (minutes), , ,

13) IPsec PSK change:

- a) 40 Apr 12 19:07:29 2018 1 EVT_KEY_ZEROIZATION, CryptoOfficer, Zeroized IPsec Profile PSK, , ,
- b) 41 Apr 12 19:07:29 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, IPsec Profile second_profile Updated, , ,

14) Import device certificate and private key:

a) 8 Apr 12 19:00:27 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Device Certificate [US Maryland Ultra-3eti Testing device1 device1@3eti.com] uploaded successfully., , ,

15) Delete device certificate and private key:

a) 73 Apr 12 19:13:26 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Deleted /csp/certs/device/cert/1.cert [US Maryland Ultra-3eti Testing device1 device1@3eti.com], , ,

A.10.6Audit Event: Failure of Key Generation/Distribution

Audit log messages are generated when any failure of key generation/distribution occurs. The following message is generated as a result of GMK rekey failure:

6593 May 11 10:38:29 2015 2 EVT_KEY_GENERATION, , failed to get random data for GMK rekeying, , 00:0e:8e:4b:ba:3d,

The following is an example of an audit log message resulting from an ephemeral key generation failure occurring on the VPN connection to the audit server 192.168.205.9:

 6023 May 11 13:10:36 2015 2 EVT_KEY_GENERATION, , audit1 VPN [192.168.205.9]: Ephemeral key generation failed, , ,

A.10.7 Audit Event: Failure of Encryption or Decryption

Audit log messages are generated when encryption or decryption of user data fails. The following is an example of an audit log message resulting from data decryption failure:

 3544 May 10 13:11:30 2015 2 EVT_ENCRYPT_DECRYPT_ERR, , audit1 VPN [192.168.205.9]: Encryption / Decryption failed, , ,

A.10.8Audit Event: Failure of Cryptographic Signature

Audit log messages are generated when cryptographic signing fails. The following is an example of an audit log message resulting from cryptographic signature failure:

 3614 May 10 13:11:52 2015 2 EVT_CRYPTO_SIGNATURE_ERR, , audit1 VPN [192.168.205.9]: Cryptographic signature failed, , ,

A.10.9Audit Event: Failure of Hashing Function

Audit log messages are generated when cryptographic hashing fails. The following is an example of an audit log message resulting from a hashing failure:

 3621 May 10 13:12:02 2015 2 EVT_CRYPTO_HASHING_ERR, , audit1 VPN [192.168.205.9]: Cryptographic hashing function failed, , ,

A.10.10 Audit Event: Failure of WPA2 Encryption or Decryption

Audit log messages are generated when WPA2 encryption or decryption fails. The following is an example of an audit log message resulting from WPA2 decryption failure:

• 3744 May 10 13:18:30 2015 2 EVT_ENCRYPT_DECRYPT_ERR, , mac80211 CCMP Replay packet, 02:15:6d:84:f4:c4, 00:0e:8e:4b:ba:3d,

A.10.11 Audit Event: Failure to Establish a HTTPS/TLS Session

Audit log messages are generated when there is a failure in the TLS handshake—for example, the following message is generated when there is a TLS handshake failure:

 70 Mar 18 11:58:41 2015 2 EVT_MANAGEMENT_CONNECTION, , TLS handshake failure - no matching cipher, 192.168.15.201, ,

Other failure reasons could include mismatching TLS versions, TLS authentication failure (if the TLS client refuse to trust the TOE's server certificate).

A.10.12 Audit Event: Establishment or Termination of a HTTPS/TLS Session

Audit log messages are generated each time a TLS session is established or terminated. The following message is generated when there is a new TLS session:

• 83 Mar 18 11:58:42 2015 1 EVT_MANAGEMENT_CONNECTION, , TLS handshake success, 192.168.15.201, ,

The following message is generated when a TLS session is terminated:

• 86 Mar 18 11:58:44 2015 1 EVT_MANAGEMENT_CONNECTION, , TLS session terminated, 192.168.15.201, ,

A.10.13 Audit Event: Failure of IPsec Security Association Establishment

Audit log messages are generated each time an IPsec Security Association fails to be established. The following is an example of an audit log message resulting from an IPsec Security Association failure:

 3244 May 10 13:12:53 2015 2 EVT_IPSEC_SA, , audit1 VPN [192.168.205.9]: Traffic selector mismatch, , ,

The device verifies that the algorithm used in the ESP SA (IKEv2 CHILD_SA) is less than or equal to the algorithm of the parent IKE_SA. If during the negotiation there is no algorithm that fits this requirement, the device will send a NO_PROPOSAL_CHOSEN message and the CHILD_SA will not be established, with the following audit events generated:

- 150 Sep 12 11:14:39 2017 2 EVT_IPSEC_SA, , ntp11 VPN [192.168.205.9]: IPsec cryptographic algorithm mismatch, , ,
- 151 Sep 12 11:14:39 2017 2 EVT_IPSEC_SA, , ntp11 VPN [192.168.205.9]: SA removed due to NO_PROPOSAL_CHOSEN notification from peer, , ,

A.10.14 Audit Event: Establishment or Termination of an IPsec Security Association

Audit log messages are generated each time an IPsec Security Association is established or terminated. The following message is generated when there is a new IPsec Security Association:

- 217 Mar 26 14:30:25 2015 1 EVT_IPSEC_SA, , SAD-add src=192.168.254.254 dst=192.168.205.9 spi=3405309099(0xcaf8e4ab) res=1, ,
- 218 Mar 26 14:30:25 2015 1 EVT_IPSEC_SA, , SAD-add src=192.168.254.254 dst=192.168.205.9 spi=3470987138(0xcee30f82) res=1, ,,

The following message is generated when an IPsec Security Association is terminated:

- 222 Mar 26 14:36:30 2015 1 EVT_IPSEC_SA, , SAD-delete src=192.168.254.254 dst=192.168.205.9 spi=3405309099(0xcaf8e4ab) res=1, ,
- 223 Mar 26 14:36:30 2015 1 EVT_IPSEC_SA, , SAD-delete src=192.168.254.254 dst=192.168.205.9 spi=3470987138(0xcee30f82) res=1, ,

A.10.15 Audit Event: Failure of Random Bit Generation

Audit log messages are generated when random bit generation fails. The following is an example of an audit log message resulting from random bit generation failure:

 3534 May 10 13:15:01 2015 2 EVT_RBG_ERR, , audit1 VPN [192.168.205.9]: Random number generator failed, , ,

A.10.16 Audit Event: Authentication Failure Handling

Audit log messages are generated when the defined number of unsuccessful authentication attempts has been met. The following message is generated when the threshold of 3 login attempts with invalid password has reached:

 145 May 12 15:00:38 2015 2 EVT_ADMIN_USER_AUTH, TestUser, User TestUser account locked (threshold of 3 login attempts with invalid password reached), , ,

A.10.17 Audit Event: Admin User Authentication

Audit log events are generated whenever an administrative user successfully or unsuccessfully attempts WEB interface access. Additionally, logout events are recorded when an administrative user successfully ends a WEB interface session by clicking the log-off button—for example, the following logs show a failed authentication attempt followed by a successful login and a logout:

- 39 Mar 18 11:49:59 2015 2 EVT_ADMIN_USER_AUTH, CryptoOfficer, httpd User CryptoOfficer from 192.168.205.208 could not be authenticated, , ,
- 41 Mar 18 11:50:07 2015 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, httpd User CryptoOfficer from 192.168.205.208 logged in successfully, , ,
- 45 Mar 18 11:55:56 2015 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, httpd User CryptoOfficer from 192.168.205.208 logged out, , ,

A.10.18 Audit Event: 802.1X Authentication

Audit log messages are generated when 802.1X authentication request is accepted or rejected for a wireless client. The following messages show a successful authentication attempt (Access Accept) and a failed authentication attempt (Access Reject) for the wireless client 8c:70:5a:40:f0:a0:

- 8104 May 12 14:42:05 2015 1 EVT_STA_AUTH, , Access Accept from auth server, , 8c:70:5a:40:f0:a0,
- 8104 May 12 14:42:05 2015 2 EVT_STA_AUTH, , Access Reject from auth server, , 8c:70:5a:40:f0:a0,

The following message is an example of audit log message showing the device received a data frame from the not associated client 8c:70:5a:40:f0:a0:

 8168 May 12 13:01:05 2015 2 EVT_KEY_TRANSFER_ERR, , IEEE 802.1x data frame from not associated STA, , 8c:70:5a:40:f0:a0,

A.10.19 Audit Event: Certificate Validation and Upload

Audit log messages are generated when the certificates fail to be validated—for example, the following messages are generated as a result of attempting to upload an invalid certificate through the WEB GUI:

- 7617 May 11 16:12:03 2015 2 EVT_CERT_VALIDATION_ERR, CryptoOfficer, Issuer certificate(s) upload failed: invalid or unsupported file format., , ,
- 7618 May 11 16:12:03 2015 2 EVT_SYSTEM_CONFIG, CryptoOfficer, Failed to upload Issuer certificate, , ,

The following message is generated a result of uploading an issuer certificate through the WEB GUI:

8239 May 12 15:36:55 2015 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Uploaded Issuer certificate,
 , ,

A.10.20 Audit Event: Failure of the TSF

Audit log messages are generated when any failure of the TSF is detected—for example, the following messages are generated as a result of RNG self-test failures:

- 7 May 11 10:01:07 2015 2 EVT_SELF_TEST, , Random Number Generator (openssl) selftest failed, , ,
- 30 May 11 10:01:08 2015 2 EVT_SELF_TEST, , The self-test failed, , ,

A.10.21 Audit Event: Changes to the System Time

Audit log messages are generated when the date or time on the device is changed—for example, the following messages are generated as a result of changing the date manually through the WEB GUI:

- 3544 Mar 18 13:00:36 2015 1 EVT_SYSTEM_CONFIG, CryptoOfficer, Updated general system configuration, , ,
- 3557 Mar 18 02:00:35 2015 1 EVT_TIME_CHANGE, , System time has been changed old_timestamp: 1426701636 new_timestamp: 1426662035, 192.168.205.208, ,

Please note the numbers in the logs above are UNIX epoch. There are many stand-alone or online tools to convert epoch to human readable time stamps.

A.10.22 Audit Event: Self-Test

During device's power up process, the firmware runs algorithm known answer tests, the audit messages shown below indicate the test results. In case of any test failure, the system will halt and the module will not be operable. The status output LED GPIO pins will be set high to indicate the system halt condition. User can reboot the device in attempts to eliminate the self-failure, if continuous failure shows up, the device should be returned to Ultra-3eTI for repairs.

Audit log messages are generated when the device's Security Functions self-test is completed. The self-test is run automatically on system boot up. The following messages are generated as each self-test function completes:

- 2 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , The system wide self-test has been triggered, , ,
- 3 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Advanced Encryption Standard ECB mode (openssl) self-test passed, , ,
- 4 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Advanced Encryption Standard CBC mode (openssl) self-test passed, , ,
- 5 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Secure Hash Algorithm 1 (openssl) self-test passed,
 , ,
- 6 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Secure Hash Algorithm 2 (openssl) self-test passed, , ,
- 7 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Random Number Generator (openssl) self-test passed, , ,

- 8 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Hashed Message Authentication Code (openssl) self-test passed, , ,
- 9 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , RSA Algorithm (openssl) self-test passed, , ,
- 10 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Triple DES (openssl) self-test passed, , ,
- 11 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , TLS-KDF Key Distribution Function (openssl) selftest passed, , ,
- 12 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , KAS (Key Agreement Scheme) (openssl) self-test passed, , ,
- 13 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Advanced Encryption Standard ECB mode (hardware kernel) self-test passed, , ,
- 14 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Advanced Encryption Standard CCM mode (hardware kernel) self-test passed, , ,
- 15 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Advanced Encryption Standard CBC mode (hardware kernel) self-test passed, , ,
- 16 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Advanced Encryption Standard CMAC mode (hardware kernel) self-test passed, , ,
- 17 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Secure Hash Algorithm 1 (hardware kernel) selftest passed, , ,
- 18 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Secure Hash Algorithm 2 (hardware kernel) selftest passed, , ,
- 19 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Hashed Message Authentication Code (hardware kernel) self-test passed, , ,
- 20 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Triple DES ECB (hardware kernel) self-test passed, , ,
- 21 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Triple DES CBC (hardware kernel) self-test passed, , ,
- 22 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , ECDSA Verification self-test passed, , ,
- 23 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Firmware Integrity Check self-test passed, , ,
- 24 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Bootloader Integrity Check self-test passed, , ,
- 25 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , Key Error Detection self-test passed, , ,
- 26 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , The self-test passed, , ,
- 27 Sep 11 14:13:13 2017 1 EVT_SELF_TEST, , power up test run, , ,

A.10.23 Audit Event: Successful or Failed System Updates

An audit log message is generated on initiation of system software updates—for example, the following messages are generated as a result of success firmware update:

- 77 Sep 12 15:10:49 2017 1 EVT_SOFTWARE_UPDATE, CryptoOfficer, System software update initiated, 192.168.15.100, ,
- 78 Sep 12 15:11:08 2017 1 EVT_SOFTWARE_UPDATE, CryptoOfficer, System software update has been successful, 192.168.15.100, ,

The following messages are generated as a result of failed firmware update due to ECDSA signature mismatch:

- 59 Sep 12 15:16:33 2017 1 EVT_SOFTWARE_UPDATE, CryptoOfficer, System software update initiated, 192.168.15.100, ,
- 61 Sep 12 15:16:40 2017 2 EVT_SOFTWARE_UPDATE, CryptoOfficer, System software update failure: ECDSA signature mismatch, 192.168.15.100, ,

The following messages are generated as a result of failed firmware update due to HMAC integrity check error:

- 85 Sep 12 15:20:32 2017 1 EVT_SOFTWARE_UPDATE, CryptoOfficer, System software update initiated, 192.168.15.100, ,
- 86 Sep 12 15:20:38 2017 2 EVT_SOFTWARE_UPDATE, CryptoOfficer, System software update failure: HMAC check error, 192.168.15.100, ,

A.10.24 Audit Event: Maximum Quota

Audit log messages are generated when the maximum quotas of system resources have been reached – for example, the following message is generated as a result of the maximum 64 wireless clients being exceeded:

• 324 Apr 29 15:48:28 2015 2 EVT_RESOURCE_QUOTA_EXCEED, , No more room for new STAs, 8c:70:5a:40:f0:a0, ,

A.10.25 Audit Event: Attempts at Unlocking an Interactive Session

Sessions are terminated, not locked, due to inactivity and, therefore, there is no unlocking audit record. The authorized administrative user is required to re-authenticate thereby creating a new session. At that time, a login audit record will be generated.

A.10.26 Audit Event: Termination of a Remote Session by the Session Locking Mechanism

The remote session will be terminated by the device when the inactivity time period exceeds the configured value. The session timeout value can be configured under the "User Login Policy" page, covered under Section 2.7.3.

Audit log messages are generated when remote sessions are terminated due to inactivity—for example, the following message is generated when a remote session is terminated due to inactivity:

• 32 Mar 18 15:07:09 2015 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, Login session timed out., ,

A.10.27 Audit Event: Termination of an Interactive Session

Audit log messages are generated whenever an authenticated administrator logs off the device—for example, the following message is generated as a result of a log off:

 37 Mar 18 15:08:00 2015 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, httpd - User CryptoOfficer from 192.168.205.208 logged out, , ,

A.10.28 Audit Event: Denial of a Session Establishment due to MAC Address Filtering

Audit log messages are generated when a wireless client's access is denied based on MAC address, time and day—for example, the following messages are generated as a result of MAC address filtering:

- 8321 May 12 16:07:35 2015 2 EVT_STA_AUTH, , 802.11 ACL is active: REJECT STA, 00:0e:8e:4b:ba:3d, ,
- 8322 May 12 16:07:35 2015 2 EVT_STA_AUTH, , 802.11 mgmt authentication failure, , 00:0e:8e:4b:ba:3d,

A.10.29 Audit Event: Initiation of a Trusted Channel

Audit log messages are generated when a trusted channel is initiated. Trusted channels include TLS/HTTPS sessions used in the WEB interface as well as TLS sessions for remote audit logging. The following is an example of an audit log message when a remote audit logging trusted channel is established:

• 70 May 11 15:12:25 2015 1 EVT_MANAGEMENT_CONNECTION, , TLS Initiated, , 192.168.205.23:8889,

The following is an example of an audit log message when a WEB interface trusted channel is established:

• 131 May 11 15:20:57 2015 1 EVT_MANAGEMENT_CONNECTION, , TLS Initiated, 192.168.205.208:23253, ,

A.10.30 Audit Event: Termination of a Trusted Channel

The following is an example of an audit log message when a remote audit logging trusted channel is terminated:

 139 May 11 15:28:36 2015 1 EVT_MANAGEMENT_CONNECTION, , TLS Terminated, , 192.168.205.23:8889,

The following is an example of an audit log message when a WEB interface trusted channel is terminated:

• 132 May 11 15:20:57 2015 1 EVT_MANAGEMENT_CONNECTION, , TLS Terminated, 192.168.205.208:23253, ,

A.10.31 Audit Event: Failure of Trusted Channel or Trusted Path Functions

The following is an example of an audit log message when a remote audit logging trusted channel fails:

• 143 May 11 15:29:34 2015 2 EVT_MANAGEMENT_CONNECTION, , Failed to initiate a TCP connection to the audit server, , 192.168.205.9:8889,

A.10.32 Audit Event: Trust Channel Connection Re-establishment

No recovery is needed as the TOE will automatically attempt to establish a new connection.

A.10.33 Audit Event: Load Certificates that Fail to Meet the NDcPP Requirements

The following are some examples of audit log messages when loading certificates that fail to meet the NDcPP requirements:

- 278 Sep 11 11:12:31 2017 2 EVT_CERT_VALIDATION_ERR, CryptoOfficer, CRL issuer does not have CRL-sign capability. CRL cannot be used!, , ,
- 335 Sep 11 11:26:50 2020 2 EVT_CERT_VALIDATION_ERR, CryptoOfficer, Certificate is expired. Make sure your system clock is accurate., , ,

A.10.34 Audit Event: Configure Session Inactivity and Time out

• 366 Sep 11 11:37:44 2017 1 EVT_ADMIN_USER_AUTH, CryptoOfficer, Changed admin-user TestUser1 role=Administrator lockout period=3, session timeout=9,,,

A.10.35 Audit Event: Detection of Modification of Channel Data for 802.11 and 802.1x

 8168 May 12 13:01:05 2015 2 EVT_KEY_TRANSFER_ERR, , IEEE 802.1x data frame from not associated STA, , 8c:70:5a:40:f0:a0,

A.10.36 Audit Event: Configure/Change the Reference Identifier for Peer

• 10 Apr 13 19:37:59 2018 1 EVT_SYSTEM_CONFIG, CryptoOfficer, NTP server 1 IPsec server id changed, old=636N-L3, new=636N-L2, , ,

A.11 Key Zeroization

The device zeroizes keys when it no longer uses them. The user can choose to reboot (Section 2.13.6) the device thus zeroizes the keys in RAM or choose "Factory Default" (Section 2.13.4) which will zeroize all keys in RAM and FLASH.

A.12 User Space Processors

The following user space processes typically run on the device:

Table 28: The List of User Space Processes

Process Name	Process Description
udevd	The device/event managing daemon to the Linux kernel
rngd	This daemon checks and feeds random data from hardware device to kernel random device
masterDaemon	The master daemon that start, stop and monitor 3eTI applications

Process Name	Process Description
syslogd	The Linux system logging daemon
klogd	The Kernel Log Daemon
configDaemon	The system configuration daemon
rtocd	The Radio Transfer Off Control (rtoc) daemon for scheduling and maintaining radio silence
syscmdd	The system command scheduler daemon
miscHWD	The miscellaneous hardware daemon that controls hardware buttons and LED displays
self-testd	The daemon for performing on-demand or periodic self-test
mini_httpd	The HTTPS web server daemon
radard	The daemon to perform radar signal detection and dynamic frequency selection
radioPHYd	The daemon that monitors and controls radio PHY settings
xcManager	The XML-based system configuration manager daemon
hostapd	This daemon provides IEEE 802.11 access point management functionality and implements IEEE 802.1X Authenticator to enforce WLAN client authentication.
auditlogd	The user space security auditing daemon
starter	The IPsec starter daemon that starts and configures the keying daemon charon
charon	The IPsec keying daemon for establishing IKEv2 VPN connections
ntpd	The daemon that sets and maintains the system time in synchronization with time servers using the Network Time Protocol (NTP)

Note: All processes listed run at hardware privilege ring 1 and software privilege as root.
Appendix B. Term Reference Guide

B.1 Acronyms and Abbreviations

3DES	-	Triple Data Encryption Standard	MAC	-	Media Access Control
3eTI	-	3e Technologies International	MAC	-	Message Authentication Code
AC	-	Alternating Current	Mbps	-	Megabits per second
AES	-	Advanced Encryption Standard	MHz	-	Megahertz
AP	-	Access Point	MIB	-	Management Information Base
AS	-	Authentication Server	MIC	-	Message Integrity Check
AWB	-	Auto-forming Wireless Bridging	MIMO	-	Multiple-input and Multiple-output
BSS	-	Basic Service Set	MP	-	Mobile Power
BSSID	-	Basic Service Set Identifier	MSDU	-	MAC Service Data Unit
CA	-	Certificate Authority	NIST	-	National Institute of Standards and Technology
CAC	-	Common Access Card	NMS	-	Network Management Station
CBC	-	Cipher Block Chaining	NTP	-	Network Time Protocol
CCM	-	Counter with CBC-MAC	OCSP	-	Online Certificate Status Protocol
CCMP	-	Counter-mode/CBC-MAC Protocol	OFDM	-	Orthogonal Frequency Division Multiplexing
CMVP	-	Cryptographic Module Validation Program	OSI	-	Open Systems Interconnection
CRL	-	Certificate Revocation List	PC	-	Personal Computer
CSV	-	Certified Server Validation	PDA	-	Personal Digital Assistant
CTR	-	Counter Mode	PEM	-	Privacy Enhanced Mail
CTS	-	Clear to Send	PHY	-	Physical Layer
DES	-	Data Encryption Standard	PIN	-	Personal Identification Number
DFS	-	Dynamic Frequency Selection	PKI	-	Public Key Infrastructure
DHCP	-	Dynamic Host Configuration Protocol	PMK	-	Primary Master Key
DLC	-	Data Link Control	PoE	-	Power-over-Ethernet
DNS	-	Domain Name System	PSK	-	Pre-Shared Key
DoD	-	Department of Defense	PTZ	-	Pan/Tilt/Zoom
DS	-	Data Server	Pwr	-	Power
DSSS	-	Direct-Sequence Spread Spectrum	RF	-	Radio Frequency
DTIM	-	Delivery Traffic Indication Message	RFC	-	Request for Comments
EAL	-	Evaluation Assurance Level	RSN	-	Robust Security Network
EAP	-	Extensible Authentication Protocol	RSSI	-	Received Signal Strength Indicator
EAPoL	-	Extensible Authentication Protocol over LAN	RSTP	-	Rapid Spanning Tree Protocol
ECB	-	Electronic Codebook	RTS	-	Request to Send
FHSS	-	Frequency-Hopping Spread Spectrum	Rx	-	Reception
FIPS	-	Federal Information Processing Standards	SHA	-	Secure Hash Algorithm
GHz	-	Gigahertz	SNMP	-	Simple Network Management Protocol
GUI	-	Graphical User Interface	SP	-	Service Pack
HTTPS	-	Hypertext Transfer Protocol Service	SS	-	Signal Strength
IANA	-	Internet Assigned Numbers Authority	SSID	-	Service Set Identifier

ICMP	-	Internet Control Message Protocol	SSL	-	Secure Socket Layer
ID	-	Identifier	STA	-	Station
IE	-	Information Element	STP	-	Spanning Tree Protocol
IEEE	-	Institute of Electrical and Electronics Engineers	TCP	-	Transmission Control Protocol
IP	-	Internet Protocol	TLS	-	Transport Layer Security
ISP	-	Internet Service Provider	TU	-	Time Unit
LAN	-	Local Area Network	Тx	-	Transmission
LED	-	Light Emitting Diode	URL	-	Uniform Resource Locator
LLC	-	Logical Link Control	VDC	-	Volts Direct Current
			VLAN	-	Virtual Local Area Network
			VPK	-	Virtual Private Network
			VPN	-	Virtual Private Network
			WAN	-	Wide Area Network
			WDS	-	Wireless Distribution System
			Wi-Fi	-	Wireless Fidelity
			WLAN	-	Wireless Local Area Network
			WPA2	_	Wi-Fi Protected Access 2

B.2 Term Definitions

Term	Definition		
Triple Data Encryption Standard (3DES)	3DES is a mode of the DES encryption algorithm that encrypts data three times.		
802.11	802.11 refers to a family of specifications developed by the IEEE for wireless LAN technology. 802.11 specifies an over-the-air interface between a wireless client and a base station or between two wireless clients. The IEEE accepted the specification in 1997		
802.11b (also referred to as 802.11 High Rate or Wi-Fi)	802.11b is an extension to 802.11 that applies to wireless LANs and provides 11 Mbps transmission (with a fallback to 5.5, 2 and 1 Mbps) in the 2.4 GHz band. 802.11b uses only DSSS. 802.11b was a 1999 ratification to the original 802.11 standard, allowing wireless functionality comparable to Ethernet.		
Access Point (AP)	An AP is a gateway set up to allow a group of LAN users access to another group or a main group. The AP does not use the DHCP server function and therefore accepts IP address assignments from the controlling network.		
Advanced Encryption Standard (AES)	AES is a symmetric 128-bit block data encryption technique developed by Belgian cryptographers Joan Daemen and Vincent Rijmen. The U.S government adopted the algorithm as its encryption technique in October 2000, replacing DES encryption. AES works at multiple network layers simultaneously.		
Bridge	A device that connects two local-area networks (LANs), or two segments of the same LAN that use the same protocol, such as Ethernet or Token-Ring.		
Dynamic Host Configuration Protocol (DHCP)	DHCP is a protocol for assigning dynamic IP addresses to devices on a network. With dynamic addressing, a device can have a different IP address every time it connects to the network. In some systems, the device's IP address can even change while it is still connected. DHCP also supports a mix of static and dynamic IP addresses. Dynamic		

Term	Definition	
	addressing simplifies network administration because the software keeps track of IP addresses rather than requiring an administrator to manage the task. This means that a new computer can be added to a network without the hassle of manually assigning it a unique IP address. Many Internet Service Providers (ISPs) use dynamic IP addressing for dial-up users.	
Network Management Station (NMS)	NMS includes management software such as, HP Openview and IBM Netview.	
Service Set Identifier (SSID)	A SSID is a Network ID unique to a network. Only clients and APs that share that the same SSID are able to communicate with each other. This string is case-sensitive Wireless LANs offer several security options, but increasing the security also mean increasing the time spent managing the system. Encryption is the key. The biggest three is from intruders coming into the LAN. You set a seven-digit alphanumeric security concalled an SSID, in each wireless device and they thereafter operate as a group.	
Wireless Local Area Network (WLAN)	A type of local-area network that uses high-frequency radio waves rather than wires to communicate between nodes.	

Appendix C. Serial I/O Interface Board

The Serial I/O Interface Board Cable is used to connect to the **SERIAL – I/O** connector on the 3e-523N as shown in Figure 107. The Serial I/O Termination connector provides secure termination for connections to serial I/O, LED status, and power as described in Table 29.





Table 29: Serial I/O Termination			
Terminal Block Label	Connection Description		
IN 1 -	General Input ¹		
IN 1 +	General Input ¹		
OUT 1 +	General Output ²		
OUT 1 -	General Output ²		
PWR -	Ground (black wire for AMR)		
IN 0 -	General Input ¹ (green wire for AMR)		
IN 0 +	General Input ¹		
OUT 0 +	General Output ²		
OUT 0 -	General Output ² (red wire for AMR)		
PY /TY	RS-485/422 receive (1/2 of pair), RS-232 Transmit (output from 3e-523N)		
104-717	Parallel to DB-9 pin 3		
RX+/RTS	RS-485/422 receive (1/2 of pair), RS-232 RTS (input to 3e-523N)		
100/1010	Parallel to DB-9 pin 7		
TX+/CTS	RS-485/422 transmit (1/2 of pair), RS-232 CTS (output from 3e-523N)		
	Parallel to DB-9 pin 8		
	RS-485/422 transmit (1/2 of pair), RS-232 Receive (input to 3e-523N)		
1.1.1.1	Parallel to DB-9 pin 2		
LED A -	Active low LED driver ³ (ties to LED Cathode, with 100 ohms limiting resistor)		
LED B -	Active low LED driver ³		
LED C -	Active low LED driver ³		



Table 29: Serial I/O Termination			
Terminal Block Label	Connection Description		
LED D -	Active low LED driver ³		
LED E -	Active low LED driver ³		
LED F -	Active low LED driver ³		
+3.3V	LED driver voltage source (ties to LED Anode terminal)		
SMB CK	SM Bus Clock signal		
SMB DT	SM Bus Data		
GND	Circuit Ground		
PWR -	Alternate power return (limited to 2A max)		
PWR +	Alternate power source (limited to 2A max)		

Notes:

- 1. Inputs limited to +13V maximum, signal drives an opto-coupler (LED side), internal circuits include current limiting resistor.
- Outputs require source voltage and current limiting. Use up to +13V and minimum 1,000 ohm limiting resistor. Source voltage (from the 3e-523N device) can be switched in using positions 1 to 3 (switch to "ON" side) of "S1" DIP switch.
- 3. LED functionality currently reserved. LED A and LED B indicate Ethernet traffic, mirrors "WAN" indicator on the 3e-523N unit.

The Serial I/O Configuration DIP switch (Figure 108) allows you to configure the interface as described in Table 30.



Figure 108: Serial I/O Configuration DIP Switch

Table 30: Serial I/O Configuration DIP Switch			
Switch	Name		
1	"ON" Connects input power rail to source both "OUT 1 +" and "IN 1 +" terminals		
2	"ON" Connects input power rail to source "IN 0 +" terminal		
3	"ON" Connects input power rail to source "OUT 0 +" terminal		
4	Set RS-485 slew rate to "low" (default, OFF position, is high slew rate)		
5	Terminate RS-485 RX pair with 120-ohm load		
6	Set to Half-Duplex RS-485 mode (must switch both 6 & 7 to "ON" position)		
7	Set to Half-Duplex RS-485 mode (must switch both 6 & 7 to "ON" position)		
8	Terminate RS-485 TX pair with 120-ohm load		

The Serial I/O Connector provides an RS232 serial I/O port via a standard DB-9 connector as shown in Table 31.

Table 31: DB-9 Serial I/O Connector			
Pin	Signal Name	Name	
1	CD	Carrier Detect	
2	RXD	Receive Data	
3	TXD	Transmit Data	
4	DTR	Data Terminal Ready	
5	GND	System Ground	
6	DSR	Data Set Ready	
7	RTS	Request to Send	
8	CTS	Clear to Send	
9	RI	Ring Indicator	

Appendix D. 3e-523E-900 Specific Operation

D.1 3e-523E-900 Overview

The 3e-523E-900 system transmits WiFi protocols over the 900MHz ISM band instead of 2.4GHz or 5GHz bands at up to 1Watt RF power. Channel selection is limited to four selections, distributed across 902 to 928 MHz. The 900MHz signal is transmitted out of the 3e-523E-900 unit at the external RF port (bulkhead N-style antenna connection) through a supplied RF cable. Antenna selections are limited to +6dBi (or lower) omni-directional of dipole construction, or a +6dBi (or lower gain) directional Yagi style.

D.2 PHY Settings Specific to 3e-523E-900

In the 3e-523E-900, wireless mode is locked as 802.11b mode. Channels are limited to 5MHz channel bandwidth and distributed as shown below.

The PHY Setting screen, see Figure 41, allows the user to access only the available channels (802.11b mode, 5MHz bandwidth per channel). Transmit power settings for the 3e-523E-900 are limited to use Fixed mode only, and up to level 4 or 5 depending on the channel. The remaining options on this page are the same as the 3e-523N.

Table 32: 3e-523E-900 Frequency Channel Numbers			
Wireless Mode	Channel No.	Fixed Power Level	
802.11b	7 (906 MHz) 8 (911 MHz) 9 (916 MHz) 10 (921 MHz)	3, 4, or 5	

Appendix E. Technical Support

E.1 Manufacturer's Statement

The 3e-523N and 3e-525N are provided with a warranty. It is not desired or expected that the user opens the device. If a malfunction is experienced and all external causes are eliminated, the user should return the unit to the manufacturer and replace it with a functioning unit.

If you are experiencing trouble with this unit, the point of contact is:

support@ultra-3eTI.com 1-800-449-3384, option 2 or visit our web site at www.ultra-3eti.com